### UC Berkeley UC Berkeley Electronic Theses and Dissertations

#### Title

String Theory, Chern-Simons Theory and the Fractional Quantum Hall Effect

**Permalink** https://escholarship.org/uc/item/0c27m3gk

Author Moore, Nathan Paul

Publication Date 2014

Peer reviewed|Thesis/dissertation

String Theory, Chern-Simons Theory and the Fractional Quantum Hall Effect

By

Nathan Paul Moore

A dissertation submitted in partial satisfaction of the

requirements for the degree of

Doctor of Philosophy

in

Physics

in the

Graduate Division

of the

University of California, Berkeley

Committee in charge:

Professor Ori Ganor, Chair Professor Petr Hořava Professor Jenny Harrison

Spring 2014

String Theory, Chern-Simons Theory and the Fractional Quantum Hall Effect

Copyright 2014 by Nathan Paul Moore

#### Abstract

String Theory, Chern-Simons Theory and the Fractional Quantum Hall Effect

by

Nathan Paul Moore

Doctor of Philosophy in Physics

University of California, Berkeley

Professor Ori Ganor, Chair

In this thesis we explore two interesting relationships between string theory and quantum field theory.

Firstly, we develop an equivalence between two Hilbert spaces: (i) the space of states of  $U(1)^n$  Chern-Simons theory with a certain class of tridiagonal matrices of coupling constants (with corners) on  $T^2$ ; and (ii) the space of ground states of strings on an associated mapping torus with  $T^2$  fiber. The equivalence is deduced by studying the space of ground states of  $SL(2,\mathbb{Z})$ -twisted circle compactifications of U(1) gauge theory, connected with a Janus configuration, and further compactified on  $T^2$ . The equality of dimensions of the two Hilbert spaces (i) and (ii) is equivalent to a known identity on determinants of tridiagonal matrices with corners. The equivalence of operator algebras acting on the two Hilbert spaces follows from a relation between the Smith normal form of the Chern-Simons coupling constant matrix and the isometry group of the mapping torus, as well as the torsion part of its first homology group.

Secondly, the Fractional Quantum Hall Effect appears as part of the low-energy description of the Coulomb branch of the  $A_1$  (2,0)-theory formulated on  $(S^1 \times \mathbb{R}^2)/\mathbb{Z}_k$ , where the generator of  $\mathbb{Z}_k$  acts as a combination of translation on  $S^1$  and rotation by  $2\pi/k$  on  $\mathbb{R}^2$ . At low-energy the configuration is described in terms of a 4+1D Super-Yang-Mills theory on a cone  $(\mathbb{R}^2/\mathbb{Z}_k)$  with additional 2+1D degrees of freedom at the tip of the cone. Fractionally charged quasi-particles have a natural description in terms of BPS strings of the (2,0)-theory. We analyze the large k limit, where a smooth cigar-geometry provides an alternative description. In this framework a W-boson can be modeled as a bound state of k quasi-particles. The W-boson becomes a Q-ball, and it can be described by a soliton solution of BPS monopole equations on a certain auxiliary curved space. We show that axisymmetric solutions of these equations correspond to singular maps from  $AdS_3$  to  $AdS_2$ , and we present some numerical results. Dedicated to my Mother and Christina.

# Contents

Li	List of Figures iv							
A	Acknowledgments v							
1	Introduction							
	1.1	Relativistic Quantum Mechanics	1					
	1.2	Chern-Simons Theory and $1/k$ Charged Particles	3					
	1.3	Supersymmetry and the (2,0) Theory	4					
	1.4	Compactification	5					
	1.5	Superstring Theory and M-Theory	5					
	1.6	Relationships between String and Field Theory	6					
<b>2</b>	Janus configurations with $SL(2, \mathbb{Z})$ -duality twists, Strings on Mapping Tori,							
	and	a Tridiagonal Determinant Formula	<b>7</b>					
	2.1	Introduction and summary of results	7					
	2.2	The $SL(2,\mathbb{Z})$ -twist	10					
	2.3	The Low-energy limit and Chern-Simons theory	11					
	2.4	Strings on a mapping torus	13					
		2.4.1 The number of fixed points	14					
		2.4.2 Isometries	14					
		2.4.3 Homology quantum numbers	15					
		2.4.4 The Hilbert space of states	16					
	2.5	Duality between strings on $\mathbf{M}_3$ and the compactified $\mathrm{SL}(2,\mathbb{Z})$ -twisted $U(1)$						
		gauge theory	17					
		2.5.1 Isomorphism of operator algebras	18					
	2.6	Discussion	19					
3	Fractional Quantum Hall Effect, Quasi-Particles, and the (2,0)-Theory							
	3.1	Introduction						
	3.2	The (2,0) theory on $(\mathbb{R}^2 \times S^1)/\mathbb{Z}_k$	22					
		3.2.1 The geometry	22					
		3.2.2 Symmetries	24					

		3.2.3	Relation to D3- $(p,q)$ 5-brane systems $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	25		
		3.2.4	Appearance of the fractional quantum Hall effect	27		
	3.3	Quasi-	$-$ particles $\dots \dots \dots$	28		
	3.4 The large $\mathbf{k}$ limit					
		3.4.1	Cigar geometry	31		
		3.4.2	Equations of motion	32		
	3.5	Integr	ally charged particles as bound states of quasi-particles	33		
		3.5.1	BPS equations	35		
		3.5.2	Derivation of the BPS equations	36		
	3.6	Analy	sis of the BPS equations	38		
		3.6.1	Manton gauge	38		
		3.6.2	Harmonic maps from $AdS_3$ to $AdS_2$	39		
		3.6.3	The abelian solution	40		
		3.6.4	Comments on (lack of) integrability	41		
	3.7	Nume	rical results	43		
Bi	ibliog	graphy		48		
A	Ар	oroof o	of the determinant identity and the Smith normal form of th	ıe		
	couj	pling c	constant matrix	53		
B Compatibility of the supersymmetric Janus configuration and the d						
	twis		56			
	B.1	Supers	symmetric Janus	56		
	B.2	Introd	ucing an $SL(2,\mathbb{Z})$ -twist $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	57		
	B.3	The su	persymmetry parameter	59		
	B.4	Exten	ding to a type-IIA supersymmetric background	59		
С	Rec	asting	BPS equations in terms of a single potential	61		
	Large VEV expansion 63					
D	Larg	ge VE	V expansion	63		

# List of Figures

3.1	(a) The geometry of $\mathbf{M}_3 \simeq (\mathbb{C} \times S^1)/\mathbb{Z}_k$ : in the coordinate system $(x_4 + ix_5, x_3)$ , the point $(r, 0)$ is identified with $(re^{-2\pi i/\mathbf{k}}, 2\pi R)$ and $(r, 2\pi \mathbf{k}R)$ : The large dots	
	indicate equivalent points: (b) The fibration $\mathbf{M}_2 \to \mathbb{C}/\mathbb{Z}_4$ , with the generic fiber	
	that is of radius $kR$ .	23
3.2	The cigar geometry with the typical scales indicated. The curvature of the cigar	
	sets the length scale $\mathbf{k}R$ , and the 4 + 1D SYM coupling constant sets the length	
	scale $g_{\rm vm}^2$	32
3.3	In the limit $va^2 \gg 1$ the soliton is approximately described by the Prasad-	
	Some field solution (of width $1/va$ ) near $r = a$ and $\rho = 0$ . Note that $\rho = -1$	
	$\sqrt{x_1^2 + x_2^2}$ and the directions $x_1, x_2$ are not drawn since they are perpendicular to	
	the $r, \theta$ directions.	34
3.4	Results of a numerical simulation with parameters $b = 2.80$ and $N = 22$ . The	
	graphs show the energy density $\Theta \equiv \mathcal{U}/\mathbf{V}$ (solid line) and the gauge invariant	
	absolute value of the scalar field $ \widetilde{\Phi}  \equiv (\widetilde{\Phi}^a \widetilde{\Phi}^a)^{1/2}$ (dashed line) for VEV $v = 1$	
	and soliton center at $a = 1$ . The graphs are on the axis $\mathbf{U} = 0$ and the horizontal	
	axis is V. At $\mathbf{V} = 0$ the value of $\Theta$ is $2.0 \times 10^{-3}$ and the value of $ \widetilde{\Phi} $ is 0.76. The	
	value of the excess energy $\mathcal{E}$ for this simulation is less than $2 \times 10^{-5}$ of $\mathcal{E}_{\text{BPS}}$ .	47
B.1	In the Janus configuration the coupling constant $\tau$ traces a portion of a semi-	
	circle of radius $4\pi D$ in the upper-half plane, whose center a is on the real axis.	

We augment it with an  $SL(2,\mathbb{Z})$  duality twist that glues  $x_3 = 2\pi$  to  $x_3 = 0$ .

58

# Acknowledgments

I would like to thank my advisor, Ori Ganor. Ori is arguably the nicest man I have ever met and easily the best advisor that anyone could ever have. I will be forever grateful to him for agreeing to take me on as a graduate student.

# Chapter 1

# Introduction

Here we give a brief introduction to the main concepts used in this thesis.

#### 1.1 Relativistic Quantum Mechanics

In order to describe the high energy behavior of the fundamental particles that make up the energy and matter in the universe, one must construct mathematical models for them that incorporate both special relativity and quantum mechanics.

In classical mechanics the equations of motion that govern the behavior of particles making up a system can be found by extremizing the action

$$S = \int dt \ L(x_i, \dot{x}_i, t),$$

which yields the Euler-Lagrange equations for the particle coordinates,  $x_i(t)$ .

In quantum mechanics a particle's position and momentum no longer commute but satisfy the canonical commutation relations

$$[x(t)_i, p(t)_j] = i\delta_{ij} \qquad p_i = \frac{\partial L}{\partial \dot{x}_i}.$$

The behavior of  $x_i$  and  $p_i$  can be found by making the Legendre transformation to the Hamiltonian picture

$$H(x_i, p_i, t) \equiv \sum_i p_i \dot{x}_i - L(x_i, \dot{x}_i, t)$$

and using the Heisenberg equations of motion, or the Feynman Path Integral. However, we see that time and space are on unequal footing since time is simply a parameter, while space is an operator, which makes it difficult to incorporate special relativity. There are two ways of alleviating this problem:

#### 1. Quantum Field Theory (QFT)

Demote space to a parameter and quantize fields  $\phi(\mathbf{x}, t)$ , governed by the action

$$S = \int d^4x \ \mathcal{L}(\phi(\mathbf{x},t),\partial_\mu \phi(\mathbf{x},t),x^\mu), \qquad \pi(\mathbf{x},t) = \frac{\partial \mathcal{L}}{\partial \dot{\phi}},$$

by imposing the commutation relations

$$[\phi(\mathbf{x},t),\pi(\mathbf{y},t)] = i\delta^3(\mathbf{x}-\mathbf{y}).$$

The Fourier modes of the field in momentum space can be interpreted as particle creation and annihilation operators that create one-particle momentum eigenstates.

The advantage of this method is that it is fairly straightforward to deal with processes that do not conserve particle number. QFT has been successfully used to describe the relativistic theory of the strong, weak and electromagnetic interactions, but runs into major difficulties in attempts to describe quantum gravity.

#### 2. String Theory

Promote time to an operator. Now we have four operators,  $X^{\mu}(\tau)$ , labeling the particle's trajectory in space-time, parameterized by a coordinate  $\tau$ . Our system is governed by an action for the one dimensional fields,  $X^{\mu}(\tau)$ ,

$$S = \int d\tau \ \mathcal{L}(X^{\mu}, \dot{X}^{\mu}, \tau).$$

For example, the simplest Poincare-invariant action that does not depend on the parameter  $\tau$  is proportional to the length of the world-line traced out by the  $X^{\mu}(\tau)$ . The Lagrangian is

$$\mathcal{L} \sim \sqrt{-\dot{X}^{\mu}\dot{X}_{\mu}},$$

and we can quantize this system by imposing the commutation relations

$$[X^{\mu}(\tau), P^{\nu}(\tau)] = i\eta^{\mu\nu}, \qquad P^{\nu} = \frac{\partial L}{\partial \dot{X}^{\nu}}$$

The main difference between this approach and the previous one is that now the Fourier modes of our one dimensional space-time fields are operators that create and destroy single particle modes rather than creating and destroying particles, which makes it more difficult to deal with processes that do not conserve particle number.

The advantage of this method is that it is relatively straightforward to write down a relativistic quantum theory of extended objects. We can quantize one-dimensional strings by using two dimensional fields,  $X^{\mu}(\tau, \sigma)$ , hence the name string theory. We can also quantize higher dimensional objects, known as p-branes, through the use of higher dimensional fields that are functions p spacial dimensions and one time dimension. While all of the forces other than gravity can be straightforwardly quantized using QFT, the graviton appears in the spectrum of closed string states and this eventually allows for the construction of a consistent theory of quantum gravity. String theory also appears to have all of the necessary ingredients to describe a relativistic quantum theory of the strong weak and electromagnetic forces, but the exact nature of how this comes about is not yet known.

#### 1.2 Chern-Simons Theory and 1/k Charged Particles

In this thesis we frequently refer to the 2+1D quantum field theory with action

$$S = \frac{k}{4\pi} \int A \wedge dA,$$

known as abelian Chern-Simons theory. A is one-form abelian gauge field, and k is an integer known as the Chern-Simons level.

1/k charged particles arise in the Fractionally Charged Quantum Hall Effect in which electrons confined to a 2D spacial surface combine with flux quanta of the magnetic field to produce fractionally charged quasiparticles. The effective QFT description of this system in terms of a Chern-Simons theory gives us a simple way of seeing how this happens.

Let  $J^{\mu}$  be the electromagnetic current of the 2+1D electrons. By charge conservation we have

$$\partial_{\mu}J^{\mu} = 0$$

Since the topology of flat 2+1D space is trivial we can write  $J^{\mu}$  globally as the curl of a 3-vector

$$J^{\mu} = \epsilon^{\mu\nu\lambda} \partial_{\nu} a_{\lambda}$$

We notice immediately that the current is invariant under the gauge transformation  $a_{\mu} \rightarrow a_{\mu} + \partial_{\mu}\Lambda$ . The effective local Lagrangian description of this system is given to leading order by a Chern-Simons term:

$$\mathcal{L} = \frac{k}{4\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} a_{\lambda} + \cdots$$

The dots indicate the kinetic term for a as well as any other terms, however, the Chern-Simons term is the only gauge invariant, dimension 3 or less, local operator that we can write down, and thus the only relevant one at long distances. Let us now couple our electron system to an external electromagnetic field via the interaction  $J^{\mu}A_{\mu}$ . If we substitute our formula for  $J^{\mu}$  as a function of a and integrate by parts (dropping a surface term) our Lagrangian becomes

$$\mathcal{L} = \frac{k}{4\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} a_{\lambda} - \frac{1}{2\pi} \epsilon^{\mu\nu\lambda} A_{\mu} \partial_{\nu} a_{\lambda} = \frac{k}{4\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} a_{\lambda} - \frac{1}{2\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} A_{\lambda}$$

The quasiparticles are defined as the entities that couple to the gauge potential through the interaction  $a_{\mu}j^{\mu}$ , where  $j^{\mu}$  is the quasiparticle current. Including quasiparticles our Lagrangian becomes

$$\mathcal{L} = \frac{k}{4\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} a_{\lambda} + a_{\mu} j^{\mu} - \frac{1}{2\pi} \epsilon^{\mu\nu\lambda} a_{\mu} \partial_{\nu} A_{\lambda}$$

Now we make the change of variables  $\tilde{j}^{\mu} = j^{\mu} - \frac{1}{2\pi} \epsilon^{\mu\nu\lambda} \partial_{\nu} A_{\lambda}$ , use Loretz gauge for  $a^{\mu}$ , and integrate out  $a^{\mu}$  to get

$$\mathcal{L} = \frac{\pi}{k} \tilde{j}^{\mu} \left( \frac{\epsilon_{\mu\nu\lambda} \partial^{\nu}}{\partial^2} \right) \tilde{j}^{\lambda} = \frac{1}{4\pi k} \epsilon^{\mu\nu\lambda} A_{\mu} \partial_{\nu} A_{\lambda} + \frac{1}{k} A_{\mu} j^{\mu} + \frac{\pi}{k} j^{\mu} \left( \frac{\epsilon_{\mu\nu\lambda} \partial^{\nu}}{\partial^2} \right) j^{\lambda}$$

The second term tells us that our quasiparticles have electric charge  $\frac{1}{k}$ .

#### 1.3 Supersymmetry and the (2,0) Theory

Supersymmetry is a symmetry between bosons and fermions. The canonical example of a supersymmetric quantum field theory is the Wess-Zumino model, whose action is given by

$$S = \int d^4x (-\partial^\mu \phi^* \partial_\mu \phi + i\psi^\dagger \bar{\sigma}^\mu \partial_\mu \psi).$$

This action describes a massless complex scalar and a massless 2-component left-handed Weyl spinor. This action is invariant under the transformations

$$\delta\phi = \epsilon\psi \qquad \delta\phi^* = \epsilon^{\dagger}\phi^{\dagger}$$
$$\delta\psi_{\alpha} = i(\sigma^{\mu}\epsilon^{\dagger})_{\alpha}\partial_{\mu}\phi \qquad \delta\psi^{\dagger}_{\dot{\alpha}} = -i(\epsilon\sigma^{\mu})_{\dot{\alpha}}\partial_{\mu}\phi^{\prime}$$

where  $\epsilon^{\alpha}$  is a two component infinitesimal Weyl spinor. The massless fermion and massless complex scalar are said to form a lefthanded chiral supermultiplet and by Noether's theorem there are two conserved supercharges associated with this symmetry.

There are many other supersymmetric quantum field theories in various dimensions with more supercharges and more complicated supermultiplets. One of the most interesting is the (2,0) Theory, which is a six dimensional supersymmetric quantum field theory with 16 supercharges. The (2,0) Theory is quite mysterious at the quantum level because it contains a tensor multiplet that has 5 scalars, 4 left-handed Weyl spinors, and a two-form gauge field, B, with self-dual field strength H = dB; H = \*H. It is difficult to quantize B using standard methods because it is very difficult to find a Lorentz covariant Lagrangian for B. The most natural Lagrangian,

$$H \wedge *H = H \wedge H = -H \wedge H = 0$$

in six dimensions. In 1996 Pasti, Sorokin and Tonin managed to write down a rather complicated Lorentz covariant Lagrangian [1], however, this only works for an abelian gauge field. There is no known way to quantize the non-abelian (2,0) Theory, and thus anything one can learn about it is of great interest.

#### 1.4 Compactification

Lower dimensional theories can be constructed from higher dimensional theories by placing the higher dimensional theories on a space-time in which one or more of the spacial dimensions form a compact subspace. At energies that are too small to resolve the compact subspace, the higher dimensional theory behaves like a lower dimensional one.

The canonical example is that of a D+1 dimensional massless scalar field on a space in which one of the spacial dimensions is a circle of length L. The equation of motion of the field in D+1 dimensions is

$$\Box_{D+1}^2 \phi(\mathbf{x}, y) = 0$$

where  $\mathbf{x}$  labels the non-compact directions, and y the circular one. Since the field is now periodic in y we can write it as a Fourier series

$$\phi(\mathbf{x}, y) = \sum_{n} \phi_n(\mathbf{x}, y) e^{\frac{iny}{L}}.$$

The equation of motion is now

$$\Box_D^2 \phi_n(\mathbf{x}) - \frac{n^2}{L^2} \phi_n = 0,$$

and we see that the D+1 dimensional theory of a single massless scalar field behaves like a D dimensional field theory of infinitely many massive scalar fields.

#### 1.5 Superstring Theory and M-Theory

By combining the bosonic  $X^{\mu}(\tau, \sigma)$  coordinates with fermionic coordinates,  $\psi^{\mu}(\tau, \sigma)$ , that are related to the bosonic ones by supersymmetry, one can construct a theory of superstrings. The latter is the simplest way of incorporating fermions into string theory. Superstring theories have string modes that can be grouped into various supersymmetry multiplets such as the gravity multiplet (which contains the graviton among other things) as well as tensor and/or vector multiplets that contain gauge bosons.

There are five distinct superstring theories that are consistent quantum theories of gravity in 10 space-time dimensions: type I, type IIA, type IIB,  $E_8 \times E_8$  Heterotic and SO(32) Heterotic. Type I consists of unoriented open and closed superstrings, has 16 supercharges, and is consistent only when the gauge group is SO(32). The two Heterotic superstring theories consist of closed oriented superstrings, have 16 supercharges, and have gauge groups of either SO(32) or  $E_8 \times E_8$ , hence the names. The type II string theories are theories of oriented closed superstrings, with 32 supercharges, and they also contain oriented open superstrings that can end on p-branes called Dp-branes. Type IIA contains only stable Dpbranes of even dimensionality while IIB contains only stable Dp-branes of odd dimensionality. The ends of the open strings in the type II theories can be charged under various gauge groups, which provides a simple way of constructing non-abelian gauge theories from them. These theories can all be related to one another through a series of dualities and are all believed to derive from a unique theory in 11 space-time dimensions known as M-theory. Unfortunately, M-theory does not contain fundamental strings, which are required for a perturbative description, hence very little is known about the full theory. M-theory does however contain M2 branes and M5 branes, and the field theory that describes the world volume of an M5 brane is the (2,0) Theory. Non-abelian forms of the (2,0) theory can be realized as the world volumes of configurations of M5 branes.

It is hoped that one day we will be able to determine how our four dimensional world arises from compactifications of M-theory and its lower dimensional descendants. In the meantime, however, one can use the relationships between the various superstring theories and M-theory to learn new things about the field theories that are associated with them.

#### 1.6 Relationships between String and Field Theory

In this thesis we explore two interesting relationships between string theory and field theory. In chapter 2 we develop an equivalence between the space of states of  $U(1)^n$  Chern-Simons theory with a certain class of tridiagonal matrices of coupling constants (with corners) on  $T^2$  and the space of ground states of strings on an associated mapping torus with  $T^2$  fiber. In chapter 3 the Fractional Quantum Hall Effect, which can be derived from a Chern-Simons theory, appears as part of the low-energy description of the Coulomb branch of the  $A_1$ (2,0)-theory formulated on  $(S^1 \times \mathbb{R}^2)/\mathbb{Z}_k$ , which is derived from M-theory.

# Chapter 2

# Janus configurations with $SL(2,\mathbb{Z})$ -duality twists, Strings on Mapping Tori, and a Tridiagonal Determinant Formula

We develop an equivalence between two Hilbert spaces: (i) the space of states of  $U(1)^n$ Chern-Simons theory with a certain class of tridiagonal matrices of coupling constants (with corners) on  $T^2$ ; and (ii) the space of ground states of strings on an associated mapping torus with  $T^2$  fiber. The equivalence is deduced by studying the space of ground states of  $SL(2, \mathbb{Z})$ twisted circle compactifications of U(1) gauge theory, connected with a Janus configuration, and further compactified on  $T^2$ . The equality of dimensions of the two Hilbert spaces (i) and (ii) is equivalent to a known identity on determinants of tridiagonal matrices with corners. The equivalence of operator algebras acting on the two Hilbert spaces follows from a relation between the Smith normal form of the Chern-Simons coupling constant matrix and the isometry group of the mapping torus, as well as the torsion part of its first homology group.

#### 2.1 Introduction and summary of results

Our goal is to develop tools for studying circle compactifications of  $\mathcal{N} = 4$  Super-Yang-Mills theory on  $S^1$  with a general SL(2,  $\mathbb{Z}$ )-duality twist (also known as a "duality wall") inserted at a point on  $S^1$ . The low-energy limit of such compactifications encodes information about the operator that realizes the SL(2,  $\mathbb{Z}$ )-duality, and can potentially teach us new facts about S-duality itself. Some previous works on duality walls and related compactifications include [2]-[11].

In this paper we consider only the abelian gauge group G = U(1), leaving the extension to nonabelian groups for a separate publication [12]. We focus on the Hilbert space of ground states of the system and study it in two equivalent ways: (i) directly in field theory; and (ii) via a dual type-IIA string theory system (extending the techniques developed in [27]). As we will show, the equivalence of these two descriptions implies the equivalence of:

(i) the Hilbert space of ground states of  $U(1)^n$  Chern-Simons theory with action

$$L = \frac{1}{4\pi} \sum_{i=1}^{n} \mathbf{k}_{i} A_{i} \wedge dA_{i} - \frac{1}{2\pi} \sum_{i=1}^{n-1} A_{i} \wedge dA_{i+1} - \frac{1}{2\pi} A_{1} \wedge dA_{n},$$

on  $T^2$ , and

(ii) the Hilbert space of ground states of strings of winding number w = 1 on a certain target space that contains the *mapping torus* with  $T^2$  fiber:

$$\mathbf{M}_3 \equiv \frac{I \times T^2}{(0, v) \sim (1, f(v))}, \qquad (v \in T^2),$$

where I = [0, 1] is the unit interval, and f is a large diffeomorphism of  $T^2$  corresponding to the SL(2,  $\mathbb{Z}$ ) matrix

$$W \equiv \begin{pmatrix} \mathbf{k}_n & -1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} \mathbf{k}_2 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} \mathbf{k}_1 & -1 \\ 1 & 0 \end{pmatrix} .$$
 (2.1)

We will explain the construction of these Hilbert spaces in detail below.

An immediate consequence of the proposed equivalence of Hilbert spaces (i) and (ii) is the identity

$$\det \begin{pmatrix} \mathbf{k}_{1} & -1 & 0 & -1 \\ -1 & \ddots & \ddots & \ddots & \\ 0 & \ddots & \ddots & \ddots & 0 \\ & \ddots & \ddots & \ddots & -1 \\ -1 & 0 & -1 & \mathbf{k}_{n} \end{pmatrix} = \operatorname{tr} \begin{bmatrix} \begin{pmatrix} \mathbf{k}_{n} & -1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} \mathbf{k}_{2} & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} \mathbf{k}_{1} & -1 \\ 1 & 0 \end{pmatrix} \end{bmatrix} - 2. \quad (2.2)$$

which follows from the equality of dimensions of the Hilbert spaces above. This is a known identity (see for instance [13]), and we will present a proof in Appendix A, for completeness.<sup>1</sup>

Moreover, equivalence of the operator algebras of the systems associated with (i) and (ii) allows us to make a stronger statement. The operator algebra of (i) is generated by Wilson loops along two fundamental cycles of  $T^2$ , and keeping only one of these cycles gives a maximal finite abelian subgroup. Let  $\Lambda \subseteq \mathbb{Z}^n$  be the sublattice of  $\mathbb{Z}^n$  generated by the columns of the Chern-Simons coupling constant matrix, which appears on the LHS of

<sup>&</sup>lt;sup>1</sup> The continuum limit of (2.2) with  $n \to \infty$  and  $\mathbf{k}_i \to 2 + \frac{1}{n^2}V(\frac{i}{n})$  might be more familiar. It leads to a variant of the Gelfand-Yaglom theorem [14] with a periodic potential:  $\det[-d^2/dx^2 + V(x)] =$ tr  $\left[P \exp \oint \begin{pmatrix} \sqrt{V} + \frac{V'}{2V} & \frac{V'}{2V} \\ -\sqrt{V} & -\sqrt{V} \end{pmatrix} dx \right] - 2$  (up to a renormalization-dependent multiplicative constant).

(2.2). Then, the abelian group generated by the maximal commuting set of Wilson loops is isomorphic to  $\mathbb{Z}^n/\Lambda$ . The operator algebra of (ii), on the other hand, is constructed by combining the isometry group of  $\mathbf{M}_3$  with the group of operators that measure the various components of string winding number in  $\mathbf{M}_3$ . The latter is captured algebraically by the Pontryagin dual  $^{\vee}(\cdots)$  of the torsion part Tor of the homology group  $H_1(\mathbf{M}_3, \mathbb{Z})$ . (The terms will be explained in more detail in §2.4.3.) Thus,  $^{\vee}$  Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  as well as the isometry group are both equivalent to  $\mathbb{Z}^n/\Lambda$ . Together,  $^{\vee}$  Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  and Isom( $\mathbf{M}_3$ ) generate a noncommutative (but reducible) group that is equivalent to the operator algebra of the Wilson loops of the Chern-Simons system in (i). The subgroup  $^{\vee}$  Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  corresponds to the group generated by the Wilson loops along one fixed cycle of  $T^2$  (let us call it "the  $\alpha$ cycle") and Isom( $\mathbf{M}_3$ ) corresponds to the group generated by the Wilson loops along another cycle (call it "the  $\beta$ -cycle"), where  $\alpha$  and  $\beta$  generate  $H_1(T^2, \mathbb{Z})$ . The situation is summarized in the following diagram:



We will now present a detailed account of the statements made above. In §2.2 we construct the  $SL(2,\mathbb{Z})$ -twist from the QFT perspective, and in §2.3 we take its low-energy limit and make connection with  $U(1)^n$  Chern-Simons theory, leading to Hilbert space (i). In §2.4 we describe the dual construction of type-IIA strings on  $\mathbf{M}_3$ . In §2.5 we develop the "dictionary" that translates between the states and operators of (i) and (ii). We conclude in §2.6 with a brief summary of what we have found so far and a preview of the nonabelian case.

#### 2.2 The $SL(2,\mathbb{Z})$ -twist

Our starting point is a free 3+1D U(1) gauge theory with action

$$I = \frac{1}{4g_{\rm ym}^2} \int F \wedge^* F + \frac{\theta}{2\pi} \int F \wedge F,$$

where F = dA is the field strength. As usual, we define the complex coupling constant

$$\tau \equiv \frac{4\pi i}{g_{\rm ym}^2} + \frac{\theta}{2\pi} \equiv \tau_1 + i\tau_2.$$

The SL(2,  $\mathbb{Z}$ ) group of dualities is generated by S and T that act as  $\tau \to -1/\tau$  and  $\tau \to \tau+1$ , respectively.

Let the space-time coordinates be  $x_0, \ldots, x_3$ . We wish to compactify direction  $x_3$  on a circle (so that  $0 \le x_3 \le 2\pi$  is a periodic coordinate), but allow  $\tau$  to vary as a function of  $x_3$  in such a way that

$$\tau(0) = \frac{\mathbf{a}\tau(2\pi) + \mathbf{b}}{\mathbf{c}\tau(2\pi) + \mathbf{d}},$$

where  $W \equiv \begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix} \in SL(2, \mathbb{Z})$  defines an electric/magnetic duality transformation. Such a compactification contains two ingredients:

- The variable coupling constant  $\tau$ ; and
- The "duality-twist" at  $x_3 = 0 \sim 2\pi$ .

We will discuss the ingredients separately, starting from the duality-twist.

The duality-twist can be described concretely in terms of an abelian Chern-Simons theory as follows. Represent the  $SL(2,\mathbb{Z})$  matrix in terms of the generators S and T (nonuniquely) as

$$\begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix} = T^{k_1} S T^{k_2} S \cdots T^{k_n} S , \qquad (2.3)$$

where  $\mathbf{k}_1, \ldots, \mathbf{k}_n$  are integers, some of which may be zero. To understand how each of the operators T and S act separately, we pretend that  $x_3$  is a time-direction and impose the temporal gauge condition  $A_3 = 0$ . At any given  $x_3$  the wave-function is formally  $\Psi(\mathcal{A})$ , where  $\mathcal{A}$  is the gauge field 1-form on the three-dimensional space parameterized by  $x_0, x_1, x_2$ . The action of the generators S and T on the wave-functions is then given by (see for instance [15, 16]):

$$S: \Psi(\mathcal{A}) \to \int e^{-\frac{i}{2\pi} \int \mathcal{A} \wedge d\mathcal{A}'} \Psi(\mathcal{A}') \mathcal{D}\mathcal{A}', \qquad T: \Psi(\mathcal{A}) \to e^{\frac{i}{4\pi} \int \mathcal{A} \wedge d\mathcal{A}} \Psi(\mathcal{A}).$$

It is now clear how to incorporate the duality twist by combining these two elements to realize the  $SL(2,\mathbb{Z})$  transformation (2.3). We have to add to the action a Chern-Simons term at  $x_3 = 0$  with additional auxiliary fields  $A_1, \ldots, A_{n+1}$  and with action

$$I_{CS} = \frac{1}{4\pi} \sum_{i=1}^{n} k_i A_i \wedge dA_i - \frac{1}{2\pi} \sum_{i=1}^{n} A_i \wedge dA_{i+1}, \qquad (2.4)$$

and then set

$$A_1 = A|_{x_3=0}$$
,  $A_{n+1} = A|_{x_3=2\pi}$ .

The second ingredient is the varying coupling constant  $\tau(x_3)$ . Systems with such a varying  $\tau$ are known as Janus configurations [17]. They have supersymmetric extensions [18]-[52] where the Lagrangian of N = 4 Super-Yang-Mills with variable  $\tau$  is modified so as to preserve 8 supercharges. In such configurations the function  $\tau(x_3)$  traces a geodesic in the hyperbolic upper-half  $\tau$ -plane, namely, a half-circle centered on the real axis [52]. In this model, the surviving supersymmetry is described by parameters that vary as a function of  $x_3$ , so that in general the supercharges at  $x_3 = 0$  are not equal to those at  $x_3 = 2\pi$ . This might have been a problem for us, since we need to continuously connect  $x_3 = 0$  to  $x_3 = 2\pi$  to form a consistent supersymmetric theory, but luckily, we also have the  $SL(2,\mathbb{Z})$ -twist, and as shown in [21], in N = 4 Super-Yang-Mills (with a fixed coupling constant  $\tau$ ), the SL(2,  $\mathbb{Z}$ ) duality transformations do not commute with the supercharges. Following the action of duality, the SUSY generators pick up a known phase. But as it turns out, this phase exactly matches the phase difference from 0 to  $2\pi$  in the Janus configuration. Therefore, we can combine the two separate ingredients and close the supersymmetric Janus configuration on the segment  $[0, 2\pi]$  with an SL $(2, \mathbb{Z})$  duality twist that connects 0 to  $2\pi$ . We describe this construction in more detail in Appendix B.

The details of the supersymmetric action, however, will not play an important role in what follows, so we will just assume supersymmetry and proceed. Thanks to mass terms that appear in the Janus configuration (which are needed to close the SUSY algebra [52]), at low-energy the superpartners of the gauge fields are all massive (see Appendix B), with masses of the order of the Kaluza-Klein scale, and we can ignore them. We will therefore proceed with a discussion of only the free U(1) gauge fields.

#### 2.3 The Low-energy limit and Chern-Simons theory

At low-energy we have to set  $A_1 = A_{n+1}$  in (2.4), since the dependence of A on  $x_3$  is suppressed. Then, the low-energy system is described by a 2+1D Chern-Simons action with gauge group  $U(1)^n$  and action

$$I = \frac{1}{4\pi} \sum_{i,j=1}^{n} K_{ij} A_i \wedge dA_j \,,$$

with coupling-constant matrix that is given by

$$K \equiv \begin{pmatrix} \mathbf{k}_{1} & -1 & 0 & -1 \\ -1 & \ddots & \ddots & \ddots & \\ 0 & \ddots & \ddots & \ddots & 0 \\ & \ddots & \ddots & \ddots & -1 \\ -1 & 0 & -1 & \mathbf{k}_{n} \end{pmatrix} .$$
 (2.5)

We now make directions  $x_1, x_2$  periodic, so that the theory is compactified on  $T^2$ , leaving only time uncompactified. The dimension of the resulting Hilbert space of states of this compactified Chern-Simons theory is  $|\det K|$ .

Next, we pick two fundamental cycles whose equivalence classes generate  $H_1(T^2, \mathbb{Z})$ . Let  $\alpha$  be the cycle along a straight line from (0,0) to (1,0), and let  $\beta$  be a similar cycle from (0,0) to (0,1), in  $(x_1, x_2)$  coordinates. We define 2n Wilson loop operators:

$$U_j \equiv \exp\left(i\oint_{\alpha}A_j\right), \qquad V_j \equiv \exp\left(i\oint_{\beta}A_j\right), \qquad j=1,\ldots,n.$$

They are unitary operators with commutation relations given by

$$U_i U_j = U_j U_i$$
,  $V_i V_j = V_j V_i$ ,  $U_i V_j = e^{2\pi i (K^{-1})_{ij}} V_j U_i$ .

 $[(K^{-1})_{ij}$  is the i, j element of the matrix  $K^{-1}$ .] In particular, for any  $j = 1, \ldots, n$  the operator  $\prod_{i=1}^{n} U_i^{K_{ij}}$  commutes with all 2n operators, and hence is a central element. In an irreducible representation, it can be set to the identity. The  $U_i$ 's therefore generate a finite abelian group, which we denote by  $\mathbf{g}$ . Similarly, we denote by  $\mathcal{G}_{\beta}$  the finite abelian group generated by the  $V_i$ 's. Both groups are isomorphic and can be described as follows. Let  $\Lambda \subseteq \mathbb{Z}^n$  be the sublattice of  $\mathbb{Z}^n$  generated by the columns of the matrix K. Then,  $\mathbb{Z}^n/\Lambda$  is a finite abelian group and  $\mathbf{g} \cong \mathcal{G}_{\beta} \cong \mathbb{Z}^n/\Lambda$ , since an element of  $\mathbb{Z}^n$  represents the powers of a monomial in the  $U_i$ 's (or  $V_i$ 's), and an element in  $\Lambda$  corresponds to a monomial that is a central element. We therefore map

$$\mathbf{g} \ni \prod_{i=1}^{n} U_i^{N_i} \mapsto (N_1, N_2, \dots, N_n) \in \mathbb{Z}^n \pmod{\Lambda}, \qquad (2.6)$$

and similarly

$$\mathcal{G}_{\beta} \ni \prod_{i=1}^{n} V_i^{M_i} \mapsto (M_1, M_2, \dots, M_n) \in \mathbb{Z}^n \pmod{\Lambda}.$$
(2.7)

We denote the operator in  $\mathbf{g}$  that corresponds to  $\mathbf{v} \in \mathbb{Z}^n / \Lambda$  by  $\mathcal{O}_{\alpha}(\mathbf{v})$ , and similarly we define  $\mathcal{O}_{\beta}(\mathbf{v}) \in \mathcal{G}_{\beta}$  to be the operator in  $\mathcal{G}_{\beta}$  that corresponds to  $\mathbf{v}$ . For  $\mathbf{u}, \mathbf{v} \in \mathbb{Z}^n / \Lambda$  we define

$$\chi(\mathfrak{u},\mathfrak{v}) \equiv e^{2\pi i \sum_{i,j} (K^{-1})_{ij} N_i M_j}, \qquad (\mathfrak{u},\mathfrak{v} \in \mathbb{Z}^n / \Lambda).$$
(2.8)

The definition is independent of the particular representatives  $(N_1, \ldots, N_n)$  or  $(M_1, \ldots, M_n)$ in  $\mathbb{Z}^n/\Lambda$ . The commutation relations can then be written as

$$\mathcal{O}_{\alpha}(\mathfrak{u})\mathcal{O}_{\beta}(\mathfrak{v}) = \chi(\mathfrak{u},\mathfrak{v})\mathcal{O}_{\beta}(\mathfrak{v})\mathcal{O}_{\alpha}(\mathfrak{u}).$$
(2.9)

We recall that for any nonsingular matrix of integers  $K \in GL(n, \mathbb{Z})$ , one can find matrices  $P, Q \in SL(n, \mathbb{Z})$  such that

$$PKQ = \operatorname{diag}(d_1, d_2, \dots, d_n) \tag{2.10}$$

is a diagonal matrix,  $d_1, \ldots, d_n$  are positive integers, and  $d_i$  divides  $d_{i+1}$  for  $i = 1, \ldots, n-1$ . The integers  $d_1, \ldots, d_n$  are unique, and we have

$$\mathbb{Z}^n/\Lambda \cong \mathbb{Z}_{d_1} \oplus \cdots \oplus \mathbb{Z}_{d_n}$$

where  $\mathbb{Z}_d$  is the cyclic group of d elements. The matrix on the RHS of (2.10) is known as the *Smith normal form* of K. The integer  $d_j$  is the greatest common divisor of all  $j \times j$  minors of K. For K of the form (2.5), the minor that is made of rows  $2, \ldots, n-1$  and columns  $1, \ldots, n-2$  is  $(-1)^{n-2}$ , so it follows that  $d_{n-2} = 1$  and therefore also  $d_1 = \cdots d_{n-2} = 1$ . We conclude that

$$\mathbf{g} \cong \mathcal{G}_{\beta} \cong \mathbb{Z}_{d_{n-1}} \oplus \mathbb{Z}_{d_n}$$
.

#### 2.4 Strings on a mapping torus

The system we studied in §2.2 has a dual description as the Hilbert space of ground states of strings of winding number w = 1 (around a 1-cycle to be defined below) on a certain type-IIA background. We will begin by describing the background geometry and then explain in §2.5 why its space of ground states is isomorphic to the space of ground states of the SL(2, Z)-twisted compactification of §2.2.

Set

$$W = \begin{pmatrix} \mathbf{k}_n & -1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} \mathbf{k}_2 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} \mathbf{k}_1 & -1 \\ 1 & 0 \end{pmatrix} = T^{\mathbf{k}_n} S \cdots T^{\mathbf{k}_2} S T^{\mathbf{k}_1} S \equiv \begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix} \in \mathrm{SL}(2, \mathbb{Z}).$$
(2.11)

We will assume that  $|\operatorname{tr} W| > 2$  so that W is a hyperbolic element of  $\operatorname{SL}(2, \mathbb{Z})$ . (The case of elliptic elements with  $|\operatorname{tr} W| < 2$  was covered in [27], and parabolic elements with  $|\operatorname{tr} W| = 2$  are conjugate to  $\pm T^{\mathbf{k}}$  for some  $\mathbf{k} \neq 0$ , and since they do not involve S, they are elementary.)

Let  $0 \leq \eta \leq 2\pi$  denote the coordinate on the interval  $I = [0, 2\pi]$  and let  $(\xi_1, \xi_2)$  denote the coordinates of a point on  $T^2$ . The coordinates  $\xi_1$  and  $\xi_2$  take values in  $\mathbb{R}/\mathbb{Z}$  (so they are periodic with period 1). We impose the identification

$$(\xi_1, \xi_2, \eta) \sim (\mathbf{d}\xi_1 + \mathbf{b}\xi_2, \mathbf{c}\xi_1 + \mathbf{a}\xi_2, \eta + 2\pi).$$
 (2.12)

The metric is

$$ds^{2} = R^{2} d\eta^{2} + \left(\frac{4\pi^{2}\rho^{2}}{\tau_{2}}\right) |d\xi_{1} + \tau(\eta)d\xi_{2}|^{2}$$

where R and  $\rho$  are constants, and  $\tau = \tau_1 + i\tau_2$  is a function of  $\eta$  (with real and imaginary parts denoted by  $\tau_1$  and  $\tau_2$ ) such that

$$\tau(\eta + 2\pi) = \frac{\mathbf{a}\tau(\eta) + \mathbf{b}}{\mathbf{c}\tau(\eta) + \mathbf{d}},$$

thus allowing for a continuous metric.

#### 2.4.1 The number of fixed points

We will need the number of fixed points of the  $SL(2,\mathbb{Z})$  action on  $T^2$ , i.e., the number of solutions to:

$$(\xi_1,\xi_2) = (\mathbf{d}\xi_1 + \mathbf{b}\xi_2, \mathbf{c}\xi_1 + \mathbf{a}\xi_2) \pmod{\mathbb{Z}^2}.$$

Let  $f: T^2 \to T^2$  be the map given by

$$f: (\xi_1, \xi_2) \to (\mathbf{d}\xi_1 + \mathbf{b}\xi_2, \mathbf{c}\xi_1 + \mathbf{a}\xi_2).$$
(2.13)

The Lefschetz fixed-point formula states that

$$\sum_{\text{fixed point } p} i(p) = \sum_{j=0}^{2} (-1)^{j} \operatorname{tr}(f_{*} | H_{j}(T^{2}, \mathbb{Z})) = 2 - \operatorname{tr} W = 2 - \mathbf{a} - \mathbf{d}.$$

The index i(p) of a fixed point is given by [22]:

$$i(p) = \operatorname{sgn} \det(\mathcal{J}(p) - \mathbf{I}) = \operatorname{sgn} \det(W - \mathbf{I}),$$

where  $\mathcal{J}(p)$  is the Jacobian matrix of the map f at p. In our case, i(p) is either +1 or -1 for all p, and therefore the number of fixed points is

$$|2 - \operatorname{tr} W| = |\det(W - \mathbf{I})| = |2 - \mathbf{a} - \mathbf{d}|.$$

#### 2.4.2 Isometries

Let  $v_1, v_2 \in \mathbb{R}/\mathbb{Z}$  be constants and consider the map

$$(\xi_1, \xi_2, \eta) \mapsto (\xi_1 + v_1, \xi_2 + v_2, \eta).$$
 (2.14)

It defines an isometry of  $\mathbf{M}_3$  if

$$\begin{pmatrix} \mathbf{a} & \mathbf{c} \\ \mathbf{b} & \mathbf{d} \end{pmatrix} \begin{pmatrix} v_2 \\ v_1 \end{pmatrix} \equiv \begin{pmatrix} v_2 \\ v_1 \end{pmatrix} \pmod{\mathbb{Z}}.$$
 (mod  $\mathbb{Z}$ ). (2.15)

Set

$$H \equiv W - \mathbf{I} = \begin{pmatrix} \mathbf{a} - 1 & \mathbf{c} \\ \mathbf{b} & \mathbf{d} - 1 \end{pmatrix}, \qquad \mathbf{v} \equiv \begin{pmatrix} v_2 \\ v_1 \end{pmatrix}.$$
(2.16)

Then, the isometries are given by  $\mathbf{v} = H^{-1} \begin{pmatrix} n_2 \\ n_1 \end{pmatrix}$  for some  $n_1, n_2 \in \mathbb{Z}$ . The set of vectors  $\mathbf{v}$  that give rise to isometries therefore live on a lattice  $\tilde{\Lambda}$  generated by the columns of  $H^{-1}$ . Since  $H \in \mathrm{GL}(2,\mathbb{Z})$  we have  $\mathbb{Z}^2 \subseteq \tilde{\Lambda}$ , and since the isometries that correspond to  $\mathbf{v} \in \mathbb{Z}^2$  are trivial, the group of isometries of type (2.14) is isomorphic to  $\tilde{\Lambda}/\mathbb{Z}^2$ . Changing basis to  $\mathbf{u} \equiv H\mathbf{v}$ , we can replace  $\mathbf{v} \in \tilde{\Lambda}/\mathbb{Z}^2$  with  $\mathbf{u} \in \mathbb{Z}^2/\Lambda'$ , where  $\Lambda' \subseteq \mathbb{Z}^2$  is the sublattice generated by the columns of H, and the group  $\mathcal{G}_{iso}$  of isometries of type (2.14) is therefore

$$\mathcal{G}_{\rm iso} \cong \widetilde{\Lambda} / \mathbb{Z}^2 \cong \mathbb{Z}^2 / \Lambda' \,.$$
 (2.17)

Its order is

$$|\mathcal{G}_{\rm iso}| = |\det H| = |2 - \mathbf{a} - \mathbf{d}|. \tag{2.18}$$

#### 2.4.3 Homology quantum numbers

To proceed we also need the homology group  $H_1(\mathbf{M}_3, \mathbb{Z})$ . Let  $\gamma$  be the cycle defined by a straight line from (0,0,0) to  $(0,0,2\pi)$ , in terms of  $(\xi_1,\xi_2,\eta)$  coordinates. Let  $\alpha'$  be the cycle from (0,0,0) to (1,0,0) and let  $\beta'$  be the cycle from (0,0,0) to (0,1,0). The homology group  $H_1(\mathbf{M}_3,\mathbb{Z})$  is generated by the equivalence classes  $[\alpha']$ ,  $[\beta']$  and  $[\gamma]$ , subject to the relations

$$[\alpha'] = \mathbf{d}[\alpha'] + \mathbf{c}[\beta'], \qquad [\beta'] = \mathbf{b}[\alpha'] + \mathbf{a}[\beta'].$$
(2.19)

Now suppose that  $(c_1 c_2)$  is a linear combination of the columns of H [defined in (2.16)] with integer coefficients. Then the relations (2.19) imply that  $c_1[\alpha'] + c_2[\beta']$  is zero in  $H_1(\mathbf{M}_3, \mathbb{Z})$ . With  $\Lambda' \subset \mathbb{Z}^2$  being the sublattice generated by the columns of H, as defined in §2.4.2, it follows that

$$H_1(\mathbf{M}_3, \mathbb{Z}) \cong \mathbb{Z} \oplus (\mathbb{Z}^2 / \Lambda'),$$
 (2.20)

where the  $\mathbb{Z}$  factor is generated by  $[\gamma]$  and  $(\mathbb{Z}^2/\Lambda')$  is generated by  $[\alpha']$  and  $[\beta']$ . In particular, the torsion part is

Tor 
$$H_1(\mathbf{M}_3, \mathbb{Z}) \cong \mathbb{Z}^2 / \Lambda'$$
. (2.21)

Denote the Smith normal form [see (2.10)] of the matrix H by  $\begin{pmatrix} d'_1 \\ d'_2 \end{pmatrix}$ . We prove in Appendix A that  $d_{n-1} = d'_1$  and  $d_n = d'_2$ , where  $d_{n-1}$  and  $d_n$  were defined in (2.10). Thus, combining (2.17) and (2.20) we have

$$\mathbb{Z}^2/\Lambda' \cong \mathcal{G}_{\mathrm{iso}} \cong \mathrm{Tor}\, H_1(\mathbf{M}_3, \mathbb{Z}) \cong \mathbb{Z}_{d_{n-1}} \oplus \mathbb{Z}_{d_n} \,.$$

The physical meaning of these results will become clear soon.

#### 2.4.4 The Hilbert space of states

As we have seen in §2.4.2, the Hilbert space of string ground states has a basis of states of the form  $|\mathbf{v}'\rangle$  with  $\mathbf{v}' \in \tilde{\Lambda}/\mathbb{Z}^2$ . In this state, the string is at  $(\xi_1, \xi_2)$  coordinates given by  $\mathbf{v}'$ . According to (2.17), an element  $\mathbf{v} \in \tilde{\Lambda}/\mathbb{Z}^2$  defines an isometry, which we denote by  $\mathcal{Y}(\mathbf{v})$ , that acts as

$$|\mathcal{Y}(\mathbf{v})|\mathbf{v}'
angle = |\mathbf{v}+\mathbf{v}'
angle, \qquad \mathbf{v},\mathbf{v}'\in\widetilde{\Lambda}/\mathbb{Z}^2$$
 .

Given the string state  $|\mathbf{v}'\rangle$ , we can ask what is the element in  $H_1(\mathbf{M}_3, \mathbb{Z})$  that represents the corresponding 1-cycle. The answer is  $[\gamma] + N'_1[\alpha'] + N'_2[\beta']$ , where the torsion part  $N'_1[\alpha'] + N'_2[\beta']$  is mapped under (2.21) to  $\mathbf{v}'$ . To see this, note that for  $0 \leq t \leq 1$ , the loops  $C_t$  in  $\mathbf{M}_3$  that are given by

$$\begin{cases} (4\pi s, tv'_1, tv'_2) & \text{for } 0 \le s \le \frac{1}{2} \\ (2\pi, tv'_1 + (2s-1)t[(\mathbf{d}-1)v'_1 + \mathbf{b}v'_2], tv'_2 + (2s-1)t[\mathbf{c}v'_1 + (\mathbf{a}-1)v'_2]) & \text{for } \frac{1}{2} \le s \le 1 \end{cases}$$

[which go along direction  $\eta$  at a constant  $(\xi_1, \xi_2)$  given by  $t\mathbf{v}'$ , and then connect  $t\mathbf{v}'$  to its  $SL(2,\mathbb{Z})$  image  $tW\mathbf{v}'$ ] are homotopic to the loop corresponding to string state  $|0\rangle$ . Setting t = 1 we find that  $C_1$  breaks into two closed loops, one corresponding to string state  $|\mathbf{v}'\rangle$ , and the other is a closed loop in the  $T^2$  fiber above  $\eta = 0$ , which corresponds to the homology element

$$((\mathbf{d}-1)v_1'+\mathbf{b}v_2')[\alpha']+(\mathbf{c}v_1'+(\mathbf{a}-1)v_2')[\beta'],$$

and this is precisely the element corresponding to  $H\mathbf{v}' \in \mathbb{Z}^2/\Lambda' \cong \text{Tor } H_1(\mathbf{M}_3, \mathbb{Z})$ , as defined in §2.4.3.

We now wish to use the torsion part of the homology to define a unitary operator  $\mathcal{R}(\tilde{\mathbf{u}})$ for every  $\tilde{\mathbf{u}} \in \mathbb{Z}^2/\Lambda'$ . This operator will measure a component of the charge associated with the homology class of the string. For this purpose we need to construct the Pontryagin dual group  $\vee$  Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$ , which is defined as the group of characters of Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  (i.e., homomorphisms from Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  to  $\mathbb{R}/\mathbb{Z}$ ). The dual group is isomorphic to  $\mathbb{Z}^2/\Lambda'$ , but not canonically. In our construction  $\tilde{\mathbf{u}}$  is naturally an element of the dual group and not the group itself. We define  $\mathcal{R}(\tilde{\mathbf{u}})$  as follows. For

$$\widetilde{\mathbf{u}} = (M_1', M_2') \in \mathbb{Z}^2 / \Lambda', \qquad \mathbf{v} = (N_1', N_2') \in \mathbb{Z}^2 / \Lambda',$$

we define the phase

4

$$\varphi(\tilde{\mathbf{u}}, \mathbf{v}) \equiv e^{2\pi i (H^{-1})_{ij} N'_i M'_j}, \qquad \tilde{\mathbf{u}} \in \mathbb{Z}^2 / \Lambda', \quad \mathbf{v} \in \mathbb{Z}^2 / \Lambda'.$$
(2.22)

This definition is independent of the representatives  $(N'_1, N'_2)$  and  $(M'_1, M'_2)$  of **v** and  $\tilde{\mathbf{u}}$ , and it corresponds to the character of Tor  $H_1(\mathbf{M}_3, \mathbb{Z})$  associated with  $\tilde{\mathbf{u}}$ . We then define the operator  $\mathcal{R}(\tilde{\mathbf{u}})$  to be diagonal in the basis  $|\mathbf{v}\rangle$  and act as:

$$\mathcal{R}(\tilde{\mathbf{u}})|\mathbf{v}
angle = \varphi(\tilde{\mathbf{u}},\mathbf{v})|\mathbf{v}
angle, \qquad \tilde{\mathbf{u}} \in \mathbb{Z}^2/\Lambda', \quad \mathbf{v} \in \mathbb{Z}^2/\Lambda'$$

From the discussion above about the homology of the string state, and from the linearity of the phase of  $\varphi(\tilde{\mathbf{u}}, \mathbf{v})$  in  $\tilde{\mathbf{u}}$  and  $\mathbf{v}$ , it follows that

$$\mathcal{R}(\tilde{\mathbf{u}})\mathcal{Y}(\mathbf{v}) = \varphi(\tilde{\mathbf{u}}, \mathbf{v})\mathcal{Y}(\mathbf{v})\mathcal{R}(\tilde{\mathbf{u}}).$$
(2.23)

# 2.5 Duality between strings on $M_3$ and the compactified $SL(2,\mathbb{Z})$ -twisted U(1) gauge theory

We can now connect the string theory model of §2.4 with the field theory model of §2.3. We claim that the Hilbert space of ground states of a compactification of a U(1) gauge theory on  $S^1$  with an SL(2, Z) twist and string ground states on  $\mathbf{M}_3$  are dual. This is demonstrated along the same lines as in [27]. We realize the (supersymmetric extension of the) U(1) gauge theory on a D3-brane along directions  $x_1, x_2, x_3$ . We compactify direction  $x_3$  on a circle with a Janus-like configuration and SL(2, Z)-twisted boundary conditions. We assume that the Janus configuration can be lifted to type-IIB, perhaps with additional fluxes, but we will not worry about the details of the lift. We then compactify  $(x_1, x_2)$  on  $T^2$  and perform T-duality on direction 1, followed by a lift from type-IIA to M-theory (producing a new circle along direction 10), followed by reduction to type-IIA along direction 2. This combined U-duality transformation transforms the SL(2, Z)-twist to the geometrical transformation (2.12). It also transforms some of the charges of the type-IIB system to the following charges of the type-IIA system:

$$D3_{123} \to F1_3$$
,  $F1_1 \to P_1$ ,  $F1_2 \to F1_{10}$ ,  $D1_1 \to F1_1$ ,  $D1_2 \to P_{10}$ . (2.24)

where  $P_j$  is Kaluza-Klein momentum along direction j,  $Dp_{j_1...j_r}$  is a Dp-brane wrapped along directions  $j_1, \ldots, j_r$ , and  $F1_j$  is a fundamental string along direction j.

Now suppose we take the limit that all directions of  $\mathbf{M}_3$  are large. The dual geometry has a Hilbert space of ground states which corresponds to classical configurations of strings of minimal length that wind once around the  $x_3$  circle. This means that the projection of their  $H_1(\mathbf{M}_3, \mathbb{Z})$  homology class on the  $\mathbb{Z}$  factor of (2.20) is required to be the generator  $[\gamma]$ . The torsion part of their homology is unrestricted. The string configurations of minimal length must have constant  $(x_1, x_2)$  which in particular means that  $(x_1, x_2)$  is invariant under the SL(2,  $\mathbb{Z}$ ) twist, i.e.,

$$\begin{pmatrix} \mathbf{a} & \mathbf{c} \\ \mathbf{b} & \mathbf{d} \end{pmatrix} \begin{pmatrix} x_2 \\ x_1 \end{pmatrix} \equiv \begin{pmatrix} x_2 \\ x_1 \end{pmatrix} \pmod{\mathbb{Z}}.$$

But this is precisely the same equation as (2.15), and indeed when the isometry that corresponds to a vector  $\mathbf{v} \in \widetilde{\Lambda}/\mathbb{Z}^2$  acts on the solution with  $(x_1, x_2) = (0, 0)$  it converts it to the solution with  $(x_1, x_2) = (v_1, v_2)$ . The dimension of the Hilbert space of ground states of the type-IIA system is therefore the order of  $\mathcal{G}_{iso}$ , which is given by (2.18). This is also the number of fixed points of the W action on  $T^2$ , as we have seen in §2.4.1. Since the number of ground states of the Chern-Simons theory is  $|\det K|$ , we conclude from the duality of the Chern-Simons theory and string theory that

$$|\det K| = |\mathcal{G}_{iso}| = |2 - \mathbf{a} - \mathbf{b}|.$$

This is the physical explanation we are giving to (2.2).

#### 2.5.1 Isomorphism of operator algebras

Going one step beyond the equality of dimensions of the Hilbert spaces, we would like to match the operator algebras of the string and field theory systems. Starting with the field theory side, realized on a D3-brane in type-IIB, consider a process whereby a fundamental string that winds once around the  $\beta$ -cycle of  $T^2$  is absorbed by the D3-brane at some time t. This process is described in the field theory by inserting a Wilson loop operator  $V_1$  at time t into the matrix element that calculates the amplitude. On the type-IIA string side, the charge  $F1_2$  that was absorbed is mapped by (2.24) to winding number along the  $\alpha'$  cycle (denoted by  $F1_{10}$ ). The operator that corresponds to  $V_1$  on the string side must therefore increase the homology class of the string state by  $[\alpha']$ . Since the state  $|\mathbf{v}\rangle$ , for  $\mathbf{v} = (N'_1, N'_2)$ , has homology class  $[\gamma] + N'_1[\alpha'] + N'_2[\beta']$ , it follows that the isometry operator  $\mathcal{Y}(\mathbf{v}')$  with  $\mathbf{v}' = (1, 0)$  does what we want. We therefore propose to identify

$$V_1 \to \mathcal{Y}(\mathbf{v}'), \quad \text{for } \mathbf{v}' = (1,0).$$

By extension, we propose to map the abelian subgroup  $\mathcal{G}_{\beta}$  generated by the Wilson loops  $V_1, \ldots, V_n$  with the isometry group generated by  $\mathcal{Y}(\mathbf{v}')$  for  $\mathbf{v}' \in \mathbb{Z}^2/\Lambda'$ .

Next, on the type-IIB side, consider a process whereby a fundamental string that winds once around the  $\alpha$ -cycle of  $T^2$  is absorbed by the D3-brane. This process is described in the field theory by inserting a Wilson loop operator  $U_1$  into the matrix element that calculates the amplitude. On the type-IIA string side, the charge  $F1_1$  that was absorbed is mapped by (2.24) to momentum along the  $\beta'$  cycle (denoted by  $P1_1$ ). The operator that corresponds to  $U_1$  on the string side must therefore increase the momentum along the  $[\alpha']$  cycle by one unit. We claim that this operator is  $\mathcal{R}(\tilde{\mathbf{u}})$  for  $\tilde{\mathbf{u}} = (1,0)$ . To see this we note that, by definition of "momentum", an operator  $\mathcal{X}$  that increases the momentum by  $M'_1$  units along the  $[\alpha']$  cycle and  $M'_2$  units along the  $[\beta']$  cycle must have the following commutation relations with the translational isometries  $\mathcal{Y}(\mathbf{v}')$ :

$$\mathcal{Y}(\mathbf{v}')^{-1}\mathcal{X}\mathcal{Y}(\mathbf{v}') = \varphi(\tilde{\mathbf{u}}, \mathbf{v}')\mathcal{X}, \qquad \tilde{\mathbf{u}} = (M'_1, M'_2) \in \mathbb{Z}^2/\Lambda'.$$

But given (2.23), this means that up to an unimportant central element, we can identify  $\mathcal{X} = \mathcal{R}(\tilde{\mathbf{u}})$ , as claimed. So, we have

$$U_1 \to \mathcal{Y}(\tilde{\mathbf{u}}), \quad \text{for } \tilde{\mathbf{u}} = (1,0),$$

and by extension, we propose to map the abelian subgroup  $\mathbf{g}$  generated by the Wilson loops  $U_1, \ldots, U_n$  with the subgroup generated by  $\mathcal{R}(\tilde{\mathbf{u}})$  for  $\tilde{\mathbf{u}} \in \mathbb{Z}^2/\Lambda'$ .

In particular,  $\mathbf{g} \cong \mathcal{G}_{\beta} \cong \mathbb{Z}^n / \Lambda$  implies that  $(\mathbb{Z}^2 / \Lambda') \cong (\mathbb{Z}^n / \Lambda)$ . This is equivalent to requiring that the Smith normal form of H is

$$P'HQ' = \operatorname{diag}(d_{n-1}, d_n)$$

where  $d_{n-1}$  and  $d_n$  are the same last two entries in the Smith normal form of K. We provide an elementary proof of this fact in Appendix A. Since the Smith normal forms of H and K are equal, the abelian groups  $\mathbb{Z}^n/\Lambda$  and  $\mathbb{Z}^2/\Lambda'$  are equivalent, and it is also not hard to see that under this equivalence  $\chi$  that was defined in (2.8) is mapped to  $\varphi$  defined in (2.22). We have the mapping

$$\mathcal{O}_{lpha}(\mathfrak{v}) o \mathcal{Y}(\mathbf{v}')\,, \qquad \mathfrak{v} \in \mathbb{Z}^n/\Lambda\,, \qquad \mathbf{v}' \in \mathbb{Z}^2/\Lambda'$$

and

$$\mathcal{O}_{\beta}(\mathfrak{u}) \to \mathcal{R}(\tilde{\mathbf{u}}), \qquad \mathfrak{u} \in \mathbb{Z}^n / \Lambda, \qquad \tilde{\mathbf{u}} \in \mathbb{Z}^2 / \Lambda'.$$

The commutation relations (2.9) are then mapped to (2.23).

#### 2.6 Discussion

We have argued that a duality between  $U(1)^n$  Chern-Simons theory on  $T^2$  with coupling constant matrix (2.5) and string configurations on a mapping torus provide a geometrical realization to the algebra of Wilson loop operators in the Chern-Simons theory. Wilson loop operators along one cycle of  $T^2$  correspond to isometries that act as translations along the fiber of the mapping torus, while Wilson loop operators along the other cycle correspond to discrete charges that can be constructed from the homology class of the string.

These ideas have an obvious extension to the case of U(N) gauge group with N > 1, where  $SL(2, \mathbb{Z})$ -duality is poorly understood. The techniques presented in this paper can be extended to construct the algebra of Wilson loop operators. The Hilbert space on the string theory side is constructed from string configurations on a mapping torus whose  $H_1(\mathbf{M}_3, \mathbb{Z})$ class maps to N under the projection map  $\mathbf{M}_3 \to S^1$ . In other words, the homology class projects to  $N[\gamma]$  when the torsion part is ignored. Such configurations could be either a single-particle string state wound N times, or a multi-particle string state. A string state with r strings with winding numbers  $N_1, \ldots, N_r$  is described by a partition  $N = N_1 + \cdots N_r$ , and the  $j^{th}$  single-particle string state is described by an unordered set of  $N_j$  points on  $T^2$ that is invariant, as a set, under the action of f in (2.13). The counterparts of the Wilson loops on the string theory side can then be constructed from operations on these sets. A more complete account of the nonabelian case will be reported elsewhere [12].

It is interesting to note that some similar ingredients to the ones that appear in this work also appeared in [31] in the study of vacua of compactifications of the free (2, 0) theory on Lens spaces. More specifically, a Chern-Simons theory with a tridiagonal coupling constant matrix and the torsion part of the first homology group played a role there as well. It would be interesting to further explore the connection between these two problems.

# Chapter 3

# Fractional Quantum Hall Effect, Quasi-Particles, and the (2,0)-Theory

The Fractional Quantum Hall Effect appears as part of the low-energy description of the Coulomb branch of the  $A_1$  (2,0)-theory formulated on  $(S^1 \times \mathbb{R}^2)/\mathbb{Z}_k$ , where the generator of  $\mathbb{Z}_k$  acts as a combination of translation on  $S^1$  and rotation by  $2\pi/k$  on  $\mathbb{R}^2$ . At low-energy the configuration is described in terms of a 4+1D Super-Yang-Mills theory on a cone  $(\mathbb{R}^2/\mathbb{Z}_k)$  with additional 2+1D degrees of freedom at the tip of the cone. Fractionally charged quasiparticles have a natural description in terms of BPS strings of the (2,0)-theory. We analyze the large k limit, where a smooth cigar-geometry provides an alternative description. In this framework a W-boson can be modeled as a bound state of k quasi-particles. The W-boson becomes a Q-ball, and it can be described by a soliton solution of BPS monopole equations on a certain auxiliary curved space. We show that axisymmetric solutions of these equations correspond to singular maps from  $AdS_3$  to  $AdS_2$ , and we present some numerical results.

#### 3.1 Introduction

The Fractional Quantum Hall Effect (FQHE) with filling-factor  $1/\mathbf{k}$  ( $\mathbf{k} \in \mathbb{Z}$ ) appears in 2+1D condensed matter systems whose low-energy effective degrees of freedom can be described by the Chern-Simons action

$$I = \frac{\mathbf{k}}{4\pi} \int \mathbf{a} \wedge d\mathbf{a} + \frac{1}{2\pi} \int \mathbf{A} \wedge d\mathbf{a} \,. \tag{3.1}$$

Here, **A** is the electromagnetic gauge field, and **a** is a 2+1D U(1) gauge field that describes the low-energy internal degrees of freedom of the system. It is related to the electromagnetic current by  $j = {}^{*}d\mathfrak{a}$ . Excited levels of the system may include quasi-particle excitations that are charged under the gauge symmetry associated with  $\mathfrak{a}$ . Such quasi-particles with one unit of  $\mathfrak{a}$ -charge will have  $1/\mathbf{k}$  electromagnetic charge.

The FQHE appears in appropriately constructed 2+1D condensed-matter systems with strongly correlated electrons in a strong magnetic field. The (Laughlin) wave-functions of the low-energy states are then holomorphic in the position of the electrons (up to a common Gaussian factor). One of the remarkable features of this system is that there is a dual description of the low-energy spectrum in which the quasi-particles are fundamental and the electrons can be viewed as bound states of the quasi-particles.

The goal of this paper is to gain a better understanding of this dual description of an integrally charged particle as a bound state of quasi-particles using a particularly intuitive string-theoretic model of the FQHE. Over the past two decades several realizations of FQHE in string theory have been constructed [50]-[55]. Generally speaking, these realizations construct the Chern-Simons action (3.1) as low-energy effective description of a (d+2)-dimensional brane compactified on a d-dimensional compact space, possibly in the presence of suitable fluxes, to yield the requisite 2 + 1D effective description. We will use a realization in terms of a compactification of the 5+1D (2,0)-theory. Our system is a special case of a general class of 2 + 1D theories obtained from the (2, 0)-theory by taking three of the dimensions to be a nontrivial manifold. A beautiful framework for understanding such compactifications has been developed in [29]-[31]. We will focus on a particular aspect of the system which is the dynamics of the quasi-particles that in the condensed-matter system can arise from impurities. The (2,0)-theory allows for a simple geometrical realization of the quasi-particles and their relationship with the integrally charged particles. In our construction, the geometry of the extra dimensions will have long 1-cycles and short 1-cycles, the short ones being  $1/\mathbf{k}$  the size of the long ones. The quasi-particles will be realized as BPS strings of the (2,0) theory wound around short 1-cycles, while the integrally charged particles (the "electrons") will be realized as strings wound around long 1-cycles. The construction in terms of D3-branes ending on (p,q) 5-branes [54] is dual to ours.

We are especially interested in the limit  $\mathbf{k} \gg 1$ , where the filling fraction becomes extremely small. This is the strong-coupling limit of the condensed-matter system, and as we will see, our model has a dual description where quasi-particles are elementary and the integrally charged particles can be described as classical solitons, or rather Q-balls, in terms of the fundamental quasi-particle fields. We will show that solutions to the equations describing these solitons correspond to harmonic maps from  $AdS_3$  to  $AdS_2$ .

The paper is organized as follows. In §3.2 we describe the (2, 0) compactification. In §3.3 we study the quasi-particles which are BPS strings and calculate their quantum numbers. In §3.4 we study the large **k** limit and write down the semiclassical action of the system. In §3.5 we develop the differential equations that describe the integrally charged particles as solitons of the fundamental quasi-particle fields in the large **k** limit. We show that they can be mapped to the equations describing a magnetic monopole on a 3D space with metric  $ds^2 = x_3^2(dx_1^2 + dx_2^2 + dx_3^2)$ . In §3.6 we analyze the soliton equations in more detail. They are not integrable in the standard sense, and we were unable to solve them in closed form, but we were able to make several observations: (i) using a rather complicated transformation we can recast the equations in terms of a single "potential" function; (ii) we present an expansion up to second order in an asymptotic corner of moduli space; (iii) we plot a numerical solution; and (iv) we propose a connection with Sine-Gordon equation in a corner of moduli space.

## **3.2** The (2,0) theory on $(\mathbb{R}^2 \times S^1)/\mathbb{Z}_k$

Our starting point is the 5+1D  $A_1$  (2,0)-theory on  $\mathbb{R}^{2,1} \times \mathbf{M}_3$ , where  $\mathbb{R}^{2,1}$  is 2+1D Minkowski space and  $\mathbf{M}_3 \simeq (\mathbb{R}^2 \times S^1)/\mathbb{Z}_{\mathbf{k}}$  is the flat, noncompact, smooth three-dimensional manifold defined as the quotient of  $\mathbb{R}^2 \times S^1$  by the isometry that acts as a simultaneous rotation of  $\mathbb{R}^2$  by an angle  $2\pi/\mathbf{k}$ , and a translation of  $S^1$  by  $1/\mathbf{k}$  of its circumference. The  $A_1$  (2,0)-theory is the low-energy limit of either type-IIB on  $\mathbb{R}^4/\mathbb{Z}^2$  [25] or of 2 M5-branes [26] (after decoupling of the center of mass variables). We are interested in the low-energy degrees of freedom that are localized near the origin of  $\mathbb{R}^2$ . The fractional quantum Hall effect, as we shall see, naturally appears in this context.

#### 3.2.1 The geometry

The space  $\mathbf{M}_3$  can be constructed as a quotient of  $\mathbb{R}^3$  as follows. We parameterize  $\mathbb{R}^3$  by  $x_3, x_4, x_5$  and set  $\mathbf{z} \equiv x_4 + ix_5$ . Then,  $\mathbf{M}_3$  is defined by the equivalence relation

$$(x_3, \mathbf{z}) \sim (x_3 + 2\pi R, \mathbf{z} e^{-2\pi i/\mathbf{k}}), \qquad [\text{defining relation of } \mathbf{M}_3]$$
(3.2)

where R is a constant parameter that sets the scale, and  $\mathbf{k} > 1$  is an integer. The Euclidean metric on  $\mathbf{M}_3$  is given by

$$ds^{2} = dx_{3}^{2} + dx_{4}^{2} + dx_{5}^{2} = dx_{3}^{2} + |d\mathbf{z}|^{2}.$$

For future reference we define the  $(2\mathbf{k})^{th}$  root of unity:

$$\omega \equiv e^{\pi i/\mathbf{k}} \,. \tag{3.3}$$

We also set

$$\mathbf{z} = re^{i\theta}$$
,

so that (3.2) can be written as

$$(x_3, r, \theta) \sim (x_3 + 2\pi R, r, \theta - \frac{2\pi}{k}).$$
 (3.4)

The  $\mathbf{z} = 0$  locus [i.e., the set of points  $(x_3, 0)$  with arbitrary  $x_3$ ] forms an  $S^1$  of radius R that we will call the minicircle and denote by  $S_m^1$ . The space  $\mathbf{M}_3 \setminus S_m^1$  (which is  $\mathbf{M}_3$  with the minicircle excluded) is a circle-bundle over a cone:

$$\begin{array}{cccc} S^1 & \longrightarrow & \mathbf{M}_3 \\ & \downarrow & \\ & \mathbb{C}/\mathbb{Z}_{\mathbf{k}} \end{array} \tag{3.5}$$

The cone  $\mathbb{C}/\mathbb{Z}_{\mathbf{k}}$  is parameterized by  $\mathbf{z}$ , subject to the equivalence relation  $\mathbf{z} \sim \omega^2 \mathbf{z}$ . In polar coordinates the cone is parameterized by  $(r, \theta)$  with  $0 < r < \infty$  and  $0 \leq \theta < 2\pi/\mathbf{k}$ . ( $\theta$  is



Figure 3.1: (a) The geometry of  $\mathbf{M}_3 \simeq (\mathbb{C} \times S^1)/\mathbb{Z}_{\mathbf{k}}$ : in the coordinate system  $(x_4 + ix_5, x_3)$ , the point (r, 0) is identified with  $(re^{-2\pi i/\mathbf{k}}, 2\pi R)$  and  $(r, 2\pi \mathbf{k}R)$ ; The large dots indicate equivalent points; (b) The fibration  $\mathbf{M}_3 \to \mathbb{C}/\mathbb{Z}_{\mathbf{k}}$  with the generic fiber that is of radius  $\mathbf{k}R$ .

understood to have period  $2\pi/\mathbf{k}$  when describing the cone.) The projection  $\mathbf{M}_3 \to \mathbb{C}/\mathbb{Z}_{\mathbf{k}}$  is given by  $(x_3, \mathbf{z}) \mapsto \mathbf{z}$ . For a given  $\mathbf{z} \neq 0$ , the fiber  $S^1$  of the fibration (3.5) over  $\mathbf{z} \simeq \omega^2 \mathbf{z}$ is given by all points  $(x_3, \mathbf{z})$  with  $0 \leq x_3 < 2\pi \mathbf{k}R$ . The equivalence (3.2) then implies  $(x_3 + 2\pi \mathbf{k}R, \mathbf{z}) \sim (x_3, \mathbf{z})$ , and so this  $S^1$  has radius  $\mathbf{k}R$ .

In order to preserve half of the 16 supersymmetries we augment (3.2) by an appropriate R-symmetry twist as follows. Let  $\text{Spin}(5) \simeq Sp(2)$  be the R-symmetry of the (2, 0)-theory. In the M5-brane realization of the (2, 0)-theory [26], Spin(5) is the group of rotations (acting on spinors) in the five directions transverse to the M5-branes, which we take to be  $6, \ldots, 10$ . We now split them into the subsets 6, 7 and 8, 9, 10. This corresponds to the rotation subgroup  $[\text{Spin}(3) \times \text{Spin}(2)]/\mathbb{Z}_2 \subset \text{Spin}(5)$ . Let  $\gamma \in \text{Spin}(5)$  correspond to a  $2\pi/\mathbf{k}$  rotation in the 6, 7 plane. We then augment the RHS of the geometrical identification (3.2) by an R-symmetry transformation  $\gamma$ . The setting now preserves 8 supersymmetries.

We now go to the Coulomb branch of the (2, 0)-theory by separating the two M5-branes of §3.2.1 in the M-theory direction  $x_{10}$ . This breaks Spin(3) to an SO(2) subgroup (corresponding to rotations in directions 8,9) which we denote by  $SO(2)_r$ . On the Coulomb branch of the (2, 0)-theory there is a BPS string whose tension we denote by  $\tilde{V}$ .

At energies  $E \ll 1/\mathbf{k}R$ , sufficiently far away from  $S_m^1$ , the dynamics of the (2, 0)-theory on  $\mathbb{R}^{2,1} \times \mathbf{M}_3$  reduces to SU(2) 4+1D Super-Yang-Mills theory on  $\mathbb{R}^{2,1} \times (\mathbb{C}/\mathbb{Z}_k)$ . The coupling constant is given by

$$\frac{4\pi}{g_{\rm vm}^2} = \frac{1}{\mathbf{k}R} \,. \tag{3.6}$$

All fields are functions of the coordinates  $(x_0, x_1, x_2, r, \theta)$ , but the periodicity  $\theta \sim \theta + 2\pi/\mathbf{k}$ 

is modified in two ways:

• The shift by  $2\pi R$  in  $x_3$ , expressed in (3.4), implies that as we cross the  $\theta = 2\pi/\mathbf{k}$  ray a translation by  $2\pi R$  in  $x_3$  is needed in order to patch smoothly with the  $\theta = 0$  ray. Since  $x_3$ -momentum corresponds to conserved instanton charge in the low-energy SYM, we find that we have to add to the standard SYM action an additional term

$$\frac{1}{16\mathbf{k}\pi} \int_{\theta=0} \operatorname{tr}(F \wedge F), \qquad (3.7)$$

where the integral is performed on the ray at  $\theta = 0$ .

• the R-symmetry twist  $\gamma$  introduces phases in the relation between values of fields at  $\theta = 0$  and at  $\theta = 2\pi/\mathbf{k}$ . Of the five (gauge group adjoint-valued) scalar fields  $\Phi^6, \ldots, \Phi^{10}$  (corresponding to M5-brane fluctuations in directions  $6, \ldots, 10$ ) the last three  $\Phi^8, \Phi^9, \Phi^{10}$  are neutral under  $\gamma$  and hence periodic in  $\theta$ , while the combination  $Z \equiv \Phi^6 + i\Phi^7$  satisfies

$$Z(x_0, x_1, x_2, r, \theta + \frac{2\pi}{\mathbf{k}}) = \omega^2 \Omega^{-1} Z(x_0, x_1, x_2, r, \theta) \Omega.$$
(3.8)

where we have included an arbitrary gauge transformation  $\Omega(x_0, x_1, x_2, r) \in SU(n)$ . The gluinos have similar boundary conditions with appropriate  $\exp(\pm \pi/\mathbf{k})$  phases.

At the origin  $\mathbf{z} = 0$ , which is the tip of the cone  $\mathbb{C}/\mathbb{Z}_k$ , boundary conditions need to be specified and additional 2+1D degrees of freedom need to be added. These degrees of freedom and their interactions with the bulk SYM fields are the main focus of this paper and will be discussed in §3.2.4. But at this point we can make a quick observation. When a BPS string of the (2,0)-theory wraps the  $S^1$  of (3.5) we get the W-boson of the effective 4+1D SYM. The circle has radius  $\mathbf{k}R$  and so the mass of the W-boson is  $2\pi\mathbf{k}R\tilde{V}$ . On the other hand, the BPS string can also wrap the minicircle  $S_m^1$  whose radius is only R. The resulting particle in 2+1D has mass  $2\pi R\tilde{V}$  which is  $1/\mathbf{k}$  of the mass of the W-boson. Its charge is also  $1/\mathbf{k}$ of the charge of the W-boson. This is our first hint that we are dealing with a system that exhibits a Fractional Quantum Hall Effect. We will soon see that indeed a BPS string that wraps  $S_m^1$  can be identified with a quasiparticle of FQHE.

#### 3.2.2 Symmetries

Now, let us discuss the symmetries of the theory at a generic point on the Coulomb branch. The continuous isometries of  $\mathbf{M}_3$  are generated by translations of  $x_3$  and rotations of the **z**-plane. We denote the latter by  $SO(2)_{\mathbf{z}}$  and normalize the respective charge so that  $d\mathbf{z}$  has charge +1. The isometry group of  $\mathbf{M}_3$  also contains a discrete  $\mathbb{Z}_2$  factor generated by the orientation-preserving isometry

$$(x_3, \mathbf{z}) \mapsto (-x_3, \overline{\mathbf{z}}).$$

This by itself does not preserve our setting because it converts the R-symmetry twist  $\gamma$  to  $\gamma^{-1}$ . To cure this problem, we introduce an extra reflection  $x_7 \to -x_7$ , and finally, in order to preserve parity we also introduce another reflection, say,  $x_9 \to -x_9$ . Altogether, we define the discrete symmetry  $\mathbb{Z}'_2$  to be generated by

$$(x_0, x_1, x_2, x_3, \mathbf{z}, x_6, x_7, x_8, x_9, x_{10}) \mapsto (x_0, x_1, x_2, -x_3, \mathbf{\overline{z}}, x_6, -x_7, x_8, -x_9, x_{10}). \qquad [\mathbb{Z}'_2]$$
(3.9)

Next, the SO(2) subgroup of the R-symmetry that corresponds to rotations in the 6-7 plane will be referred to as  $SO(2)_{\gamma}$  and normalized so that  $\Phi^6 + i\Phi^7$  has charge +1. The SU(2) = Spin(3) subgroup of the R-symmetry that corresponds to rotations in the 8,9,10 directions will be referred to as  $SU(2)_R$ . For future reference we also denote the SO(2) subgroup of rotations in the 8,9 plane by  $SO(2)_r$ .

The parity symmetry of M-theory [28], which acts as reflection on an odd number of dimensions combined with a reversal of the 3-form gauge field  $(C_3 \rightarrow -C_3)$  can also be used to construct a symmetry of our background. We define  $\mathbb{Z}_2^{\prime\prime}$  as the discrete symmetry generated by the reflection that acts as

$$x_{10} \rightarrow -x_{10}, \quad C_3 \rightarrow -C_3.$$
  $[\mathbb{Z}_2'']$ 

This symmetry preserves the M5-brane configuration and the twist. We summarize the symmetries in the following table:

$SO(2)_{\mathbf{z}}$	rotations of the $\mathbf{z}$ $(x_4 - x_5)$ plane;
$SO(2)_{\gamma}$	rotations of the $x_6 - x_7$ plane;
$SU(2)_R$	rotations of the $x_8, x_9, x_{10}$ plane;
$SO(2)_r$	rotations of the $x_8, x_9$ plane;
$\mathbb{Z}_2'$	reflection in directions $x_3, x_5, x_7, x_{10};$
$\mathbb{Z}_2''$	reflection in direction $x_{10}$ (and $C_3 \rightarrow -C_3$ );

We denote the conserved charges associated with  $SO(2)_{\mathbf{z}}$ ,  $SO(2)_{\gamma}$ , and  $SO(2)_r$  by  $q_{\mathbf{z}}$ ,  $q_{\gamma}$ , and  $q_r$ , respectively. These are the spins in the 4 - 5, 6 - 7, and 8 - 9 planes. The supersymmetry generators are also charged under these groups, and the background preserves those supercharges for which  $q_{\mathbf{z}} + q_{\gamma} = 0$ . These observations will become useful in §3.3, where we will study the quantum numbers of the quasi-particles.

#### **3.2.3** Relation to D3-(p, q)5-brane systems

As we have seen in §3.2.1, following dimensional reduction on the  $S^1$  fiber of (3.5), we get a low-energy description in terms of 4+1D SYM on the cone  $\mathbb{C}/\mathbb{Z}_k$ , interacting with additional (as yet unknown, but to be described below) degrees of freedom at the tip of the cone (at  $x_4 = x_5 = 0$ ). These additional degrees of freedom are three-dimensional and can be expressed in terms of SU(2) Chern-Simons theory coupled to the IR limit of a U(1) gauge theory with two charged hypermultiplets (with  $\mathcal{N} = 4$  supersymmetry in 2 + 1D).

The latter is the self-mirror theory introduced in [56], and named T(SU(2)) by Gaiotto and Witten [54]. The arguments leading to the identification of the degrees of freedom at the tip of the cone were presented, in a somewhat different but related context, in [27]. The idea is to relate the local degrees of freedom of M-theory on the geometry of §3.2.1 to those of a (p,q) 5-brane of type-IIB, as originally done in [34], and then map our two M5-branes to two D3-branes, to obtain the problem of two D3-branes ending on a (p,q) 5-brane. This is precisely the problem that was solved in [54] in terms of T(SU(2)), which thus also furnishes the solution to our problem. On the Coulomb branch, the gauge part of the system reduces to U(1) Chern-Simons theory interacting with T(U(1)), which reproduces (3.1). Although the details of the argument will not be needed for the rest of this paper, we will review them below for completeness. More details can be found in [27].

Our geometry in directions  $3, \ldots, 7$  is of the form  $(S^1 \times \mathbb{C}^2)/\mathbb{Z}_k$ , and leads to a  $(1, \mathbf{k})$  5-brane. This was demonstrated in [34] by replacing  $\mathbb{C}^2$  with a Taub-NUT space, whose metric can be written as

$$ds^{2} = \left(1 + \frac{\tilde{R}}{2\tilde{r}}\right)^{-1} \left(dy + \cos\tilde{\theta}\,d\tilde{\phi}\right)^{2} + \left(1 + \frac{\tilde{R}}{2\tilde{r}}\right) \left[d\tilde{r}^{2} + \tilde{r}^{2}(d\tilde{\theta}^{2} + \sin^{2}\tilde{\theta}\,d\tilde{\phi}^{2})\right],\qquad(3.10)$$

where y is a periodic coordinate with range  $0 \leq y < 2\pi$ . We then add an additional  $S^1$ , parameterized by  $x_3$  as in (3.2). The plane  $\mathbb{C}$  that appears in (3.2) is now embedded in the  $\mathbb{C}^2$  tangent space of the Taub-NUT space at the origin  $\tilde{r} = 0$ , and is recovered in the limit  $\tilde{R} \to \infty$ . In that limit, and with a change of variables  $\tilde{r} = \tilde{R}r^2$ , we can identify the  $\mathbb{C}$ plane of (3.2) as a plane at constant  $(\tilde{\theta}, \tilde{\phi})$  (say  $\tilde{\theta} = \pi/2$  and  $\tilde{\phi} = 0$ ), and the  $\mathbf{z} \equiv x_4 + ix_5$ coordinate of (3.2) is identified with

$$\mathbf{z} = r e^{iy} = \sqrt{\tilde{R}\tilde{r}} \, e^{iy}.$$

In this limit  $\hat{R} \to \infty$ , the  $x_6, x_7$  plane is identified with a transverse plane to the **z**-plane, which we can take to be given by  $\tilde{\phi} = \pm \pi/2$  and y = 0. We now return to the finite  $\tilde{R}$ geometry, and impose the  $\mathbb{Z}_{\mathbf{k}}$  equivalence of (3.2) by setting

$$(x_3, y, \tilde{r}, \tilde{\theta}, \tilde{\phi}) \sim (x_3 + 2\pi R, y - \frac{2\pi}{\mathbf{k}}, \tilde{r}, \tilde{\theta}, \tilde{\phi})$$

We then wrap two M5-branes on the  $(\tilde{\theta} = \frac{\pi}{2}, \tilde{\phi} = 0)$  subspace of this 5-dimensional geometry. In the limit  $\tilde{R} \to \infty$  this reproduces the setting of §3.2.1.

The technique that Witten employed in [34] is to convert the Taub-NUT geometry to a D6-brane by reduction on the *y*-circle from M-theory to type-IIA, and then apply T-duality on the  $x_3$ -circle to get type-IIB with a complex string coupling constant of the form

$$\tau_{\rm IIB} = \frac{2\pi i}{g_{\rm IIB}} - \frac{1}{\mathbf{k}} \,,$$

which turns out to be strongly coupled  $(g_{\text{IIB}} \to \infty)$  in the limit  $\tilde{R} \to \infty$ , but can, in turn, be converted to weak coupling with an  $\text{SL}(2,\mathbb{Z})$  transformation

$$\tau_{\rm IIB} \to \tau'_{\rm IIB} = \frac{\tau_{\rm IIB}}{\mathbf{k}\tau_{\rm IIB} + 1} = \frac{1}{\mathbf{k}} + \frac{ig_{\rm IIB}}{2\pi\mathbf{k}^2} \to \frac{1}{\mathbf{k}} + i\infty \,.$$

As explained in [34], the combined transformations convert the Taub-NUT geometry to a 5brane of (p, q)-type  $(1, \mathbf{k})$  [where  $\mathbf{k}$  is the NS5-charge and 1 is the D5-charge]. It also converts the M5-branes to D3-branes. The boundary degrees of freedom where the two D3-branes end on the  $(1, \mathbf{k})$  5-brane were found in [54] as follows. Let A denote the boundary 2 + 1Dvalue of the SU(2) gauge field of the D3-branes (with the superpartners left implicit). One then notes that

$$\begin{pmatrix} 1 \\ \mathbf{k} \end{pmatrix} = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & \mathbf{k} \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 0 \\ 1 \end{pmatrix}$$

Thus a  $(1, \mathbf{k})$  5-brane can be obtained from an NS5-brane by a combining the two transformations:  $\tau \to \tau + \mathbf{k}$ , followed by  $\tau \to -1/\tau$ . Each transformation can be implemented on the boundary conditions. The  $\tau \to \tau + \mathbf{k}$  transformation introduces a level-**k** Chern-Simons theory expressed in terms of an ancillary SU(2) gauge field that we denote by A', and the  $\tau \to -1/\tau$  (S-duality) transformation introduces 2 + 1D degrees of freedom, named T(SU(2)) by Gaoitto and Witten, that couple to both the A and A' gauge fields. T(SU(2))was identified with the Intriligator-Seiberg theory [56] that is defined as the low-energy limit of  $\mathcal{N} = 4 U(1)$  gauge theory coupled to two hypermultiplets. The theory has a classical SU(2) flavor symmetry (which will ultimately couple to, say, the gauge field A), and it also has a U(1) global symmetry under which only magnetic operators are charged, and this symmetry is enhanced to SU(2) in the (strongly coupled) low-energy limit. This SU(2) is then coupled to A'. It is also not hard to check that A is the  $r \to 0$  limit of the 4+1D gauge field on the cone. To see this, consider the  $T^2$  formed by varying  $(x_3, y)$  for fixed  $r, \theta$ , and  $\phi$ . The SL(2,  $\mathbb{Z}$ ) transformation  $\begin{pmatrix} 1 & 0 \\ \mathbf{k} & 1 \end{pmatrix}$  converts 1-cycle from (0,0) to  $(2\pi R, -2\pi/\mathbf{k})$  into the 1-cycle from (0,0) to  $(2\pi \mathbf{k}R,0)$ , and this is precisely the 1-cycle used in the reduction from the (2,0)-theory to 4+1D SYM.

#### 3.2.4 Appearance of the fractional quantum Hall effect

On the Coulomb branch the SU(2) gauge group of 4+1D SYM is broken to U(1). At energies below the breaking scale, the SU(2) gauge fields A and A' reduce to U(1) gauge fields which we denote by **A** and **a**. The theory T(SU(2)) reduces to T(U(1)) which is described by the action [54]  $(1/2\pi) \int \mathbf{A} \wedge d\mathbf{a}$ . The total gauge part of the action at the tip of the cone is therefore given by (3.1). As we have already seen, the BPS strings that wrap the minicircle  $S_m^1$  have fractional charge 1/**k** under the bulk **A** which we have now identified as the unbroken U(1) gauge field of the bulk 4+1D SYM. If we move such a string away from the tip, adiabatically, we will get a string that, in the  $(x_3, y)$  coordinates of §3.2.3, wraps the 1-cycle from (0,0) to  $(2\pi R, -2\pi/\mathbf{k})$ . This implies that it has one unit of charge under **a**, which lends credence to the proposal of identifying such a string with a quasi-particle of FQHE. The quasi-particle is confined to  $\mathbb{R}^{2,1}$ , because everywhere else a wound string is longer than the BPS bound  $2\pi R$ .

Following the breaking of SU(2) to U(1), the bulk 4+1D W-boson gets a mass. The W-boson corresponds to a (2,0)-string wound around the  $S^1$  fiber of (3.5), and the homotopy
class of the bulk  $S^1$  fiber is **k** times the homotopy class of the minicircle  $S_m^1$ . It is therefore clear that, in principle, we should be able to design a process in which a bulk *W*-boson reaches the tip of the cone and breaks-up into **k** strings that wrap the minicircle:

$$W \longrightarrow \mathbf{k}$$
 quasi-particles. (3.11)

Alternatively, it should be possible to describe the W-boson as a bound state of  $\mathbf{k}$  quasiparticles. In §3.4-§3.6, we will show how this works in the limit of large  $\mathbf{k}$ . Before we proceed to this analysis, which is the main focus of our paper, let us compute the spin quantum numbers of the quasi-particles.

### 3.3 Quasi-particles

The quasi-particle is obtained by wrapping the (2,0) BPS string on the minicircle  $S_m^1$ . Its quantum numbers can be deduced by quantization of the zero-modes of the low-energy fermions that live on the BPS string of the (2,0)-theory. Let us begin by reviewing the low-energy fermionic degrees of freedom on a BPS string. We assume that the M5-branes are in directions  $0, \ldots, 5$ , separated in direction 10, and the BPS string is in direction  $x_3$ . We first ignore the equivalence (3.2) and the R-symmetry twist. For simplicity we will now refer to rotation groups as SO(m) instead of Spin(m). Thus, the VEV breaks the R-symmetry to  $SO(4)_R \subset SO(5)_R$ , and the presence of the string breaks the Lorentz group down to  $SO(1,1) \times SO(4)$ . We will denote the last factor by  $SO(4)_T$ , and we will describe representations of  $SO(1,1) \times SO(4)_T \times SO(4)_R$  as  $(\mathbf{r_1}, \mathbf{r_2}, \mathbf{r_3}, \mathbf{r_4})_s$ , where  $(\mathbf{r_1}, \mathbf{r_2})$  is a representation of  $SO(4)_T \sim SU(2) \times SU(2)$ ,  $(\mathbf{r_3}, \mathbf{r_4})$  is a representation of  $SO(4)_R \sim SU(2) \times SU(2)$ , and  $\mathbf{s}$  is an SO(1,1) charge (spin). The representation of the unbroken supersymmetry charges is the same as the supersymmetry that is preserved by an M2-brane ending on an M5-brane. If the M2-brane is in directions 0, 3, 10 and the M5-brane is in directions 0, 1, 2, 3, 4, 5 then a preserved SUSY parameter  $\epsilon$  satisfies

$$\epsilon = \Gamma^{0123456789\natural} \epsilon = \Gamma^{03\natural} \epsilon = \Gamma^{012345} \epsilon, \qquad (3.12)$$

where we denote  $\natural \equiv 10$ , to avoid ambiguity. The SUSY parameter therefore transforms as

$$({f 2},{f 1},{f 2},{f 1})_{+rac{1}{2}}\oplus ({f 1},{f 2},{f 1},{f 2})_{-rac{1}{2}}$$
 ,

On the worldsheet of the BPS string there are 4 scalars  $X^A$  (A = 1, 2, 4, 5) that correspond to translations of the string in transverse directions. These are in the representation  $(2, 2, 1, 1)_0$ . In addition, there are fermions in

$$(\mathbf{1}, \mathbf{2}, \mathbf{2}, \mathbf{1})_{+\frac{1}{2}} \oplus (\mathbf{2}, \mathbf{1}, \mathbf{1}, \mathbf{2})_{-\frac{1}{2}}.$$
 (3.13)

Now, consider this theory on  $\mathbb{R}^{2,1} \times \mathbf{M}_3$  and let the BPS string be at rest at  $x_1 = x_2 = 0$ . It thus breaks the Lorentz group SO(2, 1) to the rotation group SO(2) in the  $x_1 - x_2$  plane, which we denote by  $SO(2)_J$ . The representations appearing in the brackets of (3.13) refer to  $SO(4)_T \times SO(4)_R$ , but in our setting, according to the discussion above, we have to reduce  $SO(4)_T \to SO(2)_J \times SO(2)_{\mathbf{z}}$  and  $SO(4)_R \to SO(2)_{\gamma} \times SO(2)_r$ . Thus, denoting representations as

$$(q_J, q_\mathbf{z}, q_\gamma, q_r)_s, \qquad (3.14)$$

we decompose the left-moving spinors of (3.13) as

$$(+\frac{1}{2},+\frac{1}{2},-\frac{1}{2},+\frac{1}{2})_{+\frac{1}{2}} \oplus (+\frac{1}{2},+\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2},-\frac{1}{2},+\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2},+\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}}$$

and the right-movers as

$$(+\frac{1}{2},-\frac{1}{2},+\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (+\frac{1}{2},-\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2})$$

These modes can be described by fermionic fields on the string worldsheet, which are functions of  $(x_0, x_3)$ . To get the quantum numbers of the lowest-energy multiplet we need to find the zero-modes of these fermionic fields. The twisted boundary conditions such as (3.8) also introduce twists on some of the modes (3.15)-(3.16). On a field  $\psi(x_0, x_3)$  with charges  $q_z$ and  $q_{\gamma}$ , these boundary conditions are

$$\psi(x_0, x_3 + 2\pi R) = \omega^{2(q_z + q_\gamma)} \psi(x_0, x_3).$$

The only zero modes are therefore of those modes with  $q_z + q_\gamma = 0$ . These have quantum numbers

$$(+\frac{1}{2},+\frac{1}{2},-\frac{1}{2},+\frac{1}{2})_{+\frac{1}{2}} \oplus (-\frac{1}{2},-\frac{1}{2},+\frac{1}{2},-\frac{1}{2})_{+\frac{1}{2}} \oplus (+\frac{1}{2},-\frac{1}{2},+\frac{1}{2},+\frac{1}{2})_{-\frac{1}{2}} \oplus (-\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2})_{-\frac{1}{2}}$$
(3.17)

Quantizing these modes gives a multiplet with quantum numbers

where the charges  $q_J^{(0)}$ ,  $q_z^{(0)}$ ,  $q_\gamma^{(0)}$ ,  $q_r^{(0)}$  still need to be determined. To determine them, consider the discrete symmetry  $\mathbb{Z}'_2$ , defined in §3.2.2. It preserves the setting and the BPS particle but does not commute with all the charges  $q_J$ ,  $q_z$ ,  $q_\gamma$ ,  $q_r$ . It acts on the charges as follows:

$$q_J \to q_J, \quad q_{\mathbf{z}} \to -q_{\mathbf{z}}, \quad q_{\gamma} \to -q_{\gamma}, \quad q_r \to q_r.$$
 [generator of  $\mathbb{Z}_2'$ ]

The constants  $q_J^{(0)}$ ,  $q_z^{(0)}$ ,  $q_\gamma^{(0)}$ ,  $q_r^{(0)}$  must therefore be chosen so that the charges (3.18) will be invariant, as a set, under  $\mathbb{Z}'_2$ . In other words,  $\mathbb{Z}'_2$  is allowed to permute the states in (3.18), but must convert an existing state to an existing state. This is only possible if both

 $q_{\mathbf{z}}^{(0)}$  and  $q_{\gamma}^{(0)}$  vanish. The BPS states are therefore in a multiplet with quantum numbers given by:

$$(q_J^{(0)} - \frac{1}{2}, 0, 0, q_r^{(0)} - \frac{1}{2}) \oplus (q_J^{(0)}, +\frac{1}{2}, -\frac{1}{2}, q_r^{(0)}) \oplus (q_J^{(0)}, -\frac{1}{2}, +\frac{1}{2}, q_r^{(0)}) \oplus (q_J^{(0)} + \frac{1}{2}, 0, 0, q_r^{(0)} + \frac{1}{2}).$$

Spin-statistics requires  $q_J^{(0)}$  and  $q_r^{(0)}$  to be half-integral. To determine them we note that the setting of (3.4) can be defined for any value of **k**, not necessarily an integer (as suggested in [34]). And so we can determine the quantum numbers from the limit  $\mathbf{k} \to \infty$  at which the multiplet must become part of the multiplet of the wrapped string of the (2,0)-theory. This determines the charges up to an overall sign (which can be determined arbitrarily and flipped with a parity transformation). So we pick  $q_J^{(0)} = -q_r^{(0)} = \frac{1}{2}$  and find the following multiplet structure:

$$(0,0,0,-1) \oplus \left(+\frac{1}{2},+\frac{1}{2},-\frac{1}{2},-\frac{1}{2}\right) \oplus \left(+\frac{1}{2},-\frac{1}{2},+\frac{1}{2},-\frac{1}{2}\right) \oplus \left(1,0,0,0\right).$$
(3.19)

As a corollary, we can immediately restrict the types of processes described in (3.11). Let us write down the  $q_J$ ,  $q_z$ ,  $q_\gamma$ , and  $q_r$  quantum numbers of the W-boson supermultiplet. The bosons (vectors and scalars) are in

$$(\pm 1, 0, 0, 0) \oplus (0, \pm 1, 0, 0) \oplus (0, 0, \pm 1, 0) \oplus (0, 0, 0, \pm 1).$$

$$(3.20)$$

and the gluinos are ingeom

$$(\pm \frac{1}{2}, \pm \frac{1}{2}, \pm \frac{1}{2}, \pm \frac{1}{2})$$
. [even number of  $(-\frac{1}{2})$ 's] (3.21)

Starting with a W-boson with charges (1, 0, 0, 0), a process such as

$$W$$
-boson  $\longrightarrow \mathbf{k}$  quasi-particles. (3.22)

can only produce **k** quasi-particles of charge (1, 0, 0, 0), and  $(\mathbf{k} - 1)$  units of orbital angular momentum need to convert into spin. We therefore expect that if the *W*-boson's velocity *u* in the  $x_1 - x_2$  plane is small, the amplitude will be suppressed as  $u^{\mathbf{k}-1}$ .

The process (3.22) also suggests that the W boson can be viewed as a bound state of **k** quasi-particles. This is similar to the well-known result in FQHE theory that the electron can be regarded as a bound state of **k** fractionally charged edge-states. The edge-states are the low-energy excitations of the Chern-Simons theory that reside on the boundary, or on impurities in the bulk. In the analogy the FQHE system, our quasi-particles correspond to external impurities that couple to the Chern-Simons theory gauge field.

Our goal is to develop a concrete description of the W-boson as a composite of  $\mathbf{k}$  quasiparticles. For this purpose we will first need to switch to a dual formulation of the low-energy theory whereby the quasi-particles are fundamental.

### 3.4 The large k limit

A weakly-coupled dual formulation of our system can be constructed in the limit  $\mathbf{k} \to \infty$ . In FQHE terminology, this is the *small filling fraction* regime which in ordinary systems corresponds to very strong interactions. More insight can be gained in this limit by choosing a different fibration structure for  $\mathbf{M}_3$  than the one represented in (3.5). While (3.5) is convenient to work with because the fibers are of constant size and are geodesics, the fibration was singular at the origin  $\mathbf{z} = 0$  — indeed the tip of the cone is singular, and the fiber over  $\mathbf{z} = 0$  is smaller by a factor of  $\mathbf{k}$  from the generic one.

Instead, in this section we will represent  $\mathbf{M}_3$  as a smooth fibration in another way. The base is the well-known cigar geometry and the fiber corresponds to a loop at constant  $|\mathbf{z}|$ . (See also [58][59] for other uses of this technique.) We will then reduce the (2,0)-theory to 4 + 1D SYM along this fiber. The fiber's size varies and the base's geometry is curved, but nevertheless this representation is very useful, as we shall see momentarily.

#### 3.4.1 Cigar geometry

To arrive at the the alternative fibration we change variables on  $\mathbf{M}_3$  from  $(x_3, \mathbf{z})$  to  $x_3$ and

$$\tilde{\mathbf{z}} \equiv \exp\left(-\frac{ix_3}{\mathbf{k}R}\right) \mathbf{z} \equiv r e^{i\overline{\theta}}$$
 (3.23)

We then write the metric as

$$ds^{2} = dx_{3}^{2} + |d\mathbf{z}|^{2} = \widetilde{\alpha}(dx_{3} + \frac{r^{2}}{\mathbf{k}R\widetilde{\alpha}}d\overline{\theta})^{2} + dr^{2} + \widetilde{\alpha}^{-1}r^{2}d\overline{\theta}^{2}, \qquad (\widetilde{\alpha} \equiv 1 + \frac{r^{2}}{\mathbf{k}^{2}R^{2}}) \qquad (3.24)$$

This metric describes a circle fibration over a cigar-like base with metric

$$ds_B^2 = dr^2 + \widetilde{\alpha}^{-1} r^2 d\overline{\theta}^2 = dr^2 + \left(\frac{\mathbf{k}^2 R^2 r^2}{\mathbf{k}^2 R^2 + r^2}\right) d\overline{\theta}^2.$$
(3.25)

We denote the cigar space by  $\Upsilon$ . Note that the cigar-metric is smooth everywhere and for  $r \gg \mathbf{k}R$  it behaves like a cylinder  $\mathbb{R}_+ \times S^1$ , where  $S^1$  has radius  $\mathbf{k}R$ . The "global angular form" of the circle fibration is

$$\chi \equiv dx_3 + \frac{r^2}{\mathbf{k}R\tilde{\alpha}}d\bar{\theta} \equiv dx_3 + R\mathfrak{a}\,, \qquad (3.26)$$

where we have defined the 1-form

$$\mathfrak{a} \equiv \frac{r^2}{\mathbf{k}R^2\widetilde{\alpha}}d\overline{\theta} = \left(\frac{\mathbf{k}r^2}{\mathbf{k}^2R^2 + r^2}\right)d\overline{\theta}\,.\tag{3.27}$$

In this context,  $\mathfrak{a}$  is a U(1) gauge field on the cigar with associated field-strength

$$d\chi \equiv d\mathfrak{a} = \frac{2\mathbf{k}^3 R^2 r}{(\mathbf{k}^2 R^2 + r^2)^2} dr \wedge d\overline{\theta} \,.$$



Figure 3.2: The cigar geometry with the typical scales indicated. The curvature of the cigar sets the length scale  $\mathbf{k}R$ , and the 4 + 1D SYM coupling constant sets the length scale  $g_{\rm vm}^2$ .

The total magnetic flux of this gauge field is  $\int_B d\mathbf{a} = 2\pi \mathbf{k}$ .

An anti-self-dual field  $H = -^{*}H$  on  $\mathbf{M}_{3} \times \mathbb{R}^{2,1}$  can be reduced along the fibers of the circle fibration (3.24) to obtain a 4 + 1D gauge field strength f on  $\Upsilon \times \mathbb{R}^{2,1}$  as follows:

$$H = \left(dx_3 + \frac{r^2}{\mathbf{k}R\widetilde{\alpha}}d\overline{\theta}\right) \wedge f - \widetilde{\alpha}^{-\frac{1}{2}}(^*f).$$
(3.28)

Here \*f is the 4+1D Hodge dual of the 2-form f on  $\Upsilon \times \mathbb{R}^{2,1}$ . The coupling constant of the effective 4 + 1D super Yang-Mills theory for f is

$$g_{\rm ym}^2 = \widetilde{\alpha}^{1/2} R = (1 + \frac{r^2}{\mathbf{k}^2 R^2})^{\frac{1}{2}} R.$$
 (3.29)

The coupling constant  $g_{ym}^2$  has dimensions of length and can be compared to the length scale set by the order of magnitude of the curvature of the cigar metric at the origin – this length-scale is  $\mathbf{k}R$ . For  $r \sim \mathbf{k}R$  we find  $g_{ym}^2 \ll \mathbf{k}R$ , and so the Yang-Mills theory is weakly coupled on length scales of the order of the curvature. The Yang-Mills theory becomes strongly coupled only when the two scales become comparable, which happens for  $r \sim \mathbf{k}^2 R$ , and therefore for large  $\mathbf{k}$  our low-energy semi-classical 4 + 1D SYM approximation is valid for  $r \ll \mathbf{k}^2 R$ . The various length scales are depicted in Figure 3.2.

#### 3.4.2 Equations of motion

The bosonic fields of maximally supersymmetric 4+1D SYM are the gauge field and 5 adjoint scalars. The scalars correspond to the relative motion of the M5-branes in directions  $x_6, \ldots, x_{10}$ . We will be interested in supersymmetric solutions where only the scalar

corresponding to direction  $x_{10}$  can be nonzero. We will therefore ignore the remaining 4 scalars, as well as the fermions, and denote the single adjoint-valued relevant scalar by  $\Phi$ . The boundary conditions at infinity are

$$\Phi \to \begin{pmatrix} v & 0\\ 0 & -v \end{pmatrix} \qquad (\text{up to a gauge transformation}),$$

where  $v \equiv 2\pi \tilde{V}$ , where  $\tilde{V}$  is the tension of the BPS string defined in §3.2.1.

We set polar coordinates in the  $x_1 - x_2$  plane by

$$\rho \equiv \sqrt{x_1^2 + x_2^2}, \qquad x_1 + ix_2 = \rho e^{i\varphi}.$$
(3.30)

The SYM theory is therefore formulated on a space with 4+1D metric

$$ds^2 = -dt^2 + dr^2 + \widetilde{\alpha}^{-1}r^2d\overline{\theta}^2 + d\rho^2 + \rho^2 d\varphi^2 \,.$$

The action contains three terms,

$$I_{\text{bosonic}} = I_{\Phi} + I_{\text{YM}} + I_{\theta} \,, \tag{3.31}$$

where  $I_{\Phi}$  is the action of the scalar field,  $I_{\text{YM}}$  is the standard Yang-Mills action with variable coupling constant, and  $I_{\theta}$  is the 4+1D  $\theta$ -term that arises due to the nonzero connection  $\mathfrak{a}$  (see (3.27)). We will only consider  $\overline{\theta}$ -independent field configurations. For such configurations the explicit expressions for the terms in the action are

$$I_{\Phi} = \frac{1}{2}R \operatorname{tr} \int \left[ (D_0 \Phi)^2 - (D_{\rho} \Phi)^2 - \frac{1}{\rho^2} (D_{\varphi} \Phi)^2 - (D_r \Phi)^2 \right] r \rho dr d\rho d\varphi dt , \qquad (3.32)$$

$$I_{\rm YM} = \frac{1}{2}R^{-1} \operatorname{tr} \int \frac{1}{\tilde{\alpha}} \left( F_{0r}^2 + F_{0\rho}^2 - F_{r\rho}^2 - \frac{1}{\rho^2}F_{r\varphi}^2 - \frac{1}{\rho^2}F_{\rho\varphi}^2 \right) r\rho dr d\rho d\varphi dt , \qquad (3.33)$$

$$I_{\theta} = \operatorname{tr} \int \frac{r^2}{\mathbf{k}R^2\tilde{\alpha}} \left( F_{0r}F_{\rho\varphi} - F_{0\rho}F_{r\varphi} + F_{0\varphi}F_{r\rho} \right) dr d\rho d\varphi dt , \qquad (3.34)$$

where  $D_{\mu}$  is the covariant derivative of an adjoint-valued field. The equations of motion are:

$$0 = D^{\beta} F_{\alpha\beta} + D_r F_{\alpha r} - \frac{1}{r} F_{\alpha r} - i \frac{r^2}{\mathbf{k}^2 R^2} [D_{\alpha} \Phi, \Phi], \qquad (3.35)$$

$$0 = D^{\beta} F_{r\beta} - i \frac{r^2}{\mathbf{k}^2 R^2} [D_r \Phi, \Phi], \qquad (3.36)$$

$$0 = D^{\alpha}D_{\alpha}\Phi + D_rD_r\Phi + \frac{1}{r}D_r\Phi.$$
(3.37)

where  $\alpha, \beta = 0, 1, 2$ .

### 3.5 Integrally charged particles as bound states of quasiparticles

We now have two alternative descriptions of the low-energy limit in terms of 4+1D SYM. In the first description, studied in §3.2, the 4+1D SYM theory is formulated on a cone, with



Figure 3.3: In the limit  $va^2 \gg 1$  the soliton is approximately described by the Prasad-Sommerfield solution (of width 1/va) near r = a and  $\rho = 0$ . Note that  $\rho = \sqrt{x_1^2 + x_2^2}$  and the directions  $x_1, x_2$  are not drawn since they are perpendicular to the  $r, \theta$  directions.

extra degrees of freedom at the tip. In the second description, studied in §3.4, the 4+1D SYM theory is formulated on a cigar geometry. The latter description is most suitable in the large **k** limit, because the strongly-coupled region is pushed to  $r = \infty$  (see Figure 3.2). The quasi-particles that we studied in §3.3 are the fundamental fields of 4+1D SYM in the cigar-setting. We have seen that **k** quasi-particles can form a bound state that is free to move into the bulk of the cone. Let us now identify this state in the cone-setting.

From the perspective of the (2, 0)-theory, the bound state is a string wrapped on the fiber of (3.5). Let us consider such a string wrapped on the fiber at the cone base point given by coordinates r = a and  $\theta = x_1 = x_2 = 0$ , with variable  $x_3$ . In the cigar variables, this reduces to a string at fixed r = a and  $x_1 = x_2 = 0$  but variable  $\overline{\theta}$ . Recall that on the Coulomb branch of SU(2) 4+1D SYM, the monopole is a 1+1D object – a monopole-string. The bound state of **k** quasi-particles is therefore associated with a monopole-string wrapped around the  $\overline{\theta}$ -circle of the cigar at r = a, as depicted in Figure 3.3. Thanks to the  $\theta$ -term (3.34), the monopole-string gains **k** units of charge, as required.

In flat space, a monopole-string is described by the Prasad-Sommerfield solution [36]. In our case, the Prasad-Sommerfield solution is a good approximation if the thickness of the monopole is small compared to the typical scale  $\mathbf{k}R$  over which the coupling constant varies, and also small compared to a. In this case, setting

$$\mathbf{w} \equiv \sqrt{(r-a)^2 + \rho^2} \,,$$

we find the gauge invariant magnitude of the scalar field near the core r = a to be given by

$$|\Phi| \equiv \sqrt{\operatorname{tr}(\Phi^2)/2} = \tilde{v} \operatorname{coth}(\tilde{v}\mathbf{w}) - \frac{1}{\mathbf{w}}, \qquad (3.38)$$

where

$$\tilde{v} = (1 + \frac{a^2}{\mathbf{k}^2 R^2})^{1/2} R v$$

is the effective VEV of the normalized scalar field  $\tilde{\alpha}^{1/2} R \Phi$  at the core (r = a) of the monopole. The "thickness" of the Prasad-Sommerfield solution is of the order of  $1/\tilde{v}$ , and the condition that the monopole should be "thin" becomes

$$a \gg \frac{1}{Rv}$$
.

If this condition is not met, the Prasad-Sommerfield solution does not provide a good approximation for the particle that corresponds to a (2,0)-string wrapped on the generic fiber (of size  $\mathbf{k}R$ ) of (3.5). Nevertheless, this is a BPS state with charge  $\mathbf{k}$ , which can be described in the large  $\mathbf{k}$  limit by a soliton solution to the equations of motion (3.35)-(3.37). The solution describes a Q-ball, and we expect the position a to be a modulus of the solution. In the next section we will derive the BPS equations that this soliton satisfies.

### 3.5.1 BPS equations

As we will derive in  $\S3.5.2$ , the BPS equations that describe static solutions that preserve the same amount of supersymmetry as a (2, 0)-string wrapped on a fiber of (3.5) are given by:

$$D_r \Phi = \frac{\mathbf{k}R}{r} F_{12} = F_{0r}, \qquad D_1 \Phi = \frac{\mathbf{k}R}{r} F_{2r} = F_{01}, \qquad D_2 \Phi = -\frac{\mathbf{k}R}{r} F_{1r} = F_{02}.$$
(3.39)

These equations imply the equations of motion (3.35)-(3.37). Assuming that  $A_r, A_1, A_2$  are time independent, we find  $D_{\mu}\Phi = F_{0\mu} = -D_{\mu}A_0$  (for  $\mu = 1, 2, r$ ), which is solved by  $\Phi = -A_0$ . So the equations are reduced to

$$D_r \Phi = \frac{\mathbf{k}R}{r} F_{12}, \qquad D_1 \Phi = \frac{\mathbf{k}R}{r} F_{2r}, \qquad D_2 \Phi = -\frac{\mathbf{k}R}{r} F_{1r}, \qquad \Phi = -A_0.$$
(3.40)

The nonzero  $A_0$  is consistent with a Q-ball [63]. It can be gauged away at the expense of creating time-varying phases for the other fields, but we will not do so. We can rewrite the first three equations of (3.40) as the Prasad-Sommerfield [36] equations

$$D_i \tilde{\Phi} = B_i \tag{3.41}$$

where

$$\widetilde{\Phi} \equiv \frac{1}{\mathbf{k}R} \Phi, \qquad B_i \equiv \frac{1}{2\sqrt{g}} g_{ij} \epsilon^{jkl} F_{kl},$$
(3.42)

are defined on a 3D auxiliary space  $\mathcal{W}$  parameterized by  $x_1, x_2, r$ , with metric  $g_{ij}$  given by

$$ds^{2} = g_{ij}dx^{i}dx^{j} = r^{2}(dr^{2} + dx_{1}^{2} + dx_{2}^{2}) = r^{2}(dr^{2} + d\rho^{2} + \rho^{2}d\varphi^{2}).$$
(3.43)

In §3.6.2 we will show that the problem of finding an axisymmetric ( $\varphi$ -independent) BPS soliton can be converted to the problem of finding a harmonic map from the  $AdS_3$  space with metric

$$ds^{2} = \frac{1}{r^{2}}(dr^{2} + d\rho^{2} + \rho^{2}d\varphi^{2})$$

to  $AdS_2$ , with a certain singular behavior along a Dirac-like string at  $\rho = 0$  and 0 < r < a.

We will conclude with a formula for the energy. For a static configuration, the energy is minus the integral of the Lagrangian that appears in the action (3.31). If we set  $A_0 = -\Phi$  the energy for static and  $\varphi$ -independent configurations can be written as

$$\mathcal{E}_{\text{static}} = \pi R \operatorname{tr} \int \frac{1}{\tilde{\alpha}} \Big[ F_{r\rho}^2 + \frac{1}{\rho^2} (F_{r\varphi} - \frac{r\rho}{\mathbf{k}R} D_\rho \Phi)^2 + \frac{1}{\rho^2} (F_{\rho\varphi} - \frac{r\rho}{\mathbf{k}R} D_r \Phi)^2 \Big] r\rho dr d\rho + \pi \operatorname{tr} \int \Big[ \partial_\rho \Big( \frac{r^2}{\mathbf{k}\tilde{\alpha}} F_{r\varphi} \Phi \Big) + \partial_r \Big( \frac{r^2}{\mathbf{k}\tilde{\alpha}} F_{\rho\varphi} \Phi \Big) \Big] dr d\rho$$
(3.44)

The term on the RHS vanishes when the BPS equations (3.40) are satisfied, and the second line is a total derivative.

#### 3.5.2 Derivation of the BPS equations

In this subsection we will explain how (3.39) were derived. (The rest of the paper does not rely on this subsection, and it can be skipped.) We wish to find the equations that describe the W-boson from (3.22) in terms of the low-energy fields of 4+1D SYM on  $\Upsilon \times \mathbb{R}^{2,1}$ , where  $\mathbb{R}^{2,1}$  corresponds to directions 0, 1, 2, and  $\Upsilon$  was defined in §3.4.1. Let us first discuss the equations on the Coulomb branch of the (2,0)-theory. The contents of the low-energy theory is a free tensor multiplet with 2-form field *B*, field strength H = dB, five scalar fields  $\Phi^6, \ldots, \Phi^{10}$ , and chiral fermionic fields  $\psi$  in the representation of  $\mathbf{4} \times \mathbf{4}$  of  $SO(5, 1) \times SO(5)$ . We assume

$$\Phi^{6} = \Phi^{7} = \Phi^{8} = \Phi^{9} = 0$$

and only allow  $\Phi^{10} \equiv \phi$  to be nonzero. The BPS equations are derived from the SUSY transformation of the fermions. Let  $\epsilon$  be a constant SUSY parameter, which we represent as a 32-component spinor on which the 10+1D Dirac matrices  $\Gamma^{I}$  (I = 0, ..., 10) can act. The BPS conditions on  $\epsilon$  are:

- Invariance of  $\epsilon$  under simultaneous rotations by  $2\pi/\mathbf{k}$  in the planes 4-5 and 6-7;
- Invariance of an M5-brane in direction  $0, \ldots, 5$  under a SUSY transformation of 10+1D SUGRA with parameter  $\epsilon$ ; and
- Invariance of an M2-brane in directions 0, 3, 10 under a SUSY transformation of 10+1D SUGRA with parameter  $\epsilon$ .

Therefore, the equations are (we set  $10 \equiv \ddagger$  in Dirac matrices):

$$\epsilon = \Gamma^{012345} \epsilon = \Gamma^{03\natural} \epsilon = \Gamma^{4567} \epsilon \,. \tag{3.45}$$

To get the BPS equations we require that the fermions  $\psi$  of the tensor multiplet of the (2,0)-theory be invariant under any SUSY transformation with a parameter  $\epsilon$  that satisfies (3.45):

$$0 = \delta \psi = (H_{\mu\nu\sigma}\Gamma^{\mu\nu\sigma} - \partial_{\mu}\phi\Gamma^{\mu\natural})\epsilon.$$
(3.46)

There are four linearly independent solutions to (3.45), and substituting these into (3.46) we find the BPS equations in the form

$$H_{03\mu} = \partial_{\mu}\phi$$
,  $H_{0ij} = 0$ ,  $(i, j = 1, 2, 4, 5)$ . (3.47)

The other components of H are determined by anti-self-duality  $H = -^*H$ . We now convert the BPS equations (3.47) to  $\Upsilon \times \mathbb{R}^{2,1}$  using (3.28) and the change of variables (3.23). To avoid ambiguity, we momentarily denote by  $x'_3$  and  $\theta'$  the coordinates before the change of variables, so that the change of variables is given by

$$x_3 = x'_3, \qquad \overline{ heta} = heta' - rac{x'_3}{\mathbf{k}R}.$$

We then find:

$$0 = H_{03'r} - \partial_r \phi = H_{03'\theta'} - \partial_{\theta'} \phi = \partial_{3'} \phi = \partial_0 \phi, \qquad 0 = H_{03'i} - \partial_i \phi, \qquad (i = 1, 2), \quad (3.48)$$

and

$$0 = H_{012} = H_{0ir} = H_{0i\theta'} = H_{0i\bar{\theta}} = H_{0r\bar{\theta}} = H_{0r\bar{\theta}}, \qquad (i = 1, 2).$$

The dual relations are

$$0 = H_{3'r\theta'} = H_{3'i\theta'} = H_{3'ir} = H_{3'12}, \qquad (i = 1, 2),$$

which are transformed in the  $x_3, \overline{\theta}$  coordinates to

$$0 = H_{3r\bar{\theta}} = H_{3i\bar{\theta}} = H_{3ir} - \frac{1}{\mathbf{k}R} H_{\bar{\theta}ir} = H_{312} - \frac{1}{\mathbf{k}R} H_{\bar{\theta}12} , \qquad (i = 1, 2) .$$
(3.49)

Next we use the anti-self-duality conditions

$$H_{03'r} = \frac{1}{r} H_{\theta'12} = \frac{1}{r} H_{\bar{\theta}12}, \qquad H_{03'1} = \frac{1}{r} H_{r\theta'2} = \frac{1}{r} H_{r\bar{\theta}2}, \qquad H_{03'2} = -\frac{1}{r} H_{r\theta'1} = -\frac{1}{r} H_{r\bar{\theta}1},$$

and the relations (3.49) to write

$$H_{03'r} = \frac{1}{r} H_{\bar{\theta}12} = \frac{\mathbf{k}R}{r} H_{312}, \quad H_{03'1} = \frac{1}{r} H_{r\bar{\theta}2} = \frac{\mathbf{k}R}{r} H_{32r}, \quad H_{03'2} = -\frac{1}{r} H_{r\bar{\theta}1} = -\frac{\mathbf{k}R}{r} H_{31r}.$$
(3.50)

Combining with (3.48), we end up with the BPS equations

$$\partial_r \phi = H_{03'r} = \frac{\mathbf{k}R}{r} H_{312}, \qquad \partial_1 \phi = H_{03'1} = \frac{\mathbf{k}R}{r} H_{32r}, \qquad \partial_2 \phi = H_{03'2} = -\frac{\mathbf{k}R}{r} H_{31r}, \quad (3.51)$$

and further combining with (3.28) we have

$$\partial_r \phi = \frac{\mathbf{k}R}{r} f_{12}, \qquad \partial_1 \phi = \frac{\mathbf{k}R}{r} f_{2r}, \qquad \partial_2 \phi = -\frac{\mathbf{k}R}{r} f_{1r}. \tag{3.52}$$

Altogether, we have

$$\partial_r \phi = \frac{\mathbf{k}R}{r} f_{12} = f_{0r}, \qquad \partial_1 \phi = \frac{\mathbf{k}R}{r} f_{2r} = f_{01}, \qquad \partial_2 \phi = -\frac{\mathbf{k}R}{r} f_{1r} = f_{02}.$$
 (3.53)

The equations (3.39) are the nonabelian extension of (3.53), and the fact that they imply the equations of motion (3.35)-(3.37) shows that no additional terms are needed.

### 3.6 Analysis of the BPS equations

We are looking for a solution to the BPS equations (3.40) that describes the (2,0)-string wrapped on the fiber of (3.5). We can assume that the string is at the origin of the  $x_1 - x_2$ coordinate system, and the solution will therefore be axisymmetric (independent of  $\varphi$ ). We expect the solution to have a modulus *a* corresponding to the position of the string in the  $x_4 - x_5$  plane. Technically, these solutions could only describe a string that is "smeared" along the angular coordinate  $\theta$  of the  $x_4 - x_5$  plane, so only the *r* coordinate of the string is fixed. The soliton is spread out in the *r* direction as well, but we can expect its core to be around  $r \sim a$ , and we will see that *a* can be defined via the boundary conditions at  $r \to \infty$  or  $\rho \to \infty$ . More generally, we can look for axisymmetric solutions that describe several (2,0)-strings centered at different *r* locations, but all at  $\rho = 0$ . This will be described by a more complicated axisymmetric solution of (3.40). To proceed, we will treat the BPS equations as Bogomolnyi-Prasad-Sommerfield monopole equations (3.41) on a curved space (3.43).

#### 3.6.1 Manton gauge

The Bogomolnyi monopole equations on  $\mathbb{R}^3$  have the renowned Prasad-Sommerfield solution [36] for one SU(2) monopole, and the general solution was given by Nahm [72]. It was given a string-theoretic interpretation in [39]. The extension to hyperbolic space is known [43], but we are unaware of an extension of Nahm's technique to the space given by the metric (3.43), and so we will proceed using other means. We adopt a remarkable ansatz developed in [45] for axially symmetric solutions. Adapted from  $\mathbb{R}^3$  to our metric (3.43) we make the Ansatz:

$$\Phi = \frac{1}{2}(\Phi_1 \sigma_1 + \Phi_2 \sigma_2), \qquad A = -[(\eta_1 \sigma_1 + \eta_2 \sigma_2)d\varphi + W_2 \sigma_3 d\rho + W_1 \sigma_3 dr], \qquad (3.54)$$

The BPS equations then reduce to

$$\partial_{\rho} \Phi_1 - W_2 \Phi_2 = -\frac{1}{r_{\rho}} (\partial_r \eta_1 - W_1 \eta_2), \qquad (3.55)$$

$$\partial_{\rho}\Phi_{2} + W_{2}\Phi_{1} = -\frac{1}{r\rho}(\partial_{r}\eta_{2} + W_{1}\eta_{1}), \qquad (3.56)$$

$$\eta_2 \Phi_1 - \eta_1 \Phi_2 = \frac{\rho}{r} (\partial_{\rho} W_1 - \partial_r W_2) \,, \tag{3.57}$$

$$\partial_r \Phi_1 - W_1 \Phi_2 = \frac{1}{r\rho} (\partial_\rho \eta_1 - W_2 \eta_2),$$
 (3.58)

$$\partial_r \Phi_2 + W_1 \Phi_1 = \frac{1}{r\rho} (\partial_\rho \eta_2 + W_2 \eta_1),$$
 (3.59)

Next, we adapt to our metric the technique developed in [41], solving (3.55)-(3.57) by setting

$$\Phi_1 = -\frac{1}{r} f^{-1} \partial_r \chi \,, \quad \Phi_2 = \frac{1}{r} f^{-1} \partial_r f \,, \quad \eta_1 = \rho f^{-1} \partial_\rho \chi \,, \quad \eta_2 = -\rho f^{-1} \partial_\rho f \,, \tag{3.60}$$

and

$$W_1 = -f^{-1}\partial_r \chi, \quad W_2 = -f^{-1}\partial_\rho \chi.$$
 (3.61)

where f and  $\chi$  are as yet undetermined real functions of r and  $\rho$ .

### **3.6.2** Harmonic maps from $AdS_3$ to $AdS_2$

We plug the ansatz (3.60)-(3.61) into (3.58)-(3.59) and get:

$$0 = f\chi_{rr} + f\chi_{\rho\rho} - 2f_r\chi_r - 2f_\rho\chi_\rho + \frac{1}{\rho}f\chi_\rho - \frac{1}{r}f\chi_r , \qquad (3.62)$$

$$0 = f_r^2 + f_\rho^2 - \chi_r^2 - \chi_\rho^2 - f f_{rr} - f f_{\rho\rho} + \frac{1}{r} f f_r - \frac{1}{\rho} f f_\rho , \qquad (3.63)$$

where subscripts  $(\cdots)_r$  and  $(\cdots)_{\rho}$  denote derivatives with respect to r and  $\rho$ , respectively.

The equations (3.62)-(3.63) can be derived from the action

$$I = \int \frac{\rho}{rf^2} (f_r^2 + f_{\rho}^2 + \chi_r^2 + \chi_{\rho}^2) dr d\rho \,. \tag{3.64}$$

We give a simple geometrical interpretation to the equations of motion (3.62)-(3.63) by considering an auxiliary  $AdS_3$  space parameterized by  $(r, \rho, \varphi)$  with metric

$$ds^{2} = \frac{1}{r^{2}}(dr^{2} + d\rho^{2} + \rho^{2}d\varphi^{2})$$

We can then interpret the function  $f(r, \rho)$  and  $\chi(r, \rho)$  as describing an axisymmetric map from  $AdS_3$  to the two-dimensional  $(f, \chi)$  "target-space." If we further endow this target-space with the  $AdS_2$  metric

$$ds^{2} = \frac{1}{f^{2}} (df^{2} + d\chi^{2}), \qquad (3.65)$$

it is then easy to see that the equations of motion derived from (3.64) describe harmonic maps

$$(f,\chi): AdS_3 \mapsto AdS_2. \tag{3.66}$$

The connection between  $AdS_2$  (the "pseudosphere") and axisymmetric solutions to monopole equations on  $\mathbb{R}^3$  was first noted in [41]. The harmonic map (3.66) is required to have a singularity along a Dirac-like string. To see this we first need to discuss the asymptotic behavior of the maps far away from the core, where the solution reduces to a U(1) monopole.

#### 3.6.3 The abelian solution

We can find a special solution to (3.62)-(3.63) by setting  $\chi = 0$ . The remaining equation (3.63) then states that log f is a harmonic map on  $AdS_3$ . Alternatively, the solution describes a U(1) monpole on the  $(x_1, x_2, r)$  space with metric (3.43), centered at (0, 0, a), which becomes a singular point. But it is easiest to construct the solution starting with 5+1D. In the abelian limit, the fields of the (2, 0) theory that are relevant to our problem reduce to a free anti-self-dual 3-form field  $H = -^*H$  and a free scalar field  $\phi$ . We start by solving (3.47) on  $\mathbb{R}^{5,1}$ , which in particular implies that  $\phi$  is harmonic. Consider a solution that describes the H and  $\phi$  fields that emanate from a (2, 0)-string centered at

$$(x_1, x_2, x_4, x_5) = (0, 0, a \cos \theta, a \sin \theta).$$

The scalar field is given by

$$\phi = v + \frac{1}{x_1^2 + x_2^2 + (x_4 - a\cos\theta)^2 + (x_5 - a\sin\theta)^2}.$$
(3.67)

But the solution that we need must be independent of  $\theta$ . We can, however, obtain it from (3.67) by "smearing":

$$\phi(x_1, x_2, r) = v + \frac{1}{2\pi} \int_0^{2\pi} \frac{d\theta}{\rho^2 + (r\cos\theta - a)^2 + r^2\sin^2\theta} = \frac{1}{\sqrt{(\rho^2 + r^2 + a^2)^2 - 4a^2r^2}} \quad (3.68)$$

and it is not hard to check that

$$A = \left(\frac{\rho^2 + a^2 - r^2}{2\sqrt{(\rho^2 + r^2 + a^2)^2 - 4a^2r^2}} - 1\right)\frac{x_2dx_1 - x_1dx_2}{\rho^2}.$$
(3.69)

The solution corresponds to

$$f = \exp \int \phi(r,\rho) r dr = e^{-\frac{1}{2}vr^2} \left(\rho^2 + r^2 - a^2 + \sqrt{(\rho^2 + r^2 + a^2)^2 - 4a^2r^2}\right).$$
(3.70)

To explore the Dirac string singularity, it is convenient to use instead of the Poincaré coordinates on  $AdS_3$ , a coordinate system with the point r = a at the origin, via the coordinate transformation,

$$\frac{\rho}{r} = \sinh\chi\,\sin\alpha\,,\quad \frac{\rho^2 + r^2 - a^2}{2ar} = \sinh\chi\,\cos\alpha\,,\qquad ds^2 = d\chi^2 + \sinh^2\chi(d\alpha^2 + \sin^2\alpha\,d\varphi^2),$$

with the inverse transformation

$$r = a \frac{\cosh \chi + \sinh \chi \cos \alpha}{1 + \sinh^2 \chi \sin^2 \alpha}, \qquad \rho = a \frac{\cosh \chi + \sinh \chi \cos \alpha}{1 + \sinh^2 \chi \sin^2 \alpha} \sinh \chi \sin \alpha.$$

In these coordinates we have, up to an unimportant constant,

$$\log f = -\frac{1}{2}va^2 \left(\frac{\cosh\chi + \sinh\chi\cos\alpha}{1 + \sinh^2\chi\sin^2\alpha}\right)^2 + \log\left(\frac{\cosh\chi + \sinh\chi\cos\alpha}{1 + \sinh^2\chi\sin^2\alpha}\right) + \log\sinh\chi + \log(1 + \cos\alpha).$$
(3.71)

The singularity in the last term at  $\alpha = \pi$  represents the Dirac string. The abelian solution must describe the asymptotic behavior of the nonabelian solution when either  $r \to \infty$  or  $\rho \to \infty$  (or both).

#### 3.6.4 Comments on (lack of) integrability

The classic BPS equations for monopoles on  $\mathbb{R}^3$  admit the well-known Nahm solutions [72], which also have a nice string-theoretic interpretation [39]. The rich properties of these solutions essentially stem from an underlying integrable structure. One way to describe the structure is to map a solution of the BPS equations to a holomorphic vector bundle over minitwistor space [37, 38]. (Minitwistor space is the space of oriented straight lines on  $\mathbb{R}^3$  and it has a complex structure.) The BPS equations arise as the integrability condition for an auxiliary set of equations in terms of an auxiliary 2-component field  $\psi$ , that require  $\psi$ 's gaugecovariant derivative along a line in  $\mathbb{R}^3$  to be related to multiplication by the scalar field  $\tilde{\Phi}$ , and also require  $\psi$  to be holomorphic in the directions transverse to the line. This technique can be extended to other metrics, such as  $AdS_3$  (whose corresponding minitwistor space also possesses a complex structure and is equivalent to  $\mathbb{CP}^1 \times \mathbb{CP}^1$ ). But this technique fails for the metric (3.43), whose space of geodesics is not complex, and the monopole equations (3.41) cannot be expressed as the integrability condition for an auxiliary system of linear differential equations, at least not in an obvious way.

Another way to see where integrability fails is to focus on axially-symmetric solutions as in [41]. Defining the symmetric  $SL(2, \mathbb{R})$  matrix

$$\mathcal{G} \equiv \frac{1}{f} \begin{pmatrix} 1 & -\chi \\ -\chi & (f^2 + \chi^2) \end{pmatrix} \,,$$

the equations of motion (3.62)-(3.63) can then be recast as

$$0 = \nabla^{\alpha} (\nabla_{\alpha} \mathcal{G} \mathcal{G}^{-1}), \qquad (3.72)$$

where the covariant derivatives are with respect to another auxiliary metric,

$$ds^{2} = dr^{2} + d\rho^{2} + \left(\frac{\rho^{2}}{r^{2}}\right)d\varphi^{2}, \qquad (3.73)$$

and  $\mathcal{G}(r,\rho)$  is, of course, assumed to be independent of  $\varphi$ . It is possible [41] to recast axially symmetric solutions of the BPS equations on  $\mathbb{R}^3$  in the form (3.72) – the metric in that case would be the Euclidean metric

$$ds^2 = dr^2 + d\rho^2 + \rho^2 d\varphi^2$$

and the connection with the  $\sigma$ -model (3.72) leads to an integrable structure. To describe the integrable structure we switch to complex coordinates,

$$\xi \equiv r + i\rho \,, \qquad \overline{\xi} \equiv r - i\rho \,,$$

and write (3.72) as the integrability condition for a system of first order linear differential equations for a two-component field  $\Psi(\xi, \overline{\xi})$ :

$$\Psi_{\xi} = \frac{1}{1+\gamma} \mathcal{G}_{\xi} \mathcal{G}^{-1} \Psi, \qquad \Psi_{\overline{\xi}} = \frac{1}{1-\gamma} \mathcal{G}_{\overline{\xi}} \mathcal{G}^{-1} \Psi,$$

where  $(\cdots)_{\xi}$  and  $(\cdots)_{\overline{\xi}}$  are derivatives with respect to  $\xi$  and  $\overline{\xi}$ , and the function  $\gamma(\xi, \overline{\xi})$  has to be suitably chosen (so that the integrability condition  $(\Psi_{\xi})_{\overline{\xi}} = (\Psi_{\xi})_{\overline{\xi}}$  will be automatically satisfied). There are, in fact, infinitely many choices for the function  $\gamma$ , but it has to be a solution of

$$\gamma_{\xi} = \frac{\gamma}{\xi - \overline{\xi}} \left( \frac{1 + \gamma}{1 - \gamma} \right), \qquad \gamma_{\overline{\xi}} = -\frac{\gamma}{\xi - \overline{\xi}} \left( \frac{1 - \gamma}{1 + \gamma} \right),$$

which are compatible (see [42] for review). This construction is easy to extend to any metric of the form

$$ds^2 = dr^2 + d\rho^2 + \Lambda(r,\rho)^2 d\varphi^2 \,,$$

as long as  $\Lambda(r, \rho)$  is harmonic (in the metric  $dr^2 + d\rho^2$ ). In our case  $\Lambda = \rho/r$  is not harmonic, so the standard integrability structure is not present.

One can also attempt to extend the technique of [39], to "probe" the solution with a string that extends in an extra dimension, say  $x_8$ . It is not hard to construct BPS string solutions that preserve some supersymmetry, compatible with that of the M5-branes and the twist. For example, in the M-theory variables we can take an M2-brane along a holomorphic curve given by

$$x_4 + ix_5 = C_0 e^{\frac{i}{\mathbf{k}R}(x_3 + ix_8)}$$

where  $C_0$  is a constant. This would translate in type-IIA to a string whose  $x_8$  coordinate varies logarithmically with r. However, this string does not preserve any common SUSY with the soliton. We were unable to find an exact solution to (3.41), and in fact, the appearance of polylogarithms in the expansion at large VEV (see Appendix D) suggests that even if a closed form exists, it is very complicated. We will therefore proceed to numerical analysis.

### 3.7 Numerical results

As a first step, we find it convenient to recast the equations in a different gauge. We begin by parameterizing the scalar field as:

$$\phi^{\alpha} = x_{\alpha}(\mathbf{f} + \mathbf{h}), \quad \phi^{3} = \mathbf{g}, \qquad (3.74)$$

and the gauge field as:

$$A^{\alpha}_{\beta} = x_{(\beta}\epsilon_{\alpha)\gamma}x^{\gamma}\mathbf{p} + \frac{1}{2}\epsilon_{\alpha\beta}\mathbf{q}, \quad A^{\alpha}_{r} = -r\epsilon_{\alpha\gamma}x_{\gamma}(\mathbf{f} - \mathbf{h}), \quad A^{3}_{\beta} = \epsilon_{\beta\gamma}x^{\gamma}\mathbf{v}, \quad A^{3}_{r} = 0.$$
(3.75)

with  $\alpha, \beta, \gamma = 1, 2, \epsilon_{\alpha\beta}$  being the anti-symmetric Levi-Civita symbol, and with  $\mathbf{f}, \mathbf{g}, \mathbf{h}, \mathbf{p}, \mathbf{q}$ , and  $\mathbf{v}$  functions of  $(r, \rho)$  only. Next, we fix the gauge by setting  $\mathbf{p} = 0$ . Defining

$$\mathbf{U} \equiv \rho^2 \,, \qquad \mathbf{V} \equiv r^2,$$

the BPS equations (3.41) reduce (after rescaling  $\phi$  by  $\mathbf{k}R$ ) to:

$$0 = \mathbf{h}\mathbf{v} - 2\frac{\partial\mathbf{h}}{\partial\mathbf{U}}\,,\tag{3.76}$$

$$0 = 2\mathbf{U}\frac{\partial \mathbf{f}}{\partial \mathbf{U}} + \mathbf{U}\mathbf{v}\mathbf{f} + 2\mathbf{f} + \frac{\partial \mathbf{q}}{\partial \mathbf{V}} + \frac{1}{2}\mathbf{g}\mathbf{q}, \qquad (3.77)$$

$$0 = \mathbf{V}(\mathbf{h} - \mathbf{f})\mathbf{g} + \frac{1}{2}\mathbf{q}\mathbf{v} + 2\mathbf{V}\frac{\partial\mathbf{f}}{\partial\mathbf{V}} + 2\mathbf{V}\frac{\partial\mathbf{h}}{\partial\mathbf{V}} - \frac{\partial\mathbf{q}}{\partial\mathbf{U}}, \qquad (3.78)$$

$$0 = \frac{\partial \mathbf{v}}{\partial \mathbf{V}} - \frac{\partial \mathbf{g}}{\partial \mathbf{U}} + \frac{1}{2}\mathbf{h}\mathbf{q}\,,\tag{3.79}$$

$$0 = \mathbf{U}\mathbf{V}(\mathbf{f}^2 - \mathbf{h}^2) + \frac{1}{4}\mathbf{q}^2 + 2\mathbf{v} + 2\mathbf{V}\frac{\partial \mathbf{g}}{\partial \mathbf{V}} + 2\mathbf{U}\frac{\partial \mathbf{v}}{\partial \mathbf{U}}.$$
 (3.80)

Let us also set

$$\mathbf{Z} \equiv \frac{1}{2a}(\rho^2 + r^2 - a^2), \qquad \mathbf{R} \equiv \sqrt{\rho^2 + \mathbf{Z}^2} = \frac{1}{2a}\sqrt{(\rho^2 + r^2 - a^2)^2 + 4a^2\rho^2}.$$
(3.81)

The advantage of the ansatz (3.74)-(3.75) is that the abelian solution (3.68)-(3.69) can be written in the form:

$$\mathbf{f} = \frac{v}{2\mathbf{R}} - \frac{1}{a\mathbf{R}^2}, \quad \mathbf{g} = \frac{v\mathbf{Z}}{\mathbf{R}} - \frac{\mathbf{Z}}{a\mathbf{R}^2}, \quad \mathbf{h} = \frac{v}{2\mathbf{R}}, \quad \mathbf{q} = \frac{a^2 + \mathbf{U} - \mathbf{V}}{a\mathbf{R}^2}, \quad \mathbf{v} = -\frac{1}{\mathbf{R}^2} - \frac{\mathbf{Z}}{a\mathbf{R}^2}.$$
(3.82)

which has no singularities except at r = a (and in particular no Dirac string).

We require that at either limit  $r \to \infty$  or  $\rho \to \infty$  the solution should reduce to the abelian solution. At the tip r = 0 the solution is required to be regular. This allows us to determine **q**, **h**, and **v** at the tip as follows. Setting **V** = 0 in (3.76), (3.78), and (3.80), we get the ordinary differential equations

$$\mathbf{h}\mathbf{v} - 2\frac{\partial \mathbf{h}}{\partial \mathbf{U}} = \frac{1}{2}\mathbf{q}\mathbf{v} - \frac{\partial \mathbf{q}}{\partial \mathbf{U}} = \frac{1}{4}\mathbf{q}^2 + 2\mathbf{v} + 2\mathbf{U}\frac{\partial \mathbf{v}}{\partial \mathbf{U}} = 0, \qquad (\mathbf{V} = 0)$$
(3.83)

which we can solve uniquely, given the known boundary conditions at  $\mathbf{U} \to \infty$ , by expressing  $\mathbf{q}$  and  $\mathbf{h}$  in terms of the function  $(1 + \mathbf{U}\mathbf{v})$  and its derivatives, and changing variables to

log U. The result is that unique solution to (3.83) that satisfies the boundary conditions at  $U \rightarrow \infty$  is

$$\mathbf{q} = \frac{4a}{\mathbf{U} + a^2}, \qquad \mathbf{v} = -\frac{2}{\mathbf{U} + a^2}, \qquad \mathbf{h} = \frac{va}{\mathbf{U} + a^2}, \qquad (\mathbf{V} = 0). \tag{3.84}$$

which is non other than the abelian solution (3.82) at  $\mathbf{V} = 0$ .

We cannot determine  $\mathbf{f}$  and  $\mathbf{g}$  at  $\mathbf{V} = 0$  so easily, and our strategy will be to find an approximate solution to (3.76)-(3.80) by the variational method, minimizing the energy of the field configuration within a certain class of trial functions of  $(\mathbf{U}, \mathbf{V})$ . For the energy we take the expression for the excess energy above the BPS bound for a static configuration of gauge field and minimally coupled adjoint scalar on a manifold given by the three dimensional metric (3.43):

$$\mathcal{E} \equiv \frac{1}{2} \operatorname{tr} \int \sqrt{g} g^{ij} (D_i \widetilde{\Phi} - B_i) (D_j \widetilde{\Phi} - B_j) d^3 x$$
  
$$= \frac{1}{2} \operatorname{tr} \int \left[ (r D_r \widetilde{\Phi} - F_{12})^2 + (r D_1 \widetilde{\Phi} - F_{2r})^2 + (r D_2 \widetilde{\Phi} - F_{r1})^2 \right] \rho d\rho (\frac{dr}{r}), \qquad (3.85)$$

where  $B_i$  and  $\tilde{\Phi}$  were defined in (3.42), and the "tr" is in the fundamental representation. Note that  $\mathcal{E}$  is different from the physical energy (3.44). The integrand in (3.85) is  $\tilde{\alpha}/r^2$  bigger than the integrand in the first term on the RHS of (3.44), but they are both minimized on the BPS configurations, and (3.85) gives more weight to the vicinity of r = 0. We can rewrite  $\mathcal{E}$  in terms of the right-hand-sides of (3.76)-(3.80),

$$\mathcal{X}_1 = \mathbf{h}\mathbf{v} - 2\frac{\partial \mathbf{h}}{\partial \mathbf{U}}, \qquad (3.86)$$

$$\mathcal{X}_2 = 2\mathbf{U}\frac{\partial \mathbf{f}}{\partial \mathbf{U}} + \mathbf{U}\mathbf{v}\mathbf{f} + 2\mathbf{f} + \frac{\partial \mathbf{q}}{\partial \mathbf{V}} + \frac{1}{2}\mathbf{g}\mathbf{q}, \qquad (3.87)$$

$$\mathcal{X}_{3} = \mathbf{V}(\mathbf{h} - \mathbf{f})\mathbf{g} + \frac{1}{2}\mathbf{q}\mathbf{v} + 2\mathbf{V}\frac{\partial\mathbf{f}}{\partial\mathbf{V}} + 2\mathbf{V}\frac{\partial\mathbf{h}}{\partial\mathbf{V}} - \frac{\partial\mathbf{q}}{\partial\mathbf{U}}, \qquad (3.88)$$

$$\mathcal{X}_4 = \frac{\partial \mathbf{v}}{\partial \mathbf{V}} - \frac{\partial \mathbf{g}}{\partial \mathbf{U}} + \frac{1}{2} \mathbf{h} \mathbf{q} \,, \tag{3.89}$$

$$\mathcal{X}_5 = \mathbf{U}\mathbf{V}(\mathbf{f}^2 - \mathbf{h}^2) + \frac{1}{4}\mathbf{q}^2 + 2\mathbf{v} + 2\mathbf{V}\frac{\partial \mathbf{g}}{\partial \mathbf{V}} + 2\mathbf{U}\frac{\partial \mathbf{v}}{\partial \mathbf{U}}, \qquad (3.90)$$

as

$$\mathcal{E} = \int \left(\frac{1}{8}\mathbf{U}^2 \mathcal{X}_1^2 + \frac{1}{8}\mathcal{X}_2^2 + \frac{\mathcal{X}_3^2}{16\mathbf{V}} + \frac{1}{4}\mathbf{U}\mathcal{X}_4^2 + \frac{\mathcal{X}_5^2}{16\mathbf{V}}\right) d\mathbf{U}d\mathbf{V}.$$
 (3.91)

We also note that the BPS bound on energy is given by

$$\mathcal{E}_{\text{BPS}} = \operatorname{tr} \int \sqrt{g} g^{ij} (B_j D_i \widetilde{\Phi}) d^3 x$$
  
=  $\operatorname{tr} \int \left[ F_{12} D_r \widetilde{\Phi} + F_{2r} D_1 \widetilde{\Phi} + F_{r1} D_2 \widetilde{\Phi} \right] \rho d\rho dr = \int d\lambda , \qquad (3.92)$ 

where the 1-form  $\lambda$  is defined by

$$\lambda \equiv \left[\frac{1}{8}\mathbf{U}\mathbf{q}\mathbf{v}(\mathbf{f}+\mathbf{h}) + \frac{1}{16}\mathbf{q}^{2}\mathbf{g} + \frac{1}{2}\mathbf{v}\mathbf{g} + \frac{1}{2}\mathbf{U}\mathbf{g}\frac{\partial\mathbf{v}}{\partial\mathbf{U}} - \frac{1}{4}\mathbf{U}(\mathbf{f}+\mathbf{h})\frac{\partial\mathbf{q}}{\partial\mathbf{U}}\right]d\mathbf{U} + \left[\frac{1}{8}\mathbf{U}\mathbf{g}\mathbf{q}(\mathbf{h}-\mathbf{f}) + \frac{1}{4}\mathbf{U}(\mathbf{h}^{2}-\mathbf{f}^{2})(1+\mathbf{U}\mathbf{v}) - \frac{1}{4}\mathbf{U}(\mathbf{f}+\mathbf{h})\frac{\partial\mathbf{q}}{\partial\mathbf{V}} + \frac{1}{2}\mathbf{U}\mathbf{g}\frac{\partial\mathbf{v}}{\partial\mathbf{V}}\right]d\mathbf{V}.$$
 (3.93)

Requiring the asymptotic behavior for large U and V to be as in (3.82), we find

$$\mathcal{E}_{BPS} = 2va^2$$
.

We construct our trial functions by modifying the abelian solution (3.82). But first we need to smooth out the singularity of that solution at  $\mathbf{V} = a^2$ , while preserving the asymptotic behavior at large **U** and **V**, as well as the behavior (3.84) at  $\mathbf{V} = 0$ . For this purpose we define:

$$\mathcal{R} \equiv \sqrt{\mathbf{U} + \mathbf{V} + a^2} = \sqrt{r^2 + \rho^2 + a^2} \tag{3.94}$$

and then define

$$\begin{split} \tilde{\mathbf{f}} &\equiv \frac{av}{\mathcal{R}^2} + \frac{2a(va^2 - 2)}{\mathcal{R}^4} - \frac{2a^3v\mathbf{U}}{\mathcal{R}^6} \,, \\ \tilde{\mathbf{g}} &\equiv v - \frac{2}{\mathcal{R}^2} - \frac{2va^2\mathbf{U}}{\mathcal{R}^4} \,, \\ \tilde{\mathbf{h}} &\equiv v \left( \frac{a}{\mathcal{R}^2} + \frac{2a^3}{\mathcal{R}^4} - \frac{2a^3\mathbf{U}}{\mathcal{R}^6} - \frac{2a^5(a^2 + \mathbf{U})}{\mathcal{R}^8} \right) \,, \\ \tilde{\mathbf{q}} &\equiv \frac{4a}{\mathcal{R}^2} - \frac{8a\mathbf{V}}{\mathcal{R}^4} \,, \\ \tilde{\mathbf{v}} &\equiv -\frac{2}{\mathcal{R}^2} - \frac{8a^2}{\mathcal{R}^4} + \frac{8a^2(a^2 + \mathbf{U})}{\mathcal{R}^6} \,, \end{split}$$

so that for fixed  ${\bf U}$  and  ${\bf V} \rightarrow \infty$  we have

$$\begin{split} \tilde{\mathbf{f}} &= \frac{v}{2\mathbf{R}} - \frac{1}{a\mathbf{R}^2} + O\left(\frac{1}{\mathbf{V}^4}\right) \,, \\ \tilde{\mathbf{g}} &= \frac{v\mathbf{Z}}{\mathbf{R}} - \frac{\mathbf{Z}}{a\mathbf{R}^2} + O\left(\frac{1}{\mathbf{V}^3}\right) \,, \\ \tilde{\mathbf{h}} &= \frac{v}{2\mathbf{R}} + O\left(\frac{1}{\mathbf{V}^4}\right) \,, \\ \tilde{\mathbf{q}} &= \frac{a^2 + \mathbf{U} - \mathbf{V}}{a\mathbf{R}^2} + O\left(\frac{1}{\mathbf{V}^3}\right) \,, \\ \tilde{\mathbf{v}} &= -\frac{1}{\mathbf{R}^2} - \frac{\mathbf{Z}}{a\mathbf{R}^2} + O\left(\frac{1}{\mathbf{V}^4}\right) \,, \end{split}$$

and  $\tilde{\mathbf{f}}, \tilde{\mathbf{g}}, \tilde{\mathbf{h}}, \tilde{\mathbf{v}}, \tilde{\mathbf{q}}$  are smooth everywhere. We also define

$$\mathcal{R}_b \equiv \sqrt{\mathbf{U} + \mathbf{V} + b^2} = \sqrt{r^2 + \rho^2 + b^2},$$

where b is a parameter to be determined dynamically by the variational principle. We now pick a sufficiently large integer N (we chose N = 20 below), and construct trial functions in

the form:

$$\begin{split} \mathbf{f} &= \tilde{\mathbf{f}} + \frac{1}{\mathcal{R}_b^{5+2N}} \sum_{n,m \ge 0}^{n+m \le N} \mathbf{f}_{m,n} \mathbf{U}^m \mathbf{V}^n \,, \\ \mathbf{g} &= \tilde{\mathbf{g}} + \frac{1}{\mathcal{R}_b^{4+2N}} \sum_{n,m \ge 0}^{n+m \le N} \mathbf{g}_{m,n} \mathbf{U}^m \mathbf{V}^n \,, \\ \mathbf{h} &= \tilde{\mathbf{h}} + \frac{\mathbf{V}}{\mathcal{R}_b^{5+2N}} \sum_{n,m \ge 0}^{n+m \le N-1} \mathbf{h}_{m,n} \mathbf{U}^m \mathbf{V}^n \,, \\ \mathbf{q} &= \tilde{\mathbf{q}} + \frac{\mathbf{V}}{\mathcal{R}_b^{4+2N}} \sum_{n,m \ge 0}^{n+m \le N-1} \mathbf{q}_{m,n} \mathbf{U}^m \mathbf{V}^n \,, \\ \mathbf{q} &= \tilde{\mathbf{v}} + \frac{\mathbf{V}}{\mathcal{R}_b^{4+2N}} \sum_{n,m \ge 0}^{n+m \le N-1} \mathbf{v}_{m,n} \mathbf{U}^m \mathbf{V}^n \,, \end{split}$$

where  $\mathbf{f}_{m,n}$ ,  $\mathbf{g}_{m,n}$ ,  $\mathbf{h}_{m,n}$ ,  $\mathbf{q}_{m,n}$ ,  $\mathbf{v}_{m,n}$  are coefficients to be determined. These expressions are designed to preserve the boundary condition (3.84), as well as the asymptotic behavior for large U and V. We then find the coefficients  $\mathbf{f}_{m,n}$ ,  $\mathbf{g}_{m,n}$ ,  $\mathbf{h}_{m,n}$ ,  $\mathbf{q}_{m,n}$ ,  $\mathbf{v}_{m,n}$  that minimize  $\mathcal{E}$ , using the Newton-Raphson method for given b, and finally we optimize b.

For example, we find for the dimensionless coefficient  $va^2 = 1$  and N = 20 that the optimal b is 2.8a. We define the energy density

$$\mathcal{U} \equiv \frac{1}{2} \operatorname{tr} \left[ (D_r \widetilde{\Phi})^2 + (D_1 \widetilde{\Phi})^2 + (D_2 \widetilde{\Phi})^2 \right] + \frac{1}{2} r^2 \operatorname{tr} \left[ F_{12}^2 + F_{1r}^2 + F_{2r}^2 \right] , \qquad (3.95)$$

for the exact solution we have

$$\mathcal{U} = \mathcal{U}_{\rm BPS} \equiv r \operatorname{tr} \left[ F_{12} D_r \widetilde{\Phi} + F_{2r} D_1 \widetilde{\Phi} + F_{r1} D_2 \widetilde{\Phi} \right].$$
(3.96)

The total energy is then

$$\mathcal{E}_{BPS} = \frac{1}{4} \int \frac{1}{\mathbf{V}} \mathcal{U}_{BPS} d\mathbf{V} d\mathbf{U}.$$

We present in Figure 3.4 our<sup>1</sup> numerical results for  $\Theta \equiv \mathcal{U}/\mathbf{V}$  as well as for the gauge invariant absolute value of the scalar field

$$|\widetilde{\Phi}| \equiv (\widetilde{\Phi}^a \widetilde{\Phi}^a)^{1/2} = \sqrt{\mathbf{U}(\mathbf{f}^2 + \mathbf{h}^2) + \mathbf{g}^2} \,.$$

The results are for  $va^2 = 1$ , and it is interesting to note that for such a relatively small value of  $va^2$ , the core of the soliton (where  $|\tilde{\Phi}| = 0$ ) is at  $r \approx 2.9$  (V = 8.3 in the graph of Figure 3.4), which is far from a = 1.

<sup>&</sup>lt;sup>1</sup>Graph drawn by Mathematica, Version 9.0, (Wolfram Research, Inc.).



Figure 3.4: Results of a numerical simulation with parameters b = 2.80 and N = 22. The graphs show the energy density  $\Theta \equiv \mathcal{U}/\mathbf{V}$  (solid line) and the gauge invariant absolute value of the scalar field  $|\tilde{\Phi}| \equiv (\tilde{\Phi}^a \tilde{\Phi}^a)^{1/2}$  (dashed line) for VEV v = 1 and soliton center at a = 1. The graphs are on the axis  $\mathbf{U} = 0$  and the horizontal axis is  $\mathbf{V}$ . At  $\mathbf{V} = 0$  the value of  $\Theta$  is  $2.0 \times 10^{-3}$  and the value of  $|\tilde{\Phi}|$  is 0.76. The value of the excess energy  $\mathcal{E}$  for this simulation is less than  $2 \times 10^{-5}$  of  $\mathcal{E}_{\text{BPS}}$ .

### Bibliography

- P. Pasti, D. P. Sorokin and M. Tonin, "On Lorentz invariant actions for chiral p-forms," Phys. Rev. D 55, 6292 (1997) [hep-th/9611100].
- [2] A. Dabholkar and J. A. Harvey, "String islands," JHEP 9902, 006 (1999) [hep-th/9809122].
- [3] O. J. Ganor, "U duality twists and possible phase transitions in (2+1)-dimensions supergravity," Nucl. Phys. B 549, 145 (1999) [hep-th/9812024].
- [4] D. Gaiotto and E. Witten, "S-Duality of Boundary Conditions In N = 4 Super Yang-Mills Theory," Adv. Theor. Math. Phys. 13, 721 (2009) [arXiv:0807.3720 [hep-th]].
- [5] O. J. Ganor and Y. P. Hong, "Selfduality and Chern-Simons Theory," arXiv:0812.1213 [hep-th].
- [6] O. J. Ganor, Y. P. Hong and H. S. Tan, "Ground States of S-duality Twisted N = 4Super Yang-Mills Theory," JHEP **1103**, 099 (2011) [arXiv:1007.3749 [hep-th]].
- [7] Y. Terashima and M. Yamazaki, "SL(2, ℝ) Chern-Simons, Liouville, and Gauge Theory on Duality Walls," JHEP 1108, 135 (2011) [arXiv:1103.5748 [hep-th]].
- [8] T. Dimofte and S. Gukov, "Chern-Simons Theory and S-duality," JHEP 1305, 109 (2013) [arXiv:1106.4550 [hep-th]].
- T. Dimofte, D. Gaiotto and S. Gukov, "Gauge Theories Labelled by Three-Manifolds," Commun. Math. Phys. 325, 367 (2014) [arXiv:1108.4389 [hep-th]].
- [10] S. Cecotti, C. Cordova and C. Vafa, "Braids, Walls, and Mirrors," arXiv:1110.2115 [hep-th].
- [11] O. J. Ganor, Y. P. Hong, R. Markov and H. S. Tan, "Static Charges in the Low-Energy Theory of the S-Duality Twist," JHEP 1204, 041 (2012) [arXiv:1201.2679 [hep-th]].
- [12] O. J. Ganor, N. Moore, H. Sun, and N. Torres-Chicon, to appear.
- [13] L. G. Molinari, "Determinants of Block Tridiagonal Matrices," Linear Algebra and its Applications 429 (2008) 2221-2226 [arXiv:0712.0681 [math-ph]].

- [14] I. M. Gelfand and A. M. Yaglom, "Integration in functional spaces and it applications in quantum physics," J. Math. Phys. 1, 48 (1960).
- [15] O. J. Ganor, "A Note on zeros of superpotentials in F-theory," Nucl. Phys. B 499, 55 (1997) [hep-th/9612077].
- [16] E. Witten, "SL(2, Z) action on three-dimensional conformal field theories with Abelian symmetry," In \*Shifman, M. (ed.) et al.: From fields to strings, vol. 2\* 1173-1200 [hepth/0307041].
- [17] D. Bak, M. Gutperle and S. Hirano, "A Dilatonic deformation of AdS<sub>5</sub> and its field theory dual," JHEP 0305, 072 (2003) [hep-th/0304129].
- [18] A. Clark and A. Karch, "Super Janus," JHEP **0510**, 094 (2005) [hep-th/0506265].
- [19] E. D'Hoker, J. Estes and M. Gutperle, "Interface Yang-Mills, supersymmetry, and Janus," Nucl. Phys. B 753, 16 (2006) [hep-th/0603013].
- [20] D. Gaiotto and E. Witten, "Janus Configurations, Chern-Simons Couplings, And The theta-Angle in N = 4 Super Yang-Mills Theory," JHEP **1006**, 097 (2010) [arXiv:0804.2907 [hep-th]].
- [21] A. Kapustin and E. Witten, "Electric-Magnetic Duality And The Geometric Langlands Program," Commun. Num. Theor. Phys. 1, 1 (2007) [hep-th/0604151].
- [22] R. Bott and L. W. Tu, Differential Forms in Algebraic Topology, Graduate Texts in Mathematics 82.
- [23] J. M. Maldacena and A. Strominger, "AdS<sub>3</sub> black holes and a stringy exclusion principle," JHEP 9812, 005 (1998) [hep-th/9804085].
- [24] A. Gadde, S. Gukov and P. Putrov, "Fivebranes and 4-manifolds," arXiv:1306.4320 [hep-th].
- [25] E. Witten, "Some comments on string dynamics," In Los Angeles 1995, Future perspectives in string theory 501-523 [hep-th/9507121].
- [26] A. Strominger, "Open *p*-branes," Phys. Lett. B **383**, 44 (1996) [hep-th/9512059].
- [27] O. J. Ganor, Y. P. Hong and H. S. Tan, "Ground States of S-duality Twisted N = 4Super Yang-Mills Theory," JHEP **1103**, 099 (2011) [arXiv:1007.3749 [hep-th]].
- [28] P. Hořava and E. Witten, "Heterotic and type I string dynamics from eleven-dimensions," Nucl. Phys. B 460, 506 (1996) [hep-th/9510209].
- [29] T. Dimofte, D. Gaiotto and S. Gukov, "Gauge Theories Labelled by Three-Manifolds," Commun. Math. Phys. 325, 367 (2014) [arXiv:1108.4389 [hep-th]].

- [30] T. Dimofte, D. Gaiotto and S. Gukov, "3-Manifolds and 3d Indices," arXiv:1112.5179 [hep-th].
- [31] A. Gadde, S. Gukov and P. Putrov, "Fivebranes and 4-manifolds," arXiv:1306.4320 [hep-th].
- [32] A. Gadde, S. Gukov and P. Putrov, "Duality Defects," arXiv:1404.2929 [hep-th].
- [33] L. Martucci, "Topological duality twist and brane instantons in F-theory," arXiv:1403.2530 [hep-th].
- [34] E. Witten, "New 'gauge' theories in six-dimensions," JHEP 9801, 001 (1998) [Adv. Theor. Math. Phys. 2, 61 (1998)] [hep-th/9710065].
- [35] H. Linander and F. Ohlsson, "(2,0) theory on circle fibrations," JHEP **1201**, 159 (2012) [arXiv:1111.6045 [hep-th]].
- [36] M. K. Prasad and C. M. Sommerfield, "An Exact Classical Solution for the 't Hooft Monopole and the Julia-Zee Dyon," Phys. Rev. Lett. 35, 760 (1975).
- [37] N. J. Hitchin, "Monopoles And Geodesics," Commun. Math. Phys. 83, 579 (1982).
- [38] N. J. Hitchin, "On the Construction of Monopoles," Commun. Math. Phys. 89, 145 (1983).
- [39] D. -E. Diaconescu, "D-branes, monopoles and Nahm equations," Nucl. Phys. B 503, 220 (1997) [hep-th/9608163].
- [40] M. K. Murray, "Monopoles," In \*Bouwknegt, P. (ed.) et al.: Geometric analysis and applications to quantum field theory\* 119-135 [math-ph/0101035].
- [41] P. Forgacs, Z. Horvath and L. Palla, "Generating The Bps One Monopole By A Backlund Transformation," Phys. Rev. Lett. 45, 505 (1980).
- [42] D. Korotkin and H. Nicolai, "Separation of variables and Hamiltonian formulation for the Ernst equation," Phys. Rev. Lett. 74, 1272 (1995) [hep-th/9412072].
- [43] M. F. Atiyah, "Magnetic monopoles in hyperbolic spaces," In \*Atiyah, M.: Collected works, vol. 5\* 579-611. (In \*Bombay 1984, Proceedings, Vector bundles on algebraic variables\* 1-34).
- [44] G. Neugebauer and D. Kramer, "A method for the construction of stationary einsteinmaxwell fields. (in german)," Annalen Phys. 24, 62 (1969).
- [45] N. S. Manton, "Complex Structure of Monopoles," Nucl. Phys. B 135, 319 (1978).
- [46] G. 't Hooft, "Magnetic Monopoles in Unified Gauge Theories," Nucl. Phys. B 79, 276 (1974).

- [47] Weisstein, Eric W. "General Confluent Hypergeometric Differential Equation." and "Confluent Hypergeometric Function of the First Kind." From MathWorld–A Wolfram Web Resource.
- [48] Weisstein, Eric W. "Modified Bessel Function of the Second Kind." From MathWorld–A Wolfram Web Resource.
- [49] J. H. Brodie, L. Susskind and N. Toumbas, "How Bob Laughlin tamed the giant graviton from Taub - NUT space," JHEP 0102, 003 (2001) [hep-th/0010105].
- [50] S. Hellerman and L. Susskind, "Realizing the quantum Hall system in string theory," hep-th/0107200.
- [51] O. Bergman, Y. Okawa and J. H. Brodie, "The Stringy quantum Hall fluid," JHEP 0111, 019 (2001) [hep-th/0107178].
- [52] D. Gaiotto and E. Witten, "Janus Configurations, Chern-Simons Couplings, And The theta-Angle in N = 4 Super Yang-Mills Theory," JHEP **1006**, 097 (2010) [arXiv:0804.2907 [hep-th]].
- [53] E. Keski-Vakkuri and P. Kraus, "Quantum Hall Effect in AdS/CFT," JHEP 0809, 130 (2008) [arXiv:0805.4643 [hep-th]].
- [54] D. Gaiotto and E. Witten, "S-Duality of Boundary Conditions In N = 4 Super Yang-Mills Theory," Adv. Theor. Math. Phys. **13** (2009) [arXiv:0807.3720 [hep-th]].
- [55] M. Fujita, W. Li, S. Ryu and T. Takayanagi, "Fractional Quantum Hall Effect via Holography: Chern-Simons, Edge States, and Hierarchy," JHEP 0906, 066 (2009) [arXiv:0901.0924 [hep-th]].
- [56] K. A. Intriligator and N. Seiberg, "Mirror symmetry in three-dimensional gauge theories," Phys. Lett. B 387, 513 (1996) [hep-th/9607207].
- [57] T. Kitao, K. Ohta and N. Ohta, "Three-dimensional gauge dynamics from brane configurations with (p,q) - five-brane," Nucl. Phys. B 539, 79 (1999) [hep-th/9808111].
- [58] N. Nekrasov and E. Witten, "The Omega Deformation, Branes, Integrability, and Liouville Theory," JHEP 1009, 092 (2010) [arXiv:1002.0888 [hep-th]].
- [59] S. Hellerman, D. Orlando and S. Reffert, "BPS States in the Duality Web of the Omega deformation," JHEP 1306, 047 (2013) [arXiv:1210.7805 [hep-th]].
- [60] O. J. Ganor, "Six-dimensional tensionless strings in the large N limit," Nucl. Phys. B 489, 95 (1997) [hep-th/9605201].
- [61] S. K. Domokos, C. Hoyos and J. Sonnenschein, "Deformation Constraints on Solitons and D-branes," JHEP 1310, 003 (2013) [arXiv:1306.0789 [hep-th]].

- [62] R. Throngren, "Framed Wilson Operators on the Boundaries of Novel SPT Phases," arXiv:1404.4385 [hep-th].
- [63] S. R. Coleman, "Q Balls," Nucl. Phys. B 262, 263 (1985) [Erratum-ibid. B 269, 744 (1986)].
- [64] W. Nahm, "A Simple Formalism for the BPS Monopole," Phys. Lett. B 90, 413 (1980).
- [65] A. Hanany and E. Witten, "Type IIB superstrings, BPS monopoles, and threedimensional gauge dynamics," Nucl. Phys. B 492, 152 (1997) [hep-th/9611230].
- [66] E. Witten, "Small instantons in string theory," Nucl. Phys. B 460, 541 (1996) [hep-th/9511030].
- [67] M. R. Douglas, "Branes within branes," In \*Cargese 1997, Strings, branes and dualities\* 267-275 [hep-th/9512077].
- [68] O. Aharony, O. Bergman, D. L. Jafferis and J. Maldacena, "N = 6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals," JHEP **0810**, 091 (2008) [arXiv:0806.1218 [hep-th]].
- [69] J. Bagger and N. Lambert, "Modeling Multiple M2's," Phys. Rev. D 75, 045020 (2007) [hep-th/0611108].
- [70] J. Bagger and N. Lambert, "Gauge symmetry and supersymmetry of multiple M2branes," Phys. Rev. D 77, 065008 (2008) [arXiv:0711.0955 [hep-th]].
- [71] A. Gustavsson, "Algebraic structures on parallel M2-branes," Nucl. Phys. B 811, 66 (2009) [arXiv:0709.1260 [hep-th]].
- [72] W. Nahm, "A Simple Formalism for the BPS Monopole," Phys. Lett. B 90, 413 (1980).

### Appendix A

### A proof of the determinant identity and the Smith normal form of the coupling constant matrix

Molinari gave an elegant proof [13] to a generalization of (2.2) using only polynomial analysis. Here we present an alternative basic linear-algebra proof for (2.2). At the same time we also demonstrate that the Smith normal form of the coupling constant matrix Kdefined in (2.5),

$$K = \begin{pmatrix} \mathbf{k}_1 & -1 & 0 & \ddots & -1 \\ -1 & \ddots & \ddots & \ddots & \ddots \\ 0 & \ddots & \ddots & \ddots & 0 \\ \ddots & \ddots & \ddots & \ddots & -1 \\ -1 & \ddots & 0 & -1 & \mathbf{k}_n \end{pmatrix}$$

,

is identical to the Smith normal form of

$$H = W - \mathbf{I} = \begin{pmatrix} \mathbf{a} - 1 & \mathbf{b} \\ \mathbf{c} & \mathbf{d} - 1 \end{pmatrix},$$

where W was defined in (2.11).

We begin by moving the first row of K to the end, to get  $K'_1$ . We have

$$\det K = (-1)^n \det K_1'$$

but both K and  $K'_1$  have the same Smith normal form. For clarity, we will present explicit matrices for the n = 5 case. We get:

$$K_1' \equiv \begin{pmatrix} -1 & \mathbf{k}_2 & -1 & 0 & 0\\ 0 & -1 & \mathbf{k}_3 & -1 & 0\\ 0 & 0 & -1 & \mathbf{k}_4 & -1\\ -1 & 0 & 0 & -1 & \mathbf{k}_5\\ \mathbf{k}_1 & -1 & 0 & 0 & -1 \end{pmatrix} \,.$$

We will now show how to successively define a series of matrices

$$K'_{2}, \dots, K'_{n-1} = \begin{pmatrix} -1 & & & \\ & \ddots & & \\ & & -1 & & \\ & & \mathbf{a} - 1 & \mathbf{b} \\ & & & \mathbf{c} & \mathbf{d} - 1 \end{pmatrix}$$

related to each other by row and column operations that preserve the Smith normal form. At each step, we need to keep track of a  $2 \times 2$  block of  $K'_m$  formed from the elements on the  $(n-1)^{th}$  and  $n^{th}$  rows and the  $m^{th}$  and  $(m+1)^{st}$  columns.

$$H'_{m} \equiv \begin{pmatrix} [K'_{m}]_{(n-1)m} & [K'_{m}]_{(n-1)(m+1)} \\ [K'_{m}]_{nm} & [K'_{m}]_{n(m+1)} \end{pmatrix}$$

At the outset we have

$$H_1' \equiv \begin{pmatrix} [K_1']_{(n-1)1} & [K_1']_{(n-1)2} \\ [K_1']_{n1} & [K_1']_{n2} \end{pmatrix} = \begin{pmatrix} -1 & 0 \\ \mathbf{k}_1 & -1 \end{pmatrix}$$

As will soon be clear from the construction, the matrix  $K'_m$  has the following block form:

$$K'_{m} = \begin{pmatrix} -\mathbf{I}_{m-1} & & & & \\ & -1 & \mathbf{k}_{m+1} & -1 & * & * & * \\ & & -1 & \mathbf{k}_{m+2} & * & * & * \\ & & & & X_{n-m-4} & * & * \\ & & & & & I_{n-m-4} & * & * \\ & & & & & & I_{n-m-4} & * & * \\ & & & & & & I_{n-m-4} & * & * \\ & & & & & & I_{n-m-4} & * & * \\ & & & & & & I_{n-m-4} & * & I_{n-m-4} \\ & & & & & & I_{n-m-4} & * & I_{n-m-4} \\ & & & & & & I_{n-m-4} & I_{n-m-4} \\ & & & & & & I_{n-m-4} & I_{n-m-4} \\ & & & & & & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} & I_{n-m-4} \\ & & & I_{n-m-4} &$$

where  $\mathbf{I}_{m-1}$  is the  $(m-1) \times (m-1)$  identity matrix, \* represents a block of possibly nonzero elements,  $X_{n-m-4}$  represents a nonzero  $(n-m-4) \times (n-m-4)$  matrix and empty positions are zero. To get  $K'_{m+1}$  from  $K'_m$  we perform the following row and column operations on  $K'_m$ :

- Add  $[H'_m]_{11}$  times the  $m^{th}$  row to the  $(n-1)^{st}$  row;
- Add  $[H'_m]_{21}$  times the  $m^{st}$  row to the  $n^{th}$  row;
- For j = m + 1, ..., n, add  $[K'_m]_{mj}$  times the  $m^{th}$  column to the  $j^{th}$  column.

It is not hard to see that these operations produce a matrix that fits the general form (A.1) with  $m \to m+1$ . Tracking how the bottom two rows transform, we find that for m < n-2,

$$H'_{m+1} = \begin{pmatrix} [H'_{m+1}]_{11} & [H'_{m+1}]_{12} \\ [H'_{m+1}]_{21} & [H'_{m+1}]_{22} \end{pmatrix} = \begin{pmatrix} [H'_m]_{12} + \mathbf{k}_{m+1}[H'_m]_{11} & -[H'_m]_{11} \\ [H'_m]_{22} + \mathbf{k}_{m+1}[H'_m]_{21} & -[H'_m]_{21} \end{pmatrix} = H'_m \begin{pmatrix} \mathbf{k}_{m+1} & 1 \\ -1 & 0 \end{pmatrix} .$$

,

Since, by definition,  $H'_1 = \begin{pmatrix} -1 & 0 \\ \mathbf{k}_1 & -1 \end{pmatrix}$ , it follows that

$$H'_{n-2} = \begin{pmatrix} -1 & 0 \\ \mathbf{k}_1 & -1 \end{pmatrix} \begin{pmatrix} \mathbf{k}_2 & 1 \\ -1 & 0 \end{pmatrix} \cdots \begin{pmatrix} \mathbf{k}_{n-2} & 1 \\ -1 & 0 \end{pmatrix} .$$

It can then be easily checked that the last two steps yield:

$$H'_{n} = H'_{n-2} \begin{pmatrix} \mathbf{k}_{n-1} & 1\\ -1 & 0 \end{pmatrix} \begin{pmatrix} \mathbf{k}_{n} & 1\\ -1 & 0 \end{pmatrix} - \begin{pmatrix} 1 & 0\\ 0 & 1 \end{pmatrix}$$

### Appendix B

### Compatibility of the supersymmetric Janus configuration and the duality twist

In this section we describe the details of the supersymmetric Lagrangian. As explained in §2.2, the system is composed of two ingredients: (i) the supersymmetric Janus configuration; and (ii) an  $SL(2,\mathbb{Z})$  duality twist. We will now review the details of both ingredients and demonstrate that their combination preserves supersymmetry.

### **B.1** Supersymmetric Janus

Extending the work of [17]-[19], Gaiotto and Witten [52] have constructed a supersymmetric deformation of  $\mathcal{N} = 4$  Super-Yang-Mills theory with a complex coupling constant  $\tau$  that varies along one direction, which we denote by  $x_3$ . We will now review this construction, using the same notation as in [52]. First, the real and imaginary parts of the coupling constant are defined as

$$\tau = \frac{\theta}{2\pi} + \frac{2\pi i}{\mathbf{e}^2},\tag{B.1}$$

It is taken to vary along a semi-circle on the upper half  $\tau$ -plane, centered on the real axis:

$$\tau = a + 4\pi D e^{2i\psi},\tag{B.2}$$

where  $\psi(x_3)$  is an arbitrary function.

The action is defined as

$$I = I_{N=4} + I' + I'' + I'''$$

where  $I_{N=4}$  is the standard  $\mathcal{N} = 4$  action, modified only by making  $\tau$  a function of  $x_3$ , and I', I'', and I''' are correction terms listed below. We will list the actions for a general gauge group, as derived by Gaiotto and Witten, although the application in this paper is for a U(1) gauge group, and so several terms drop out. The bosonic fields are: a gauge field  $A_{\mu}$  ( $\mu = 0, 1, 2, 3$ ), 3 adjoint-valued scalar fields  $X^a$  (a = 1, 2, 3) and 3 adjoint-valued scalar

fields  $Y^p$  (p = 1, 2, 3). In the U(1) case,  $X^a$  and  $Y^p$  are real scalar fields. In the type-IIB realization on D3-branes, the D3-brane is in directions 0, 1, 2, 3,  $X^a$  corresponds to fluctuations in directions 4, 5, 6, and  $Y^p$  corresponds to directions 7, 8, 9. The fermionic fields are encoded in a 16-dimensional Majorana-Weyl spinor  $\Psi$  on which even products of the 9+1D Dirac matrices  $\Gamma_0, \ldots, \Gamma_9$  act. Products of pairs from the list  $\Gamma_0, \ldots, \Gamma_3$  correspond to generators of the Lorentz group SO(1,3), while products of pairs from the list  $\Gamma_4, \Gamma_5, \Gamma_6$  correspond to generators of the R-symmetry subgroup  $SO(3)_X$  acting on  $X^1, X^2, X^3$ , and products of pairs from the list  $\Gamma_7, \Gamma_8, \Gamma_9$  correspond to generators of the R-symmetry subgroup  $SO(3)_Y$ acting on  $Y^1, Y^2, Y^3$ . We have the identity  $\Gamma_{0123456789} = 1$ .

The additional terms are

$$I' = \frac{i}{\mathbf{e}^2} \int d^4 x \operatorname{Tr} \overline{\Psi} (\alpha \Gamma_{012} + \beta \Gamma_{456} + \gamma \Gamma_{789}) \Psi ,$$
  

$$I'' = \frac{1}{\mathbf{e}^2} \int d^4 x \operatorname{Tr} \left( \mathbf{u} \epsilon^{\mu\nu\lambda} (A_\mu \partial_\nu A_\lambda + \frac{2}{3} A_\mu A_\nu A_\lambda) + \frac{\mathbf{v}}{3} \epsilon^{abc} X_a [X_b, X_c] + \frac{\mathbf{w}}{3} \epsilon^{pqr} Y_p [Y_q, Y_r] \right) ,$$
  

$$I''' = \frac{1}{2\mathbf{e}^2} \int d^4 x \operatorname{Tr} \left( \mathbf{r} X_a X^a + \widetilde{\mathbf{r}} Y_p Y^p \right) ,$$

where

$$-\frac{1}{4}\mathbf{u} = \alpha = -\frac{1}{2}\psi', \qquad -\frac{1}{4}\mathbf{v} = \beta = -\frac{\psi'}{2\cos\psi}, \qquad -\frac{1}{4}\mathbf{w} = \gamma = \frac{\psi'}{2\sin\psi}, \qquad (B.3)$$

$$\mathbf{r} = 2(\psi' \tan \psi)' + 2(\psi')^2, \qquad \widetilde{\mathbf{r}} = -2(\psi' \cot \psi)' + 2(\psi')^2.$$
 (B.4)

As we are working with a U(1) gauge group, we will not need the cubic terms in I''. They are nevertheless listed here for reference, and they will become relevant for extensions to a nonabelian gauge group.

To describe the preserved supersymmetry we follow Gaiotto-Witten and work in a spinor representation where

$$\Gamma_{0123} = -\Gamma_{456789} = \begin{pmatrix} 0 & -\mathbf{I} \\ \mathbf{I} & 0 \end{pmatrix}, \quad \Gamma_{3456} = \begin{pmatrix} 0 & \mathbf{I} \\ \mathbf{I} & 0 \end{pmatrix}, \quad \Gamma_{3789} = \begin{pmatrix} \mathbf{I} & 0 \\ 0 & -\mathbf{I} \end{pmatrix},$$

where I is an  $8 \times 8$  identity matrix. The surviving supersymmetries are those parameterized by a 16-component  $\varepsilon_{16}$  which takes the form

$$\varepsilon_{16} = \begin{pmatrix} \cos(\frac{\psi}{2})\varepsilon_8\\ \sin(\frac{\psi}{2})\varepsilon_8 \end{pmatrix}, \qquad (B.5)$$

where  $\varepsilon_8$  is an arbitrary constant 8-component spinor.

### **B.2** Introducing an $SL(2, \mathbb{Z})$ -twist

Here  $\psi$  is a function of  $x_3$  such that  $\tau(x_3)$  traces a geodesic on  $\tau$ -plane with metric  $|d\tau|^2/\tau_2^2$ . We pick the parameters a and D so that the semi-circle (B.2) will be invariant



Figure B.1: In the Janus configuration the coupling constant  $\tau$  traces a portion of a semi-circle of radius  $4\pi D$  in the upper-half plane, whose center a is on the real axis. We augment it with an  $SL(2,\mathbb{Z})$  duality twist that glues  $x_3 = 2\pi$  to  $x_3 = 0$ .

under

$$au 
ightarrow rac{\mathbf{a} au + \mathbf{b}}{\mathbf{c} au + \mathbf{d}} \, .$$

This amounts to solving the two equations

$$(a - 4\pi D) = \frac{\mathbf{a}(a - 4\pi D) + \mathbf{b}}{\mathbf{c}(a - 4\pi D) + \mathbf{d}}, \qquad (a + 4\pi D) = \frac{\mathbf{a}(a + 4\pi D) + \mathbf{b}}{\mathbf{c}(a + 4\pi D) + \mathbf{d}}$$

The solution is:

$$a = \frac{\mathbf{a} - \mathbf{d}}{2\mathbf{c}}, \qquad 4\pi D = \frac{\sqrt{(\mathbf{a} + \mathbf{d})^2 - 4}}{2|\mathbf{c}|}$$

and is real for a hyperbolic element of  $SL(2, \mathbb{Z})$  (with  $|\mathbf{a} + \mathbf{d}| > 2$ ). Note that it is important to have both  $(a \pm 4\pi D)$  as fixed-points of the  $SL(2, \mathbb{Z})$  transformation, so as not to reverse the orientation of the  $\tau(x_3)$  curve, and not create a discontinuity in  $\tau'(x_3)$ . So, given  $\mathbf{a}$ ,  $\mathbf{b}$ ,  $\mathbf{c}$ ,  $\mathbf{d}$ , our configuration is constructed by first calculating a and D, and then picking an arbitrary  $\psi(2\pi)$  with a corresponding  $\tau(2\pi) = a + 4\pi D e^{2i\psi(2\pi)}$ . Next, we calculate the  $SL(2, \mathbb{Z})$  dual  $\tau(0) = (\mathbf{a}\tau(2\pi) + \mathbf{b})/(\mathbf{c}\tau(2\pi) + \mathbf{d})$  and match it to a point on the semicircle according to  $\tau(0) = a + 4\pi D e^{2i\psi(0)}$ . The function  $\psi(x_3)$  can then be chosen arbitrarily, as long as it connects  $\psi(0)$  to  $\psi(2\pi)$ . It can then be checked that  $\mathbf{r}$  and  $\tilde{\mathbf{r}}$  are continuous at  $x_3 = 2\pi$ .

At low-energy, the mass parameters  $\mathbf{r}$  and  $\tilde{\mathbf{r}}$  in I''' make the scalar fields  $(X^a \text{ and } Y^p)$ massive. Note that in principle, the parameters can be locally negative [although this can be averted by choosing  $\psi(x_3)$  so that  $\psi'' = 0$ ], but the effective 2+1D masses, [obtained by solving for the spectrum of the operators  $-d^2/dx_3^2 + \mathbf{r}(x_3)$ , and  $-d^2/dx_3^2 + \tilde{\mathbf{r}}(x_3)$ ] have to be positive, since the configuration is supersymmetric and the BPS bound prevents us from having a profile of  $X^a(x_3)$  or  $Y^p(x_3)$  with negative energy. Similar statements hold for the fermionic masses in I'.

### **B.3** The supersymmetry parameter

As explained in [21], the SL(2,  $\mathbb{Z}$ ) duality transformation acts nontrivially on the SUSY generators. Define the phase  $\varphi$  by

$$e^{i\varphi} = \frac{|\mathbf{c}\tau + \mathbf{d}|}{\mathbf{c}\tau + \mathbf{d}}$$

Then, the SUSY transformations act on the supersymmetry parameter as

$$\varepsilon \to e^{\frac{1}{2}\varphi\Gamma_{0123}}\varepsilon$$
.

(See equation (2.25) of [21].)

We can now check that

$$\frac{\mathbf{c}\tau + \mathbf{d}|}{\mathbf{c}\tau + \mathbf{d}} = e^{i(\tilde{\psi} - \psi)}, \qquad (B.6)$$

where  $\tilde{\psi}$  is defined by

$$\tilde{\tau} \equiv \frac{\mathbf{a}\tau + \mathbf{b}}{\mathbf{c}\tau + \mathbf{d}} \equiv a + 4\pi D e^{2i\tilde{\psi}}.$$

It follows from (B.6) that the Gaiotto-Witten phase that is picked up by the supersymmetry parameter as it traverses the Janus configuration from  $\eta = 0$  (corresponding to angular variable  $\psi$ ) to  $\eta = 2\pi$  (corresponding to  $\tilde{\psi}$ ) is precisely canceled by the Kapustin-Witten phase of the SL(2, Z)-duality twist. The entire "Janus plus twist" configuration is therefore supersymmetric.

### **B.4** Extending to a type-IIA supersymmetric background

In section §2.4 we assumed that there is a lift of the gauge theory construction to type-IIB string theory and, following a series of dualities, we obtained a type-IIA background with NSNS fields turned on. Here we would like to outline how such a lift might be constructed. We start with the well-known  $AdS_3 \times S^3 \times T^4$  type-IIB background, and perform S-duality (if necessary) to get the 3-form flux to be NSNS. Then, take  $AdS_3$  to be of Euclidean signature and replace  $T^4$  with  $\mathbb{R}^4$ , which we then Wick rotate to  $\mathbb{R}^{1,3}$ . We take the  $AdS_3$  metric in the form

$$ds^{2} = \frac{r^{2}}{r_{1}r_{5}}(-dt^{2} + dx_{5}^{2}) + \frac{r_{1}}{r_{5}}\sum_{i=6}^{9} dx_{i}^{2} + \frac{r_{1}r_{5}}{r^{2}}dr^{2} + r_{1}r_{5}d\Omega_{3}^{2}$$
$$H^{(RR)} = \frac{2r_{5}^{2}}{g}(\epsilon_{3} + \epsilon_{6}^{*}\epsilon_{3}), \qquad e^{\phi} = \frac{gr_{1}}{r_{5}}$$

where  $\epsilon_3$  is the volume form on the unit sphere, and  $\epsilon_6^*$  is the Hodge dual in the six dimensions  $x_0, \ldots, x_5$  (of  $AdS_3 \times S^3$ ), and where  $r_1, r_5$  are constants. (We follow the notation of [23].)

We need to change variables  $r \to x_3$ ,  $t \to ix_1$  and  $x_9 \to ix_0$ , and perform S-duality (where the RHS of arrows are the variables of §2.5). We then compactify directions  $x_1$  and  $x_2$  so that  $0 \le x_i < 2\pi L_i$  (i = 1, 2). As a function of  $x_3$ , we define the Kähler modulus of the  $x_1 - x_2$  torus to be

$$\rho = i \frac{4\pi^2 r_1^2 L_1 L_2}{x_3^2}$$

Finally, we perform T-duality on direction  $x_5$  to replace  $\rho$  with the complex structure  $\tau$  of the resulting  $T^2$ . In an appropriate limit, this gives a solution where  $\tau$  goes along a straight perpendicular line in the upper half plane. We can convert it to a semi-circle with an  $SL(2, \mathbb{R})$  transformation.

### Appendix C

# Recasting BPS equations in terms of a single potential

The action (3.64) is invariant under dilatations that act as

$$f(r,\rho) \to f(\lambda r,\lambda \rho), \qquad \chi(r,\rho) \to \chi(\lambda r,\lambda \rho).$$

The components of the associated Noether current are given by

$$J^{r} = \frac{\rho f_{r}^{2}}{2f^{2}} + \frac{\rho^{2} f_{r} f_{\rho}}{rf^{2}} - \frac{\rho f_{\rho}^{2}}{2f^{2}} + \frac{\rho \chi_{r}^{2}}{2f^{2}} + \frac{\rho^{2} \chi_{r} \chi_{\rho}}{rf^{2}} - \frac{\rho \chi_{\rho}^{2}}{2f^{2}},$$
  
$$J^{\rho} = \frac{\rho^{2} f_{\rho}^{2}}{2rf^{2}} + \frac{\rho f_{r} f_{\rho}}{f^{2}} - \frac{\rho^{2} f_{r}^{2}}{2rf^{2}} + \frac{\rho^{2} \chi_{\rho}^{2}}{2rf^{2}} + \frac{\rho \chi_{r} \chi_{\rho}}{f^{2}} - \frac{\rho^{2} \chi_{r}^{2}}{2rf^{2}},$$

The equations of motion (3.63)-(3.62) imply the conservation equation<sup>1</sup>

$$(J^r)_r + (J^\rho)_\rho = 0,$$

which implies that there exists a potential function  $\Phi$  such that

$$J^{\rho} = \Phi_r \,, \qquad J^r = -\Phi_{\rho} \,. \tag{C.1}$$

To proceed, we think of the functions f and  $\chi$  as defining a change of coordinates from  $(f, \chi)$  to  $(r, \rho)$  [similar to (3.66), except with the  $\phi$  coordinate absent]. In  $(r, \rho)$  coordinates, the  $AdS_2$  metric (3.65) becomes:

$$ds^{2} = \mathfrak{G}_{rr}dr^{2} + 2\mathfrak{G}_{r\rho}drd\rho + \mathfrak{G}_{\rho\rho}d\rho^{2}, \qquad (C.2)$$

where the metric  $\mathfrak{G}$  can be expressed, using (C.1), as:

$$\begin{split} \mathfrak{G}_{rr} &= -\frac{r^2}{r^2 + \rho^2} \big( \Phi_{\rho\rho} + \Phi_{rr} + \frac{1}{r} \Phi_r + \frac{1}{\rho} \Phi_\rho \big) \,, \\ \mathfrak{G}_{\rho\rho} &= -\frac{r^2}{r^2 + \rho^2} \big( \Phi_{\rho\rho} + \Phi_{rr} - \frac{1}{r} \Phi_r - \frac{1}{\rho} \Phi_\rho \big) \,, \\ \mathfrak{G}_{r\rho} &= \frac{r}{\rho(r^2 + \rho^2)} \big( r \Phi_r - \rho \Phi_\rho \big) \,. \end{split}$$

<sup>&</sup>lt;sup>1</sup>We stress that this is not directly related to the stress-energy tensor in the original fields  $A_i$  and  $\Phi$ . These generally vanish in BPS configurations [61].

 $\Phi$  then satisfies a nonlinear differential equation that states that the Ricci scalar of (C.2) is R = -2. In order to incorporate the Dirac string for r < a, the function  $\Phi$  must diverge like log  $\rho$  as  $\rho \to 0$  and r < a. For large a, the solution to f and  $\chi$  is given by adapting the Prasad-Sommerfield solution as given by [41]:

$$f = \frac{\rho \sinh \mathbf{R}}{\mathbf{R} + \mathbf{R} \cosh \mathbf{R} \cosh \mathbf{Z} - \mathbf{Z} \sinh \mathbf{Z} \sinh \mathbf{R}}, \qquad \chi = \frac{\mathbf{Z} \cosh \mathbf{Z} \sinh \mathbf{R} - \mathbf{R} \sinh \mathbf{Z} \cosh \mathbf{R}}{\mathbf{R} + \mathbf{R} \cosh \mathbf{R} \cosh \mathbf{Z} - \mathbf{Z} \sinh \mathbf{Z} \sinh \mathbf{R}}$$
(C.3)

where **Z** and **R** are given in (3.81), we have set the VEV v = 1, and we have used **R** as a substitute for the distance from the core of the monopole. From this we find,

$$\Phi \to -\frac{1}{4}\rho^2 + \frac{1}{2}\log\rho - \log\mathbf{R} + \log\sinh\mathbf{R}.$$
 (C.4)

We also note that the abelian solution

$$f = \left(\frac{\mathbf{R} - \mathbf{Z}}{2a}\right) e^{-\frac{1}{2}vr^2}, \qquad \chi = 0.$$

can be derived from the potential

$$\Phi = \frac{1}{4}v^2r^2\rho^2 + \frac{1}{2}v(2a\mathbf{R} + r^2 - \rho^2) + \log\left[\frac{2a\mathbf{R}}{(\mathbf{R} - \mathbf{Z})(a + \mathbf{R} + \mathbf{Z})}\right].$$

Finally, we note that a change of variables,

$$r = ae^{\tau}\cos\sigma$$
,  $\rho = ae^{\tau}\sin\sigma$ ,

converts the metric to the more compact form

$$ds^{2} = -\cos^{2}\sigma(\Phi_{\sigma\sigma} + \Phi_{\tau\tau})(d\sigma^{2} + d\tau^{2}) + \cot\sigma\left[\Phi_{\sigma}(d\sigma^{2} - d\tau^{2}) + 2\Phi_{\tau}d\sigma d\tau\right].$$

## Appendix D Large VEV expansion

In this section we will discuss the behavior of the solution to (3.40) for large VEV v. Since the dimensionless combination is  $va^2$ , we can just as well discuss fixed v and large a, which means that the core of the monopole solution is far from the tip. Let us set<sup>1</sup>  $x_3 \equiv r-a$  and rescaling  $\phi = a\Phi/\mathbf{k}R$ , so that equations (3.40) can be rewritten as

$$(1 + \frac{x_3}{a})D_i\phi = \frac{1}{2}\epsilon_{ijk}F_{jk} \tag{D.1}$$

where in this section i, j, k = 1, 2, 3 refer to  $x_1, x_2, x_3$  with Euclidean metric

$$ds^2 = dx_1^2 + dx_2^2 + dx_3^2$$

In the limit  $a \to \infty$ , (D.1) reduce to Bogomolnyi's equations and the one-monopole solution is [36]:

$$A_i^{a(0)} = \epsilon_{iaj} x_j K(R), \qquad \phi^{a(0)} = x_a H(R),$$
 (D.2)

where

$$H \equiv \frac{1}{R} \coth R - \frac{1}{R^2}, \qquad K \equiv \frac{1}{R \sinh R} - \frac{1}{R^2},$$
 (D.3)

and here

$$R^2 = \sum_{i=1}^3 x_i^2 \,.$$

We set

$$b \equiv \frac{1}{a}$$
,  $\vec{\ell} \equiv (0, 0, b)$ ,

so that  $\frac{x_3}{a} = \vec{\ell} \cdot \vec{x}$ , and (D.1) can be written as:

$$(1 + \vec{\ell} \cdot \vec{x})D_i\phi = \frac{1}{2}\epsilon_{ijk}F_{jk}.$$
 (D.4)

<sup>&</sup>lt;sup>1</sup>We hope the no confusion will arise with the coordinate  $x_3$  that was used in §3.2.1. That coordinate plays no role here, and the only coordinates relevant for this section are  $x_1, x_2$  and  $r = a + x_3$ .
We can now expand around the Prasad-Sommerfield solution:

$$A_i = A_i^{(0)} + bA_i^{(1)} + b^2 A_i^{(2)} + \cdots, \qquad \phi = \phi^{(0)} + b\phi^{(1)} + b^2 \phi^{(2)} + \cdots,$$

where we set the  $0^{th}$  order terms to the Prasad-sommerfield solution (D.2).

At order O(b) we write all possible terms that are allowed by spherical symmetry and we keep only the terms that are also invariant under the parity symmetry

$$\phi^{a}(\vec{x},\vec{\ell}) \to -\phi^{a}(-\vec{x},-\vec{\ell}), \qquad A^{a}_{i}(\vec{x},\vec{\ell}) \to -A^{a}_{i}(-\vec{x},-\vec{\ell}).$$
 (D.5)

The general expression is then

$$b\phi^{a(1)} = \ell_a \mathbf{f}_{1,1}(R) + x_a(\ell_k x_k) \mathbf{f}_{1,2}(R) , \qquad (D.6)$$

$$bA_i^{a(1)} = x_a \epsilon_{ijk} x_j \ell_k \mathbf{f}_{1,3}(R) + x_i \epsilon_{ajk} x_j \ell_k \mathbf{f}_{1,4}(R) + \epsilon_{aij} \ell_j \mathbf{f}_{1,5}(R) , \qquad (D.7)$$

and we note the identity

$$x_{[i}\epsilon_{a]jk}x_{j}\ell_{k} = \frac{1}{2}\epsilon_{aij}\ell_{j}R^{2} - \frac{1}{2}(\ell_{k}x_{k})\epsilon_{aij}x_{j}, \qquad (D.8)$$

which is the reason why we did not include a term of the form  $(\ell_k x_k) \epsilon_{aij} x_j \mathbf{f}_{1,6}$ . The coefficients  $\mathbf{f}_{1,1}, \ldots, \mathbf{f}_{1,5}$  are unknown functions of R.

We also have the freedom to apply an infinitesimal O(b) gauge transformation which takes the form

$$\delta\phi^a = \epsilon_{abc}\lambda^b\phi^c, \qquad \delta A^a_i = \partial_i\lambda^a - \epsilon_{abc}A^b_i\lambda^c$$

with

$$\lambda^a = \epsilon_{abc} x_b \ell_c \mathbf{g}_{1,1}(R)$$

This gives

$$\delta\phi^a = \epsilon_{abc}\epsilon_{bde}x_d\ell_e \mathbf{g}_{1,1}x^c H = -x_a\ell_k x_k \mathbf{g}_{1,1}H + \ell_a R^2 \mathbf{g}_{1,1}H , \qquad (D.9)$$

$$\delta A_i^a = -\epsilon_{iab} \ell_b \mathbf{g}_{1,1} + \frac{1}{R} x_i \epsilon_{abc} x_b \ell_c \mathbf{g}'_{1,1} + x_a \epsilon_{ibc} x_b \ell_c \mathbf{g}_{1,1} K \,. \tag{D.10}$$

Using this gauge transformation we can set one of the parameters in (D.6)-(D.7) to zero. We choose to set

$$\mathbf{f}_{1.5} = 0.$$
 (D.11)

We end up with the general form of the O(b) correction:

$$b\phi^{a(1)} = \ell_a \mathbf{f}_{1,1}(R) + x_a(\ell_k x_k) \mathbf{f}_{1,2}(R), \qquad (D.12)$$

$$bA_i^{a(1)} = x_a \epsilon_{ijk} x_j \ell_k \mathbf{f}_{1,3}(R) + x_i \epsilon_{ajk} x_j \ell_k \mathbf{f}_{1,4}(R) \,. \tag{D.13}$$

Plugging (D.2) and (D.12)-(D.13) into (D.4) and comparing terms of order O(b) we get:

$$HK - \frac{1}{R}H' = \frac{1}{R}(\mathbf{f}'_{1,2} + \mathbf{f}'_{1,3}) - K\mathbf{f}_{1,2} - K\mathbf{f}_{1,3} + (K - H)\mathbf{f}_{1,4}, \qquad (D.14)$$

$$0 = \frac{1}{R}\mathbf{f}_{1,1}' + (1 + R^2 K)\mathbf{f}_{1,3} + R^2 H \mathbf{f}_{1,4}, \qquad (D.15)$$

$$0 = R\mathbf{f}'_{1,3} + K\mathbf{f}_{1,1} - \mathbf{f}_{1,2} + 3\mathbf{f}_{1,3} + (1 + R^2K)\mathbf{f}_{1,4}$$
(D.16)

$$-H(1+R^{2}K) = K\mathbf{f}_{1,1} + (1+R^{2}K)\mathbf{f}_{1,2} + \mathbf{f}_{1,4}, \qquad (D.17)$$

These are ordinary inhomogeneous linear differential equations in  $\mathbf{f}_{1,1}, \ldots, \mathbf{f}_{1,4}$ . Note that  $\mathbf{f}_{1,4}$  can be eliminated from (D.17), so the general solution is given be an arbitrary solution of the full equations (D.14)-(D.17) plus a linear combination of three linearly independent solutions of the homogeneous equations:

$$0 = \frac{1}{R} (\mathbf{f}'_{1,2} + \mathbf{f}'_{1,3}) - K\mathbf{f}_{1,2} - K\mathbf{f}_{1,3} + (K - H)\mathbf{f}_{1,4}, \qquad (D.18)$$

$$0 = \frac{1}{R}\mathbf{f}_{1,1}' + (1 + R^2 K)\mathbf{f}_{1,3} + R^2 H \mathbf{f}_{1,4}, \qquad (D.19)$$

$$0 = R\mathbf{f}_{1,3}' + K\mathbf{f}_{1,1} - \mathbf{f}_{1,2} + 3\mathbf{f}_{1,3} + (1 + R^2 K)\mathbf{f}_{1,4}$$
(D.20)

$$0 = K\mathbf{f}_{1,1} + (1 + R^2 K)\mathbf{f}_{1,2} + \mathbf{f}_{1,4}, \qquad (D.21)$$

The general solution to (D.14)-(D.17) that is nonsingular at R = 0 is

$$\mathbf{f}_{1,1} = -\frac{R}{2\sinh R} + c_1 \left(\frac{R\cosh^2 R}{\sinh^3 R} - \frac{\cosh R}{\sinh^2 R}\right) + c_2 \left(\frac{3R}{\sinh R} - \frac{3R^2\cosh R}{\sinh^2 R} + \frac{R^3\cosh^2 R}{\sinh^3 R}\right), \tag{D.22}$$

$$\mathbf{f}_{1,2} = \frac{1}{2R^2} + \frac{1-2\cosh R}{2R\sinh R} + c_1\left(\frac{1}{R^4} - \frac{\cosh^2 R}{R\sinh^3 R} + \frac{\cosh R - 1}{R^2\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh R} + \frac{(3\cosh R - 1)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh R} + \frac{(3\cosh R - 1)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh R} + \frac{(3\cosh R - 1)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh R} + \frac{(3\cosh R - 1)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh R} + \frac{(3\cosh R - 1)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R\sinh^3 R} + \frac{(2\cosh R - 3)}{\sinh^2 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\sinh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R} + \frac{(2\cosh R - 3)}{R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3 R}\right) + c_2\left(-\frac{R\cosh^2 R}{\cosh^3$$

$$\mathbf{f}_{1,3} = \frac{1}{2R^2} - \frac{1}{2R} \coth R + c_1 \left( -\frac{1}{R^4} + \frac{\cosh R}{R \sinh^3 R} \right) + c_2 \left( \frac{R \cosh R}{\sinh^3 R} + \frac{\cosh R}{R \sinh R} - \frac{2}{\sinh R} \right), \tag{D.24}$$

$$\mathbf{f}_{1,4} = c_1 \left( -\frac{1}{R^3 \sinh R} - \frac{\cosh R}{R^2 \sinh^2 R} + \frac{1 + \cosh^2 R}{R \sinh^3 R} \right) + c_2 \left( \frac{R(1 + \cosh R^2)}{\sinh^3 R} + \frac{3}{R \sinh R} - \frac{5 \cosh R}{\sinh^2 R} \right), \tag{D.25}$$

where  $c_1, c_2$  are undtermined constants. Note that the functions (D.22)-(D.25) have a regular power series expansion at R = 0 with nonnegative even powers of R only. We note that there is another homogeneous solution that we discarded because it is singular at R = 0:

$$\mathbf{f}_{1,1} = c_3(\frac{\cosh^2 R}{\sinh^3 R}), \qquad \qquad \mathbf{f}_{1,2} = -c_3(\frac{\coth R}{R^4} + \frac{1}{R^3 \sinh^2 R} + \frac{\cosh^2 R}{R^2 \sinh^3 R}), \qquad \qquad \mathbf{f}_{1,3} = c_3(\frac{\coth R}{R^4} + \frac{1}{R^3 \sinh^2 R} + \frac{\cosh^2 R}{R^2 \sinh^3 R}), \qquad \qquad \mathbf{f}_{1,4} = c_3(\frac{\cosh R}{R^3 \sinh^2 R} + \frac{1 + \cosh^2 R}{R^2 \sinh^3 R}).$$

$$(D.26)$$

We are left with two unknown parameters  $c_1, c_2$  but one can be a adjusted to zero by shifting the center of the zeroth order solution:  $\vec{x} \to \vec{x} + c_0 \vec{\ell}$ , followed by a suitable gauge transformation to fix the  $\mathbf{f}_{1,6} = 0$  gauge, to set  $c_1 = 0$ . The parameter  $c_2$  is undetermined at this point, since it depends on the proper boundary conditions at  $R = \infty$  and at R = -1/b.

Now we move on to order  $O(b^2)$ . The general ansatz at this order is:

$$b^{2}\phi^{a(2)} = \ell^{2}x_{a}\mathbf{f}_{2,1}(R) + [\ell_{a}(\ell_{k}x_{k}) - \frac{1}{3}\ell^{2}x_{a}]\mathbf{f}_{2,3}(R) + x_{a}[(\ell_{k}x_{k})(\ell_{m}x_{m}) - \frac{1}{3}\ell^{2}R^{2}]\mathbf{f}_{2,4}(R), \quad (D.27)$$
  

$$bA_{i}^{a(2)} = \ell^{2}\epsilon_{iak}x_{k}\mathbf{f}_{2,2}(R) + x_{a}\epsilon_{ijk}x_{j}\ell_{k}(\ell_{m}x_{m})\mathbf{f}_{2,5}(R) + x_{i}\epsilon_{ajk}x_{j}\ell_{k}(\ell_{m}x_{m})\mathbf{f}_{2,6}(R) + \epsilon_{aij}[\ell_{j}(\ell_{m}x_{m}) - \frac{1}{3}\ell^{2}x_{j}]\mathbf{f}_{2,7}(R) + (\ell_{i}\epsilon_{ajk}x_{j}\ell_{k} - \frac{1}{3}\ell^{2}\epsilon_{aji}x_{j})\mathbf{f}_{2,8}(R), \quad (D.28)$$

where we have separated the different terms according to whether they can be expressed in terms of the spin-0 combination  $\ell^2 \equiv \ell_k \ell_k$  or the spin-2 combination  $\ell_k \ell_m - \frac{1}{3} \ell^2 \delta_{km}$ . We again used the identity (D.8) to eliminate the term  $\epsilon_{aij} x_j (\vec{\ell} \cdot \vec{x})^2$ , and we also note the identity  $\ell_{[i}\epsilon_{a]jk}x_{j}\ell_{k} = \frac{1}{2}\epsilon_{aij}\ell_{j}(\ell_{k}x_{k}) - \frac{1}{2}\ell^{2}\epsilon_{aij}x_{j}$ , which we used to eliminate a term of the form  $\ell_{a}\epsilon_{ijk}x_{j}\ell_{k}\mathbf{f}_{2,9}$ . At order  $O(b^{2})$  the possible gauge parameters are of the form:

$$\lambda^a = \epsilon_{abc} x_b \ell_c(\ell_k x_k) \mathbf{g}_{2,1}(R) \,,$$

and we use the corresponding gauge transformation to gauge fix  $\mathbf{f}_{2,8} = 0$ . Our parameters  $\mathbf{f}_{2,1}, \mathbf{f}_{2,2}$  have spin-0, while  $\mathbf{f}_{2,3}, \ldots, \mathbf{f}_{2,7}$  have spin-2. The spin-2 equations are:

$$0 = \frac{1}{R}\mathbf{f}_{2,4}' - \frac{1}{R}\mathbf{f}_{2,5}' - K\mathbf{f}_{2,4} + K\mathbf{f}_{2,5} + (H - K)\mathbf{f}_{2,6} + \frac{1}{R}\mathbf{f}_{1,2}' - K\mathbf{f}_{1,2} - H\mathbf{f}_{1,4} - \mathbf{f}_{1,2}\mathbf{f}_{1,4} - \mathbf{f}_{1,3}\mathbf{f}_{1,4}, \qquad (D.29)$$

$$0 = \frac{1}{R}\mathbf{f}_{2,7}' + H\mathbf{f}_{2,7} + K\mathbf{f}_{2,3} + (1 + R^2K)\mathbf{f}_{2,4} - 2\mathbf{f}_{2,6} + K\mathbf{f}_{1,1} + \mathbf{f}_{1,2} + R^2K\mathbf{f}_{1,2} - \mathbf{f}_{1,1}\mathbf{f}_{1,3}, \qquad (D.30)$$

$$0 = \frac{1}{R}\mathbf{f}_{2,3}' - (1+R^2K)\mathbf{f}_{2,5} + (1-R^2H)\mathbf{f}_{2,6} + (K-H)\mathbf{f}_{2,7}$$
(D of)

$$+\frac{1}{R}\mathbf{f}_{1,1} + R^{2}H\mathbf{f}_{1,4} + \mathbf{f}_{1,1}\mathbf{f}_{1,3} + \mathbf{f}_{1,1}\mathbf{f}_{1,4} + R^{2}\mathbf{f}_{1,2}\mathbf{f}_{1,4} + R^{2}\mathbf{f}_{1,3}\mathbf{f}_{1,4}, \qquad (D.31)$$

$$0 = R\mathbf{f}_{2,5}' - \frac{1}{R}\mathbf{f}_{2,7}' - K\mathbf{f}_{2,3} + 2\mathbf{f}_{2,4} + 4\mathbf{f}_{2,5} + (2 + R^{2}K)\mathbf{f}_{2,6} + K\mathbf{f}_{2,7}$$

 $-K\mathbf{f}_{1,1} + \mathbf{f}_{1,2} + \mathbf{f}_{1,1}\mathbf{f}_{1,3} + R^2\mathbf{f}_{1,3}\mathbf{f}_{1,4}, \qquad (D.32)$ 

$$0 = \mathbf{f}_{2,3} - R^2 \mathbf{f}_{2,6} - \mathbf{f}_{2,7} - R^2 \mathbf{f}_{1,1} \mathbf{f}_{1,3} - R^4 \mathbf{f}_{1,3} \mathbf{f}_{1,4} \,. \tag{D.33}$$

The spin-0 equations are:

$$0 = \mathbf{f}_{2,1}' + \frac{1}{R}\mathbf{f}_{2,1} + \frac{2}{R}(1 + R^2K)\mathbf{f}_{2,2} + \frac{1}{3}\mathbf{f}_{1,1}' + \frac{1}{3}R^2\mathbf{f}_{1,2}' + \frac{2}{3}R\mathbf{f}_{1,2} - \frac{2}{3}R\mathbf{f}_{1,1}\mathbf{f}_{1,4}, \quad (D.34)$$
  
$$0 = \mathbf{f}_{2,2}' + (\frac{2}{R} + RH)\mathbf{f}_{2,2} + \frac{1}{R}(1 + R^2K)\mathbf{f}_{2,1}$$

$$+\frac{1}{3}RK\mathbf{f}_{1,1} + \frac{1}{3}R(1+R^2K)\mathbf{f}_{1,2} + \frac{1}{2}R\mathbf{f}_{1,1}\mathbf{f}_{1,3} - \frac{1}{3}R^3\mathbf{f}_{1,3}\mathbf{f}_{1,4}.$$
 (D.35)

We first solve the spin-0 equations. The general solution is given by:

$$\mathbf{f}_{2,1} = \frac{R^2}{36\sinh^2 R} + \frac{1}{6}R\coth R + c_5\left(\frac{1}{\sinh^2 R} - \frac{1}{R}\coth R\right) + c_6\left(\frac{1}{R\sinh^2 R}\right), \qquad (D.36)$$

$$\mathbf{f}_{2,2} = \frac{R^2 \cosh R}{36 \sinh^2 R} - \frac{R}{8 \sinh R} + c_5 \left(\frac{\cosh R}{\sinh^2 R} - \frac{1}{R \sinh R}\right) + c_6 \left(\frac{\cosh R}{R \sinh^2 R}\right). \tag{D.37}$$

Since  $c_6$  multiplies an *R*-odd and singular solution, we set  $c_6 = 0$ . The unknown  $c_5$  needs to be determined by the boundary conditions at  $R = \infty$  and R = -a.

Now, we move on to the spin-2 equations. First we look for a solution of the homogeneous spin-2 part:

$$0 = \frac{1}{R}\mathbf{f}_{2,4}' - \frac{1}{R}\mathbf{f}_{2,5}' - K\mathbf{f}_{2,4} + K\mathbf{f}_{2,5} + (H - K)\mathbf{f}_{2,6}, \qquad (D.38)$$

$$0 = \frac{1}{R}\mathbf{f}'_{2,7} + H\mathbf{f}_{2,7} + K\mathbf{f}_{2,3} + (1 + R^2K)\mathbf{f}_{2,4} - 2\mathbf{f}_{2,6}, \qquad (D.39)$$

$$0 = \frac{1}{R}\mathbf{f}'_{2,3} - (1+R^2K)\mathbf{f}_{2,5} + (1-R^2H)\mathbf{f}_{2,6} + (K-H)\mathbf{f}_{2,7}, \qquad (D.40)$$

$$0 = R\mathbf{f}_{2,5}' - \frac{1}{R}\mathbf{f}_{2,7}' - K\mathbf{f}_{2,3} + 2\mathbf{f}_{2,4} + 4\mathbf{f}_{2,5} + (2 + R^2 K)\mathbf{f}_{2,6} + K\mathbf{f}_{2,7}, \qquad (D.41)$$

$$0 = \mathbf{f}_{2,3} - R^2 \mathbf{f}_{2,6} - \mathbf{f}_{2,7} \,. \tag{D.42}$$

The general solution that is well behaved as  $R \to \infty$  is:

$$\mathbf{f}_{2,3}^{(homog)} = c_7 \{ \frac{4R}{\sinh R} \} + c_8 \{ \frac{4}{R^4 \sinh R} \}, \qquad (D.43)$$

$$\mathbf{f}_{2,4}^{(homog)} = c_7 \left\{ \frac{6\cosh R - 4}{R\sinh R} - \frac{2}{\sinh^2 R} \right\} + c_8 \left\{ -\frac{4(\cosh R + 1)}{R^6 \sinh R} - \frac{2}{R^5 \sinh^2 R} \right\}, \tag{D.44}$$

$$\mathbf{f}_{2,5}^{(homog)} = c_7 \left\{ -\frac{2\cosh R}{\sinh^2 R} - \frac{4\cosh R - 6}{R\sinh R} \right\} + c_8 \left\{ -\frac{4(\cosh R + 1)}{R^6 \sinh R} - \frac{2\cosh R}{R^5 \sinh^2 R} \right\}, \tag{D.45}$$

$$\mathbf{f}_{2,6}^{(homog)} = c_7 \left\{ -\frac{2}{R\sinh R} + \frac{2\cosh R}{\sinh^2 R} \right\} + c_8 \left\{ \frac{8}{R^6 \sinh R} + \frac{2\cosh R}{R^5 \sinh^2 R} \right\},\tag{D.46}$$

$$\mathbf{f}_{2,7}^{(homog)} = c_7 \{ \frac{6R}{\sinh R} - \frac{2R^2 \cosh R}{\sinh^2 R} \} + c_8 \{ -\frac{4}{R^4 \sinh R} - \frac{2 \cosh R}{R^3 \sinh^2 R} \}.$$
(D.47)

Additionally, there are two more linearly independent solutions that grow exponentially as  $R \to \infty$ . They are given by:

$$\mathbf{f}_{2,3}^{(homog)} = c_9 \{ -\frac{2\cosh^2 R}{R^2 \sinh R} + \frac{6\cosh R}{R^3} - \frac{6\sinh R}{R^4} \} + c_{10} \{ -\frac{6\cosh R}{R^4} - \frac{2\cosh R}{R^2} + \frac{6\cosh^2 R}{R^3 \sinh R} \},$$
(D.48)

$$\mathbf{f}_{2,4}^{(homog)} = c_9 \left\{ \frac{6\sinh R}{R^6} - \frac{3(1+2\cosh R)}{R^5} + \frac{2\cosh^2 R}{R^4\sinh R} + \frac{1}{R^3\sinh^2 R} \right\} + c_{10} \left\{ \frac{6(1+\cosh R)}{R^6} - \frac{3(\cosh R+2\cosh^2 R)}{R^3\sinh R} + \frac{2\cosh R}{R^4} - \frac{3}{R^4\sinh^2 R} \right\}, \qquad (D.49)$$

$$\mathbf{f}_{2,5}^{(homog)} = c_9 \{ \frac{6\sinh R}{R^6} - \frac{3(2 + \cosh R)}{R^5} + \frac{2\coth R}{R^4} + \frac{\cosh R}{R^3 \sinh^2 R} \} + c_{10} \{ \frac{6\cosh R}{R^6} + \frac{6}{R^6} - \frac{3\cosh^2 R}{R^5 \sinh R} - \frac{6\coth R}{R^5} + \frac{2}{R^4} - \frac{3\cosh R}{R^4 \sinh^2 R} \}, \quad (D.50)$$

$$\mathbf{f}_{2,6}^{(homog)} = c_9 \left\{ -\frac{12\sinh R}{R^6} + \frac{9\cosh R}{R^5} - \frac{2\cosh^2 R}{R^4\sinh R} - \frac{\cosh R}{R^3\sinh^2 R} \right\} \\ + c_{10} \left\{ -\frac{12\cosh R}{R^6} + \frac{9\cosh^2 R}{R^5\sinh R} + \frac{3\cosh R}{R^4\sinh^2 R} - \frac{2\cosh R}{R^4} \right\},$$
(D.51)

$$\mathbf{f}_{2,7}^{(homog)} = c_9 \{ \frac{6\sinh R}{R^4} - \frac{3\cosh R}{R^3} + \frac{\cosh R}{R\sinh^2 R} \} + c_{10} \{ \frac{6\cosh R}{R^4} - \frac{3\cosh^2 R}{R^3\sinh R} - \frac{3\cosh R}{R^2\sinh^2 R} \}.$$
(D.52)

Once we have a complete linearly independent set of solutions to the homogeneous problem, we can find the solution to the inhomogeneous problem by integration. When we perform the integration we obtain complicated expressions that contain polylogarithms

$$\operatorname{Li}_n(z) = \sum_{k=1}^{\infty} \frac{z^k}{k^n}.$$

For example, if we set  $c_2 = 0$  in (D.22)-(D.25), we get:

$$\begin{aligned} \mathbf{f}_{2,3}^{(inhomog)} &= -\frac{9}{2R^4 \sinh R} \operatorname{Li}_4(e^{-2R}) + \left[\frac{3}{R^3} (\sinh R - \frac{2}{\sinh R}) - \left(\frac{3}{R^4} + \frac{1}{R^2}\right) \cosh R\right] \operatorname{Li}_3(e^{-2R}) \\ &- \left[\frac{3}{R^2 \sinh R} + \left(\frac{6}{R^3} + \frac{2}{R}\right) \cosh R - \frac{6 \sinh R}{R^2}\right] \operatorname{Li}_2(e^{-2R}) \\ &+ \left[\left(\frac{6}{R^2} + 2\right) \cosh R - \frac{6}{R} \sinh R\right] \log(1 - e^{-2R}) - \frac{1}{\sinh R} \left(\frac{1}{2} + \frac{45}{2R^3} + \frac{2}{R} + \frac{59}{120}R\right) \\ &+ \left(\frac{R^2}{8 \sinh^2 R} + \frac{45}{2R^4} + \frac{15}{2R^2} + \frac{2}{R} + 2 + \frac{2}{3}R\right) \cosh R - \left(\frac{45}{2R^3} + \frac{2}{R} + 2 + \frac{2}{3}R\right) \sinh R , \end{aligned}$$
(D.53)

$$\begin{split} \mathbf{f}_{2,4}^{(inhomog)} &= \left(\frac{9}{2R^6} \sinh R + \frac{9}{2R^6} \coth R + \frac{1}{4R^6} \sinh R + \frac{1}{2R^5} \coth R + \frac{6}{R^5} \sinh R + \frac{3}{R^4} \sinh^2 R - \frac{3}{R^5} \sinh R\right] \operatorname{Li}_3(e^{-2R}) \\ &+ \left[\frac{6}{R^5} + \frac{3}{2R^3} \sinh^2 R + \left(\frac{6}{R^5} + \frac{1}{R^3}\right) \cosh R + \frac{6}{R^4} \cosh R + \frac{3}{R^4} \sinh^2 R - \frac{3}{R^5} \sinh R\right] \operatorname{Li}_3(e^{-2R}) \\ &+ \left[\frac{6}{R^3} \sinh R - \frac{3}{R^3} \coth R - \left(\frac{6}{R^4} + \frac{2}{R^2}\right) \cosh R + \frac{6}{R^4} \cosh R - \frac{6}{R^4} \sinh R\right] \operatorname{Li}_2(E^{-2R}) \\ &+ \left(\frac{45}{2R^5} + \frac{2}{R^3} + \frac{2}{R^2} + \frac{2}{3R}\right) \sinh R + \left(\frac{4R}{4R^4} + \frac{1}{R^2} + \frac{1}{4R} + \frac{3}{120}\right) \frac{1}{\sinh^2 R} - \frac{6}{R^4} \sinh^2 R \\ &+ \left(\frac{45}{2R^5} + \frac{2}{R^3} + \frac{1}{2R^2} + \frac{59}{120R}\right) \frac{1}{\sinh R} + \left(\frac{45}{4R^5} + \frac{2}{R^3} - \frac{1}{2R^2} + \frac{1}{5R}\right) \coth R \\ &- \left(\frac{45}{2R^6} + \frac{15}{R^4} + \frac{2}{R^3} + \frac{2}{R^2} + \frac{2}{3R}\right) \cosh R - \frac{45}{2R^6} - \frac{2}{R^3} + \frac{1}{8R^2}, \qquad (D.54) \\ \mathbf{f}_{2,5}^{(inhomog)} &= \left(\frac{9}{2R^6} \sinh R + \frac{9}{2R^6} \cosh R - \frac{3}{2R^5} \sinh R + \frac{6}{R^5} \coth R + \frac{15}{2R^5} \sinh R + \frac{3}{R^4 \sinh^2 R}\right) \operatorname{Li}_3(e^{-2R}) \\ &+ \left(\frac{6}{R^5} + \frac{2}{R^3} + \frac{6}{R^5} \cosh R - \frac{3}{R^3} \sinh R + \frac{3}{R^4} \cosh R + \frac{15}{R^6 \sinh^2 R}\right) \operatorname{Li}_2(e^{-2R}) \\ &+ \left(\frac{6}{R^5} + \frac{2}{R^3} + \frac{6}{R^5} \cosh R - \frac{3}{R^3} \sinh R + \frac{3}{R^4} \sinh R\right) \log(1 - e^{-2R}) \\ &- \left(\frac{6}{R^4} + \frac{2}{R^2} + \frac{6}{R^5} \cosh R - \frac{3}{R^3} \sinh R + \frac{3}{R^3 \sinh R}\right) \log(1 - e^{-2R}) \\ &- \left(\frac{6}{R^4} + \frac{2}{R^3} - \frac{3}{2R^2} - \frac{3}{2R} - \left(\frac{45}{2R^5} + \frac{2}{R^3} + \frac{1}{R^2}\right) \cosh R \\ &+ \left(\frac{45}{4R^5} + \frac{2}{R^3} - \frac{3}{2R^2} - \frac{2}{2R} - \left(\frac{4}{2R^5} + \frac{2}{R^3} + \frac{1}{R^2}\right) \cosh R \\ &+ \left(\frac{45}{4R^5} + \frac{2}{R^3} - \frac{3}{2R^2} - \frac{2}{2R} - \left(\frac{4}{2R^5} + \frac{2}{R^3} + \frac{1}{R^2}\right) \cosh R \\ &+ \left(\frac{45}{4R^5} + \frac{2}{R^3} - \frac{3}{2R^2} - \frac{2}{2R}\right) \operatorname{Li}_4(e^{-2R}) \\ &- \left[\left(\frac{6}{R^4} + \frac{1}{R^4} + \frac{2}{R^3}\right) \sinh R + \left(\frac{45}{2R^5} + \frac{2}{R^3} + \frac{1}{R^2}\right) \cosh R \\ \\ &+ \left(\frac{45}{4R^5} + \frac{2}{R^3} - \frac{3}{2R^2} - \frac{2}{2R}\right) \operatorname{Li}_4(e^{-2R}) \\ &- \left[\left(\frac{6}{R^4} + \frac{1}{R^4}\right) \cosh R - \frac{9}{R^3} \sinh R + \frac{2}{R^5} \sinh^2 R \right] \operatorname{Li}_2(e^{-2R}) \\ &- \left[\left(\frac{6}{R^4} + \frac{1}{R^4}\right) \cosh R - \frac{9}{R^3} \sinh R + \frac{2}{R^5} \frac{2}{R^3} \sin R^2}\right] \operatorname{Li}_2(e^{-2R}) \\ &- \left[\left(\frac{6}{R^4} + \frac{1}{R^4} + \frac{1}{3}$$

We note that the combinations of polylogarithms that appear here are the results of the integrals

$$\int R^3 \coth R \, dR = -\frac{3}{4} \operatorname{Li}_4(e^{-2R}) - \frac{3}{2} R \operatorname{Li}_3(e^{-2R}) - \frac{3}{2} R^2 \operatorname{Li}_2(e^{-2R}) + R^3 \log(1 - e^{-2R}) + \frac{1}{4} R^4,$$

and

$$\int R^2 \coth R \, dR = -\frac{1}{2} \operatorname{Li}_3(e^{-2R}) - R \operatorname{Li}_2(e^{-2R}) + R^2 \log(1 - e^{-2R}) + \frac{1}{3}R^3$$

When we turn on  $c_2 \neq 0$  we get additional terms, but they can be expressed as rational functions of  $e^R$  and R and do not cancel the polylogarithms. In any case, this demonstrates that a simple solution to the BPS equations (3.40), involving only basic functions, does not exist.